MECHANISMS OF DRUG RESISTANCE IN GLIAL CELLS

•

by

Catherine L. Nutt

Department of Pathology

Submitted in partial fulfilment of the requirements for the degree of Doctor of Philosophy

Faculty of Graduate Studies The University of Western Ontario London, Ontario September, 1997

[©] Catherine L. Nutt, 1997



of Canada

Acquisitions and Bibliographic Services

395 Wellington Street Ottawa ON K1A 0N4 Canada du Canada

Acquisitions et services bibliographiques

395, rue Wellington Ottawa ON K1A 0N4 Canada

Your file Votre reference

Our file Notre référence

The author has granted a nonexclusive licence allowing the National Library of Canada to reproduce, loan, distribute or sell copies of this thesis in microform, paper or electronic formats.

The author retains ownership of the copyright in this thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without the author's permission. L'auteur a accordé une licence non exclusive permettant à la Bibliothèque nationale du Canada de reproduire, prêter, distribuer ou vendre des copies de cette thèse sous la forme de microfiche/film, de reproduction sur papier ou sur format électronique.

L'auteur conserve la propriété du droit d'auteur qui protège cette thèse. Ni la thèse ni des extraits substantiels de celle-ci ne doivent être imprimés ou autrement reproduits sans son autorisation.

0-612-28512-X

Canadä

ABSTRACT

The most common human primary intracranial tumors are the gliomas. Despite intensive research, attempts to develop effective chemotherapy regimens for gliomas have met with limited success. In this thesis, two main approaches were taken to test the hypothesis that biochemical properties intrinsic to specific subtypes of glial tumors play a role in the resistance of these tumors to chemotherapeutic agents. The first approach was to examine biochemical properties inherent to the normal glial cell of origin. To accomplish this, the possibility of differential expression of drug resistance genes in cultures of purified rat glial cells was explored. Although the response of normal rat glial cells to BCNU appeared to be dependent on more than one mechanism of drug resistance, evidence of roles for O⁶-methylguanine-DNA methyltransferase (MGMT) and the glutathioneglutathione-S-transferase detoxification pathway in response to BCNU were demonstrated. The second approach of this thesis was to examine the impact of a genetic alteration, found commonly in astrocytic gliomas, on the response of otherwise normal astrocytes to BCNU. Mutations of the p53 gene, and subsequent loss of p53 function, are a common occurrence in astrocytic tumors. Therefore, the impact of the loss of p53 function on response of mouse astrocytes to BCNU was examined. Wild type p53 appeared to protect mouse astrocytes from the cytotoxic effects of BCNU in a gene-dosedependent fashion. This resistance displayed by wild type astrocytes could not be explained by either p53-dependent apoptosis or differential cell cycle arrest. In addition, although results supported a role for p53 in the regulation of MGMT, the p53-dependent BCNU resistance displayed by mouse astrocytes appeared to be mediated by a non-MGMT mechanism. Studies presented here demonstrated that knockout p53 astrocytes could undergo apoptosis both in the absence of drug and at increased levels in response to BCNU treatment. It is hypothesized that DNA damage in these knockout astrocytes may be detected by an alternative damage-sensing mechanism and subsequently, the cells are triggered to undergo p53-independent apoptosis. Together, these experiments provide support for the hypothesis that biochemical properties intrinsic to specific subtypes of glial tumors play a role in resistance to chemotherapeutic agents.

Keywords: cancer, tumor, brain, chemotherapy, drug resistance, alkylating agent, nitrosourea, BCNU, astrocytoma, oligodendroglioma, glial cell, astrocyte, oligodendrocyte progenitor, oligodendrocyte, O⁶-methylguanine-DNA methyltransferase, glutathione, glutathione-*S*-transferase, p53, cell cycle arrest, apoptosis

This thesis is dedicated in loving memory of

David John Briggs

...an unspoken promise made many years ago...

This one is for you, Grandpa.

ACKNOWLEDGEMENTS

I would like to begin by thanking my co-supervisors, Dr. Greg Cairncross and Dr. Ann Chambers for their continued support and guidance through my graduate training. Greg has been extremely generous over the years and his patience has been unparalleled. I can only hope that my perseverance (stubbornness?!) has repayed some of the debt owed; I knew those cells would grow! Ann's open-door policy was one that I could always count on, especially on Sunday mornings when I needed someone with whom to share exciting news. Her career guidance over this last year also has been extremely helpful. I am immeasurably grateful to them both.

I would like to thank Dr. Jim Koropatnick and Dr. Chris Naus, members of my advisory committee, for stimulating discussions, advice and guidance. In particular, I would like to thank Jim for always being free to discuss science and his seemingly neverending willingness to write letters of recommendation.

I would like to acknowledge those who taught me the many technical skills I have acquired in my years as a graduate student. Dr. Linda Bambrick taught me the original rat glial cell culture system used in Chapter 3 and Drs. Margot Mayer and Mark Noble were kind enough to teach me the purified rat glial cell culture system utilized in Chapter 4. Mike Keeney, Lori Anderson, Jennifer Koster, and Mike Kadri were all extremely generous with their time when it came to teaching me flow cytometry. Lastly (but certainly not least), I am exceedingly grateful for the overwhelming source of information and assistance provided over the years by Sylvia Hill, the technical genius in my life.

I also would like to acknowledge the contributions of others to the work presented in this thesis. Michelle Schoenhaus and Dr. Dan Yarosh performed the MGMT assays on rat glial cells described in Chapter 3 and Natasha Loktionova and Dr. Tony Pegg performed the MGMT assays on mouse glial cells described in Chapter 7.

I wish to thank Debbie Aristone, Patricia Grant, Pat Finlay, Pat Gray, Maggie Stawarski, Betty Gardiner, Gail Howard, Denise Power, Karl Swartz and Paul Payson for their excellent assistance and friendship.

Personal stipends were provided, in part, by an Ontario Graduate Scholarship and a Dean of Medicine Graduate Student Fellowship (University of Western Ontario).

For friendship, advice and never-ending support, I wish to thank Don van Meyel, Andrea McCart, Nancy Kerkvliet, Valentia Lee-Brotherton, Alan Tuck, Stewart Daly, Wayne Bocchinfuso, Allen Grolla, David Dales, and Randy Florell. Dr. Mady Zlatescu has filled Don's vacant spot nicely and I thank her for her friendship over the last year. In particular, I would like to express deep gratitude to Sylvia Hill. She has provided me with a place of sanctuary down the hall for my entire time as a graduate student. She, like no one else, could put a smile on my face during times of tribulation (lab or life) and was a great friend through it all.

I would like to thank those outside of the lab who have provided support, friendship and sanity checks over the last few years. My friends outside the lab have been great; their understanding when it came to hectic schedules and missed events has been greatly appreciated. I would like to express great gratitude to the entire Boomhower family. In particular, Rosa, Bob, and Roberta have continued to support me beyond my greatest expectations and for that, I can not thank them enough. The support and understanding of family has been overwhelming. In particular, I would like to thank my parents who really came through for me these last few months (not to mention these last few weeks!!). Words could never express my gratitude for their continued support.

Finally, I would like to thank Ken Boomhower. He shared the last four years of my graduate career with me and, through it all, was a constant source of care and support. No one could make me feel better about the trials and tribulations of graduate school than he could. I am sure there is no way to make him comprehend fully how much his being there meant to me. The nightly "carpal tunnel relief" phone calls were a never-ending source of comfort and are sure to be the only thing I will miss about writing a thesis.

vii

TABLE OF CONTENTS

Page

	••
	.11
	۷ ::
	V1
	.11
	.11
	.11
	V1
	H
CHAPTER 1 - INTRODUCTION	1
1.1 Thesis Overview	1
1.2 Glial Neoplasms	3
1.2.1 Introduction	3
1.2.2 Pathology	3
1.2.3 Epidemiology	5
1.2.4 Risk Factors	6
1.2.5 Molecular Tumor Biology	7
1.2.6 Treatment	.1
1.2.6.1 Surgery	1
1.2.6.2 Radiation Therapy	.1
1.2.6.3 Chemotherapy	2
1.2.7 Summary	3
1.3 Experimentally Induced Brain Tumors	.4
1.4 Glial Cell Lineages in the Mammalian Brain	.5
1.4.1 Glial Cells in vitro	5
1.4.2 Glial Cells in vivo	7
1.5 Mechanisms of Resistance to BCNU	.8
1.5.1 Introduction	.8
1.5.2 O ⁶ -Methylguanine-DNA Methyltransferase	9
1.5.3 Glutathione and Glutathione-S-Transferases	1
1.5.4 Nucleotide-Excision Repair Proteins	3
1.6 Additional Mechanisms of Drug Resistance Relevant to Glial Tumors2	4
1.6.1 Introduction	4
1.6.2 Multidrug Resistance (P-glycoprotein)	.5
1.6.3 Multidrug Resistance-Associated Protein	5
1.6.4 Metallothionein 2	6
1.6.5 Topoisomerase II	6

1.7 The p53 Tumor Suppressor	27
1.7.1 Introduction	27
1.7.2 The <i>p53</i> gene	29
1.7.3 The p53 protein	30
1.7.4 The Role of p53 in Response to DNA-Damaging Agents	31
1.7.4.1 Introduction	31
1.7.4.2 Cell Cycle Arrest	32
1.7.4.3 Apoptosis	33
1.7.4.4 p53-Regulated Drug Resistance Mechanisms	35
1.7.4.5 Summary	36
CHAPTER 2 - MATERIALS AND METHODS	37
2.1 Materials	37
2.1.1 Cell Culture Materials and Reagents	37
2.1.2 Molecular Biology Reagents	38
2.1.3 Primers, Probes and Plasmids	38
2.2 Rat Primary Cell Cultures	39
2.2.1 Isolation of Glial Cell Enriched Cultures	39
2.2.1.1 Immunocytochemistry	40
2.2.2 Isolation of Purified Oligodendrocyte Progenitors	and
Oligodendrocytes	40
2.2.2.1 Immunocytochemistry	43
2.3 Mouse Primary Cell Cultures	43
2.3.1 Mouse Husbandry	43
2.3.2 Mouse $p53$ Genotyping	44
2.3.3 Isolation of Astrocytes and Fibroblasts	47
2.4 Cell Proliferation Assay	47
2.5 Cytotoxicity Assay	48
2.6 MGMT Assavs	49
2.7 Northern Blot Analysis	50
2.7.1 RNA Isolation and Selection of mRNA	50
2.7.2 Gel Electrophoresis of mRNA	51
2.7.3 Northern Transfer	52
2.7.4 Radioactive Labelling of Hybridization Probes	52
2.7.5 Hybridization of Blots	53
2.7.6 PhosphorImage and Densitometry Analysis	54
2.8 Flow Cytometry	54
2.9 Detection of Apontosis	55
2.9.1 ApopTag TM In Situ Apoptosis Detection Kit	55
2.9.2 Cell Death Detection ELISA	56
	23
CHAPTER 3 - THE ROLE OF OF METHYL GUANINE-DNA METHYL TRANSFERA	SE
IN THE RESPONSE OF GLIAL LINEAGE CELLS TO BONU	58
3.1 Introduction	58

3.2	Results
	3.2.1 Immunochemical characterization of rat glial cell enriched cultures
	3.2.2 Effect of BCNU concentration on rat glial cell viability 59
	3.2.3 MGMT activity in rat glial cell cultures
3.3	Discussion
CHAPTE	R 4 - DIFFERENTIAL EXPRESSION OF DRUG RESISTANCE GENES
IN	CULTURES OF PURIFIED GLIAL CELLS
4.1	
4.2	Kesults
	4.2.1 Immunochemical characterization of cultures of purified rat ghas cells
	4.2.2 Differential expression of drug resistance genes in purified rat glial cells.
	4.2.3 Effect of BCNU and MNU treatment on the viability of purified rat
	glial cells
	4.2.4 Effect of BO prefeatment on the response of purfied fat ghat cens
	4.2.5 Effect of BSO and OTZ pretreatment on the response of astrocytes
	to BCNU
4.3	Discussion
CHAPTE	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND
CHAPTE CE	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU
CHAPTE CE	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU
CHAPTE CE 5.1	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU 95 Introduction 95
CHAPTE CE 5.1 5.2	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU95Introduction95Results96
CHAPTE CE 5.1 5.2	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU
CHAPTE CE 5.1 5.2	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU 95 Introduction 95 Results 96 5.2.1 Effect of p53 on the viability of mouse astrocytes exposed to BCNU 96
CHAPTE CE 5.1 5.2	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU 95 Introduction 95 Results 96 5.2.1 Effect of p53 on the viability of mouse astrocytes exposed to BCNU 96 5.2.2 Effect of p53 on the viability of mouse astrocytes and fibroblasts in
CHAPTE CE 5.1 5.2	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU 95 Introduction 95 Results 96 5.2.1 Effect of p53 on the viability of mouse astrocytes exposed to BCNU. 96 5.2.2 Effect of p53 on the viability of mouse astrocytes and fibroblasts in response to cytotoxic treatment. 99
CHAPTE CE 5.1 5.2	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU 95 Introduction 95 Results 96 5.2.1 Effect of p53 on the viability of mouse astrocytes exposed to BCNU. 96 5.2.2 Effect of p53 on the viability of mouse astrocytes and fibroblasts in response to cytotoxic treatment. 99 5.2.3 Effect of BCNU treatment on p53-regulated gene expression. 99
CHAPTE CE 5.1 5.2	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU 95 Introduction 95 Results 96 5.2.1 Effect of p53 on the viability of mouse astrocytes exposed to BCNU 96 5.2.2 Effect of p53 on the viability of mouse astrocytes and fibroblasts in response to cytotoxic treatment. 99 5.2.3 Effect of BCNU treatment on p53-regulated gene expression. 99 5.2.4 Cell cycle distribution of mouse astrocytes treated with BCNU or radiation 112
CHAPTE CE 5.1 5.2	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU 95 Introduction 95 Results 96 5.2.1 Effect of p53 on the viability of mouse astrocytes exposed to BCNU 96 5.2.2 Effect of p53 on the viability of mouse astrocytes and fibroblasts in response to cytotoxic treatment. 99 5.2.3 Effect of BCNU treatment on p53-regulated gene expression. 99 5.2.4 Cell cycle distribution of mouse astrocytes treated with BCNU or radiation. 112 Discussion 123
CHAPTE CE 5.1 5.2 5.3	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU 95 Introduction 95 Results 96 5.2.1 Effect of p53 on the viability of mouse astrocytes exposed to BCNU. 96 5.2.2 Effect of p53 on the viability of mouse astrocytes and fibroblasts in response to cytotoxic treatment. 99 5.2.3 Effect of BCNU treatment on p53-regulated gene expression. 99 5.2.4 Cell cycle distribution of mouse astrocytes treated with BCNU or radiation. 112 Discussion 123
CHAPTE CE 5.1 5.2 5.3 CHAPTE	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU Introduction 95 Introduction 95 Results 96 5.2.1 Effect of p53 on the viability of mouse astrocytes exposed to BCNU. 96 5.2.2 Effect of p53 on the viability of mouse astrocytes and fibroblasts in response to cytotoxic treatment. 99 5.2.3 Effect of BCNU treatment on p53-regulated gene expression. 99 5.2.4 Cell cycle distribution of mouse astrocytes treated with BCNU or radiation. 112 Discussion 123 R 6 - THE EFFECT OF p53 STATUS ON THE APOPTOTIC RESPONSE
CHAPTE CE 5.1 5.2 5.3 CHAPTE OF	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU Introduction 95 Results 96 5.2.1 Effect of p53 on the viability of mouse astrocytes exposed to BCNU 96 5.2.2 Effect of p53 on the viability of mouse astrocytes and fibroblasts in response to cytotoxic treatment. 99 5.2.3 Effect of BCNU treatment on p53-regulated gene expression. 99 5.2.4 Cell cycle distribution of mouse astrocytes treated with BCNU or radiation. 112 Discussion 123 R 6 - THE EFFECT OF p53 STATUS ON THE APOPTOTIC RESPONSE MOUSE ASTROCYTES EXPOSED TO BCNU. 128
CHAPTE CE 5.1 5.2 5.3 CHAPTE OF 6.1	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU 95 Introduction 95 Results 96 5.2.1 Effect of p53 on the viability of mouse astrocytes exposed to BCNU. 96 5.2.2 Effect of p53 on the viability of mouse astrocytes and fibroblasts in response to cytotoxic treatment. 99 5.2.3 Effect of BCNU treatment on p53-regulated gene expression. 99 5.2.4 Cell cycle distribution of mouse astrocytes treated with BCNU or radiation. 112 Discussion 123 R 6 - THE EFFECT OF p53 STATUS ON THE APOPTOTIC RESPONSE MOUSE ASTROCYTES EXPOSED TO BCNU 128 Introduction 128
CHAPTE CE 5.1 5.2 5.3 CHAPTE OF 6.1 6.2	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU 95 Introduction 95 Results 96 5.2.1 Effect of p53 on the viability of mouse astrocytes exposed to BCNU 96 5.2.2 Effect of p53 on the viability of mouse astrocytes and fibroblasts in response to cytotoxic treatment. 99 5.2.3 Effect of BCNU treatment on p53-regulated gene expression. 99 5.2.4 Cell cycle distribution of mouse astrocytes treated with BCNU or radiation. 112 Discussion 123 R 6 - THE EFFECT OF p53 STATUS ON THE APOPTOTIC RESPONSE MOUSE ASTROCYTES EXPOSED TO BCNU 128 Introduction 128 Results 129
CHAPTE CE 5.1 5.2 5.3 CHAPTE OF 6.1 6.2	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU
CHAPTE CE 5.1 5.2 5.3 CHAPTE OF 6.1 6.2	R 5 - THE EFFECT OF p53 STATUS ON THE VIABILITY AND LL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU

6.2.3 Detection of apoptosis in mouse astrocytes after BCNU treatment
6.2.4 Effect of BCNU treatment on the expression of <i>bar</i> , <i>bcl</i> ₂ , and <i>bcl</i> ₂
$x_1 + x_2 = 134$
6.3 Discussion
CHAPTER 7 - THE EFFECT OF p53 STATUS ON O ⁶ -METHYLGUANINE-DNA
METHYLTRANSFERASE ACTIVITY IN MOUSE ASTROCYTES
7.1 Introduction 147
7.2 Results
7.2.1 Effect of p53 status on MGMT protein and mRNA in mouse astrocytes and fibroblasts.
7.2.2 Effect of BG pretreatment on the response of mouse astrocytes and
fibroblasts to BCNU.
7.2.3 Effect of BCNU and BG treatment on MGMT levels 154
7.3 Discussion
CHAPTER 8 - SUMMARY AND CONCLUSIONS 167
APPENDIX I
REFERENCES
VITA

LIST OF TABLES

Table	Description]	Page
6.1	Percent BAD in Astrocyte Flow Cytometry	· · · · · · · · · · · · · · · · · · ·	130

LIST OF FIGURES

Figure	e Description I	Page
2.1	Mouse <i>p53</i> genotyping by PCR	. 46
3.1	Immunochemical characterization of rat glial cell enriched cultures	61
3.2	Effect of BCNU concentration on rat glial cell viability	63
3.3	MGMT activity in rat glial cell cultures expressed as molecules/cell	66
4.1	Immunochemical characterization of purified rat glial cell cultures	. 71
4.2	Differential expression of drug resistance genes in cultures of purified cells	glial 73
4.3	Densitometry analysis of <i>MGMT</i> and <i>GST</i> mRNA expression in purified rat cells	glial 75
4.4	Densitometry analysis of p53, MDR and MT mRNA expression in purified glial cells	i rat . 77
4.5	Effect of BCNU treatment on purified rat astrocytes, oligodendrocyte progenitiand oligodendrocytes	tors, 80
4.6	Effect of MNU treatment on purified rat astrocytes, oligodendrocyte progenia and oligodendrocytes	tors, 83
4.7	The effect of cell culture medium on the viability of rat astrocytes in response BCNU	se to . 85
4.8	Effect of BG pretreatment on the response of purified rat glial cells to BC	NU 87
4.9	Effect of BG pretreatment on the response of purified rat glial cells to MNU	U 89
4.10	Effect of BSO and OTZ pretreatment on the response of rat astrocyte BCNU	s to . 92
5 .1	Effect of p53 status on the viability of mouse astrocytes treated with BCNU	J 98

5.2	Effect of p53 status on the viability of mouse astrocytes in response to cytotoxic treatment
5.3	Effect of p53 status on the viability of mouse fibroblasts in response to cytotoxic treatment
5.4	Effect of BCNU treatment on the mRNA expression of $p53$ and $p53$ -regulated genes in wild type $p53$ mouse astrocytes
5.5	Effect of BCNU treatment on the mRNA expression of p53-regulated genes in knockout <i>p53</i> mouse astrocytes
5.6	Effect of BCNU treatment on the mRNA expression of $p53$ and p53-regulated genes in wild type $p53$ mouse fibroblasts
5.7	Effect of BCNU treatment on the mRNA expression of p53-regulated genes in knockout p53 mouse fibroblasts
5.8	Densitometry analysis of changes in <i>mdm</i> -2 mRNA expression in response to BCNU treatment
5.9	Densitometry analysis of changes in WAF1 mRNA expression in response to BCNU treatment
5.10	Summary of cell cycle distribution of mouse astrocytes treated with 20 μ g/ml BCNU for 2 hrs or 2 Gy radiation
5.11	Cell cycle distribution of mouse astrocytes treated with 20 μ g/ml BCNU for 2 hrs
5.12	Cell cycle distribution of mouse astrocytes treated with 2 Gy radiation 122
6.1	Detection of apoptotic mouse astrocytes after treatment with BCNU 133
6.2	Semi-quantitative analysis of apoptosis in mouse cells
6.3	Effect of BCNU treatment on the mRNA expression of <i>bax</i> , <i>bcl-x</i> , and <i>bcl-2</i> in wild type <i>p53</i> mouse astrocytes
6.4	Effect of BCNU treatment on the mRNA expression of <i>bax</i> , <i>bcl-x</i> , and <i>bcl-2</i> in knockout <i>p53</i> mouse astrocytes
6.5	Effect of BCNU treatment on the mRNA expression of <i>bax</i> , <i>bcl-x</i> , and <i>bcl-2</i> in wild type <i>p53</i> mouse fibroblasts

6.6	Effect of BCNU treatment on the mRNA expression of <i>bax</i> , <i>bcl-x</i> , and <i>bcl-2</i> in knockout <i>p53</i> mouse fibroblasts
7.1	Effect of p53 on MGMT activity in mouse astrocytes and fibroblasts 150
7.2	Effect of BG treatment on MGMT activity in mouse astrocytes and fibroblasts
7.3	Effect of BG pretreatment on the response of wild type <i>p53</i> and knockout mouse astrocytes to BCNU
7.4	Effect of BG pretreatment on the response of wild type <i>p53</i> and knockout mouse fibroblasts to BCNU
7.5	Effect of BCNU and BG treatment on MGMT mRNA expression 160
7.6	Effect of BCNU treatment and radiation on MGMT activity in wild type <i>p53</i> and knockout mouse astrocytes
7.7	Effect of BCNU treatment and radiation on MGMT activity in wild type <i>p53</i> and knockout mouse fibroblasts

LIST OF APPENDICES

Appen	idix Description	Page
I	Certification of Approval of Animal Research	171

ABBREVIATIONS

+/+	wild type
+/-	heterozygous
-/-	knockout
ABC	ATP-binding cassette
ABTS	2,2'-azino-di-[3-ethylbenzthiazoline sulfonate(6)]
AGT	O ⁶ -alkylguanine-DNA alkyltransferase
AMCA	7-amino-4-methylcoumarin-3-acetic acid
ATCC	American Type Culture Collection
ATP	adenosine triphosphate
%BAD	percent background aggregates and debris
bax	bcl-associated x
bcl-2	B-cell leukemia/lymphoma-2
bcl-x	B-cell leukemia/lymphoma-x
BCNU	1,3-bis(2-chloroethyl)-1-nitrosourea
bFGF	basic fibroblast growth factor
BG	O ⁶ -benzylguanine
bp	base pair(s)
BSA	bovine serum albumin
BSO	DL-buthionine-[S,R]-sulfoximine
BTSG	Brain Tumor Study Group
°C	degrees Celsius
CCNU	1-(2-chloroethyl)-3-cyclohexyl-1-nitrosourea
CD44	cluster of differentiation 44
CDC	cell-division-cycle
CDK	cyclin-dependent kinase
cDNA	complementary deoxyribonucleic acid
cGy	centigray
CIP1	cyclin-dependent kinase-interacting protein-1

cm	centimetre(s)
CNS	central nervous system
⁶⁰ Co	cobalt-60
cpm	counts per minute
DAB	diaminobenzidine
dATP	2'-deoxyadenosine 5'-triphosphate
dCTP	2'-deoxycytidine 5'-triphosphate
DEPC	diethylpyrocarbonate
dGTP	2'-deoxyguanosine 5'-triphosphate
DMEM	Dulbecco's Modified Eagles Medium
DMSO	dimethylsulfoxide
DNA	deoxyribonucleic acid
DNase	deoxyribonuclease
dNTPs	deoxynucleotides
DTT	dithiothreitol
dTTP	2'-deoxythymidine 5'-triphosphate
ECOG	Eastern Cooperative Oncology Group
EDTA	ethylenediamine tetraacetic acid
EGF	epidermal growth factor
EGFR	epidermal growth factor receptor
ELISA	enzyme-linked immunosorbent assay
ENU	N-ethyl-N-nitrosourea
ERCC	excision repair cross-complementing
FBS	fetal bovine serum
FGF	fibroblast growth factor
γ	gamma
g	gram(s)
GADD	growth arrest and DNA damage inducible
GAPDH	glyceraldehyde-3-phosphate dehydrogenase

.

GC	galactocerebroside
GFAP	glial fibrillary acidic protein
GST	glutathione-S-transferase
HEPES	N-2-hydroxyethylpiperazine-N'2-ethanesulphonic acid
hr	hour(s)
Ig	immunoglobulin
IGF	insulin-like growth factor
JNK1	c-Jun kinase 1
kb	kilobase(s)
kD	kilodalton(s)
L	litre(s)
М	molar
mdm-2	murine double minute-2
MDR	multidrug resistance
MGAP	mouse glyceraldehyde-3-phosphate dehydrogenase
MGMT	O ⁶ -methylguanine-DNA methyltransferase
μCi	microcurie(s)
μg	microgram(s)
μl	microlitre(s)
μm	micrometre(s)
mg	milligram(s)
ml	millilitre(s)
mm	millimetre(s)
mM	millimolar
min	minute(s)
MMAC1	mutated in multiple advanced cancers 1
MNNG	N-methyl-N-nitro-N-nitrosoguanidine
MNU	N-methyl-N-nitrosourea
MOPS	3-(N-morpholino)propanesulphonic acid

mRNA	messenger ribonucleic acid
MRP	multidrug resistance-associated protein
MT	metallothionein
MTT	3-[4,5-dimethylthiazol-2-yl]-2,5-diphenyltetrazolium bromide
MW	molecular weight
Ν	normal
NER	nucleotide-excision repair
ng	nanogram(s)
nm	nanometre(s)
O2A	oligodendrocyte-type 2 astrocyte
OTZ	(-)-2-oxo-4-thiazolidine-carboxylic acid
PBS	phosphate buffered saline
PCNA	proliferating cell nuclear antigen
PCR	polymerase chain reaction
PDGF	platelet-derived growth factor
RAN-2	rat neural antigen-2
RNA	ribonucleic acid
RNase	ribonuclease
rpm	revolutions per minute
RTOG	Radiation Therapy Oncology Group
SDS	sodium dodecylsulphate
sec	second(s)
SSC	salt sodium citrate
Taq	Thermus aquaticus
TCA	trichloroacetic acid
TdT	terminal deoxynucleotidyl transferase
TE	Tris-EDTA
Tris	tris(hydroxymethyl)aminomethane
UV	ultraviolet
VEGF	vascular endothelial growth factor

- WAF1 wild type p53-activated fragmant
- WHO World Health Organization

CHAPTER 1

INTRODUCTION

1.1 THESIS OVERVIEW

Despite intensive research, attempts to develop effective chemotherapy regimens for gliomas have met with limited success. Significant progress was made when Cairncross and Macdonald (1988) reported that anaplastic oligodendrogliomas responded to a chemotherapy regimen of procarbazine, CCNU, and vincristine (PCV). It has been confirmed since that oligodendrogliomas and mixed oligoastrocytomas respond predictably to nitrosourea-based chemotherapy, especially PCV, whereas astrocytomas are relatively chemoresistant. Presumably oligodendroglioma cells are inherently susceptible to the cytotoxic effects of PCV, but why? In addition, a subset of astrocytomas have been observed to benefit from chemotherapy. What intrinsic properties of this subset of astrocytomas set them apart from the majority of astrocytic tumors? The answers to these questions are undoubtedly important as they may lead to better therapies for oligodendrogliomas and to new treatment strategies for anaplastic astrocytomas and glioblastomas.

In this thesis, two main approaches were taken to test the hypothesis that biochemical properties intrinsic to specific subtypes of glial tumors play a role in the resistance of these tumors to chemotherapeutic agents. The first approach was to examine biochemical properties inherent to the normal glial cell of origin. To accomplish this objective, the possibility of differential expression of drug resistance mechanisms in astrocytes and oligodendrocytes was investigated. In Chapter 3, astrocytes and mixed cultures of oligodendrocyte lineage cells were isolated from neonatal rat. The role of a DNA repair protein O⁶-methylguanine-DNA methyltransferase (MGMT) in the response of these cells to the clinically relevant alkylating agent, BCNU was examined. In Chapter 4, purified cultures of astrocytes, oligodendrocyte progenitors, and oligodendrocytes were obtained from neonatal rat. A panel of six drug resistance genes (MGMT, GST- μ , GST- π ,

p53, MDR, and MT) were probed for differential expression in these cells and the role of two mechanisms, MGMT and the glutathione-glutathione-S-transferase detoxification pathway, were examined in greater detail.

The second approach was to examine the impact of a specific genetic alteration, found commonly in astrocytic gliomas, on the response of otherwise normal astrocytes to BCNU. Mutations of the p53 gene, and subsequent loss of p53 function, are a common occurrence in astrocytic tumors. To determine the impact of the loss of p53 function on response of astrocytes to BCNU, astrocytes were isolated from TSG-p53 mice; this experimental system allowed me to examine the consequences of p53 gene dosage on drug resistance on an otherwise normal cellular background. In Chapter 5, the effect of p53 status on the viability of mouse astrocytes exposed to BCNU was examined and furthermore, whether this effect was drug-specific or cell-specific was investigated. In addition, the effect of BCNU treatment on the expression of three p53-regulated genes (mdm-2, WAF1, and GADD45) that may play a role in the cell cycle response to DNA damage was determined and the cell cycle response of astrocytes to DNA damaging treatments (ie. BCNU and radiation) was explored. In Chapter 6, the effect of p53 status on the ability of mouse astrocytes to undergo apoptosis when exposed to BCNU was examined. Furthermore, a semi-quantitative analysis of apoptosis in mouse cells in response to BCNU was performed and correlated with p53 status. The effect of BCNU treatment on the expression of two p53-regulated genes (bax and bcl-2) and one additional cell death gene (bcl-x) that may play a role in the apoptotic response to DNA damage also was investigated. Finally, in Chapter 7, the effect of p53 status on MGMT protein and mRNA levels was determined and the consequences of altered MGMT activity on the response of astrocytes to BCNU explored.

The remainder of Chapter 1 provides an overview of the present state of knowledge in this area of research; Chapter 2 describes the materials and methods employed throughout this thesis; and Chapter 8 summarizes the overall conclusions of this research.

1.2 GLIAL NEOPLASMS

1.2.1 Introduction

The most common human primary intracranial tumors are the gliomas. This family of brain tumors contains three main subtypes, each classified according to their presumed cell of origin: astrocytomas from astrocytic cells, oligodendrogliomas from oligodendrocytic cells, and ependymomas from ependymal or subependymal cells. Gliomas comprised of more than one cell type are termed mixed tumors and are named for both cell types; for example, oligoastrocytomas are marked by the presence of both oligodendrocytic and astrocytic cells. These cellular origins of gliomas are debated still since immunohistochemical staining for markers of more than one glial cell lineage has been observed simultaneously in histologically pure gliomas (Sarkar et al., 1988; Bishop and de la Monte, 1989; de la Monte, 1989; Kros et al., 1990; Cruz Sanchez et al., 1991; Cutarelli et al., 1991; Piepmeier et al., 1993). Most studies of the cellular origin of gliomas have focused on astrocytomas, oligodendrogliomas, and mixed oligoastrocytomas (Sarkar et al., 1988; Bishop and de la Monte, 1989; de la Monte, 1989; Cutarelli et al., 1991; Piepmeier *et al.*, 1993). Using immunochemical markers for normal glial cells. Bishop and de la Monte (1989) suggested that the majority of astrocytomas (approximately 75%) arise from the type 1 astrocyte lineage and the remainder from type 2 astrocytes of the O2A cell lineage (refer to section 1.4.1 for details of glial cell types). de la Monte (1989) utilized the same techniques to study oligodendrogliomas and mixed oligoastrocytomas and suggested that they arise from cells of the oligodendrocyte progenitor (O2A) lineage.

1.2.2 Pathology

Astrocytic gliomas can be divided broadly into two groups: the more common diffuse astrocytomas and a second group of relatively well circumscribed tumors comprised of the rarer subtypes such as the pilocytic juvenile astrocytoma, pleomorphic xanthoastrocytoma, and subependymal giant cell astrocytoma (Lopes *et al.*, 1993). Within the classification of diffuse astrocytomas, the World Health Organization (WHO)

designations recognize three histological subtypes: low grade astrocytoma, anaplastic astrocytoma, and glioblastoma multiforme (Lopes et al., 1993; Kleihues et al., 1993; Kleihues et al., 1995; Berger et al., 1996; Louis and Cavenee, 1997). Low grade astrocytomas display a tendency to diffusely infiltrate surrounding brain parenchyma, often resulting in the entrapment of pre-existing cells. Mitotic activity is absent but nuclear atypia may be present. Phenotypically, neoplastic cells may vary with respect to size, prominence, number of cell processes, and abundance of cytoplasmic glial filaments. Depending on the prevailing cell phenotype, astrocytomas can be further classified into two major variants, the more common fibrillary astrocytoma and the rarer gemistocytic astrocytoma. Anaplastic astrocytomas are distinguished from low grade tumors by increased cellularity, marked nuclear atypia and cellular pleomorphism, increased mitotic activity, and a moderate increase in vascular endothelial cell proliferation. The most malignant designation, glioblastoma multiforme, is used to denote tumors that are poorly differentiated and display an exaggeration of the features of anaplastic astrocytoma. However, tumor necrosis is the histological feature that distinguishes glioblastoma from an anaplastic astrocytoma.

Oligodendrogliomas and mixed oligoastrocytomas are diffuse tumors that are clinically and biologically the most closely related to the diffuse, fibrillary astrocytoma (Louis and Cavenee, 1997). Many of the histologic features used to grade astrocytomas also are used to grade oligodendrogliomas and mixed gliomas; these include cellularity, pleomorphism, mitotic activity, vascular changes, and necrosis (Berger *et al.*, 1996). Oligodendrogliomas have variable histologic appearances, the most common arrangement being a pseudolobular pattern of tumor cells surrounded by a vascular network (Lopes *et al.*, 1993; Berger *et al.*, 1996). In general, oligodendroglioma cells have round, regular, monotonous nuclei, distinct cellular borders with clearing of the cytoplasm and lack fibrillary cytoplasmic processes (Berger *et al.*, 1996). With increasing malignancy, oligodendrogliomas can become highly cellular and pleomorphic, approaching an appearance of glioblastoma multiforme with the presence of necrosis. Although some may classify necrotic oligodendrogliomas as glioblastoma multiforme, the preferred

classification is anaplastic oligodendroglioma (Berger *et al.*, 1996). Many, if not most, oligodendrogliomas contain a regional or intimate cellular mixture of astrocytoma cells. For the classification of mixed oligoastrocytoma, the proportion of the minor component is usually at least 10 to 25 percent.

Ependymomas are slowly growing tumors, usually well circumscribed. They have an almost epithelial appearance and are commonly calcified. Other histologic features may include perivascular pseudorosettes and ependymal rosettes (Berger *et al.*, 1996). The more aggressive anaplastic ependymoma is characterized by a marked increase in mitotic activity (Kleihues *et al.*, 1995).

1.2.3 Epidemiology

In 1997, an estimated 60 700 deaths from cancer will occur in Canada (National Cancer Institute of Canada, 1997). Of these, 1 380 are estimated to result from brain cancers, making brain tumor the ninth leading cause of cancer death among adult Canadians (National Cancer Institute of Canada, 1997). Cancers of the brain are the second most common cause of cancer death in children under 15 years of age (National Cancer Institute of Canada, 1996). The age-standardized incidence rate for Canadian males is estimated to be 8 per 100 000 people in 1997, with a slightly lower incidence rate of 6 per 100 000 in Canadian females (National Cancer Institute of Canada, 1997). It has been suggested that brain tumor incidence is on the rise in North America, especially among older patients (Greig et al., 1990; Mao et al., 1991). The possibility that improved diagnostic methods alone can account for these increases has been debated (Desmeules et al., 1992; Helseth, 1995). In comparison, the average annual percent change in age-standardized incidence in Canada (1985-1992) has been 0.7 for males and -0.1 for females, neither figure showing a statistically significant change (National Cancer Institute of Canada, 1997). In 1997, the estimated ratio of deaths to newly diagnosed cases of brain tumor in Canada is 0.64, outranked only by cancers of the esophagus, pancreas, and lung, and multiple myeloma (National Cancer Institute of Canada, 1997).

Gliomas represent the majority of primary intracranial brain tumors, with astrocytic tumors being the most common (Louis and Cavenee, 1997). Relative proportions of astrocytomas are age-dependent with indolent astrocytoma variants and lower grade diffuse fibrillary astrocytomas being more prevalent in children and adults under the age of 40 and glioblastomas predominating in patients over 45 years of age. The incidence of most gliomas, with the exception of glioblastoma multiforme, actually decreases with increasing age. Oligodendrogliomas, mixed oligoastrocytomas, and ependymomas comprise approximately 10-15% of all gliomas. Patients with high grade gliomas have an extremely poor prognosis; only 32% of patients with glioblastoma multiforme are alive one year after diagnosis and their five year survival rate is 7% or less (Holowaty *et al.*, 1996). Patients with low grade gliomas, on the other hand, often can survive five to ten years or longer (Leighton *et al.*, 1997).

1.2.4 Risk Factors

The epidemiology of glioma has provided some suggestions of possible environmental or occupational risk factors but few definitive observations have been made. A higher than expected incidence of brain tumors has been observed as a result of exposure to pesticides, herbicides, and fertilizers (Musicco *et al.*, 1982), and various substances prevalent in the petrochemical industry (Moss, 1985) and health professions (McLaughlin *et al.*, 1987). The credibility of these observations is difficult to determine; aside from an association between vinylchloride and gliomas, there are no common chemical or environmental factors among these observations (Moss, 1985). Electromagnetic fields also have been implicated as a possible risk factor for gliomas (Savitz and Loomis, 1995) but many studies do not support these findings (Wrensch *et al.*, 1993).

A possible inherent risk factor for glioma has been suggested recently by Bondy et al. (1996). In a case-control study, they demonstrated that lymphocytes of glioma patients were more sensitive to the induction of DNA strand breaks by γ -irradiation. Perhaps an increased sensitivity to this and other mutagens could be a predisposing risk factor for glioma.

Clues to a hereditary basis for glioma have come from the study of genetic and hereditary syndromes associated with an increased incidence of glial tumors (Bondy *et al.*, 1996). These syndromes include Li-Fraumeni syndrome, tuberous sclerosis, neurofibromatosis types 1 and 2, and Turcot's syndrome. In addition, multiple studies have described familial aggregations of brain tumors in the absence of a known predisposing syndrome (Ikizler *et al.*, 1992; Bondy *et al.*, 1996). These studies, although inconclusive, suggest that a role may exist for inherited genetic susceptibility in the genesis of brain tumors.

1.2.5 Molecular Tumor Biology

In 1976, Nowell proposed a model whereby the *in vivo* evolution of tumor cell populations occurs through the stepwise accumulation of growth-advantageous genetic changes in cells that were initially of normal genotype and phenotype. In general, the more growth-advantageous mutations that accumulated, the more aggressive or malignant would be the tumor. It is now known that the genetic changes that occur during the evolution of tumor cells can be grouped into two categories: those involving the activation through overexpression and/or mutation of gene products conferring growth advantage (ie. oncogenes) and those involving the functional loss or inactivation of gene products having a negative growth-regulatory function (ie. tumor suppressor genes) (James and Collins, 1993). Gliomas are no exception to this model and much work into the molecular biology of brain tumors has occurred in the last few years.

Of all glial tumors, the molecular biology of astrocytomas has received the most attention. One of the earliest genetic alterations to occur in astrocytomas is mutation of the *p53* gene or loss of chromosome 17p (*p53* is localized to 17p13). This alteration is found in approximately 25% of astrocytic gliomas of all malignancy stages (James *et al.*, 1988; el Azouzi *et al.*, 1989; James *et al.*, 1989; Fults *et al.*, 1989; Nigro *et al.*, 1989; Chung *et al.*, 1991; Hayashi *et al.*, 1991; Mashiyama *et al.*, 1991; Frankel *et al.*, 1992; Fults *et al.*, 1992; von Deimling *et al.*, 1992; Non Deimling *et al.*, 199

1992a; Aka et al., 1993; del Arco et al., 1993; Hunter et al., 1993; Louis et al., 1993; Newcomb et al., 1993; Wu et al., 1993; Bello et al., 1994; Koga et al., 1994; Kraus et al., 1994; Louis, 1994; Rasheed et al., 1994). Overexpression of growth factors and their receptors are also a common feature of astrocytomas. Changes in expression of platelet-derived growth factor (PDGF), fibroblast growth factors (FGFs), insulin-like growth factors I and II (IGF-I and IGF-II), and vascular endothelial growth factor (VEGF) have been well documented (Antoniades et al., 1992; Brem et al., 1992; Gillaspy et al., 1992; Hermanson et al., 1992; Plate et al., 1992; Trojan et al., 1993; Plate and Risau, 1995; Westermark et al., 1995). Of particular interest is PDGF; PDGF ligands and receptors are expressed at approximately equal levels in all grades of astrocytoma, suggesting that such overexpression is important in the initial stages of tumor formation (Louis and Cavenee, 1997).

One of the hallmarks of astrocytomas is their tendency to infiltrate the surrounding brain. These invasive abilities often are apparent in low grade as well as high grade astrocytomas, suggesting that the invasive phenotype is another feature acquired early in glial tumorigenesis. A variety of cell surface molecules are differentially expressed in astrocytomas, including CD44 glycoproteins, gangliosides, and integrins (Louis and Cavenee, 1997). In addition, many of the growth factors expressed in astrocytic tumors, such as FGF, EGF, and VEGF, have been shown to stimulate tumor cell migration (Pilkington, 1994).

A less common genetic change in astrocytomas involves chromosome 22. Allelic loss of chromosome 22q has been observed in about one quarter of astrocytomas, suggesting the presence of a glioma tumor suppressor gene within that region (James *et al.*, 1988).

The transition of astrocytoma to anaplastic astrocytoma is accompanied by a marked increase in malignant behavior and a number of molecular abnormalities have been associated with this tumor of higher grade. The most prevalent changes involve a critical cell cycle regulatory complex that includes the p16, CDK4, cyclin D1, and retinoblastoma (Rb) proteins (Venter *et al.*, 1991; Henson *et al.*, 1994; Jen *et al.*, 1994;

Schmidt *et al.*, 1994; Arap *et al.*, 1995; Collins, 1995; Nishikawa *et al.*, 1995; Sonoda *et al.*, 1995; James and Olson, 1996; Ueki *et al.*, 1996; Zhang *et al.*, 1996; Dirks *et al.*, 1997; Louis and Cavenee, 1997). Individual components of this pathway can be altered in up to half of anaplastic astrocytomas and in the majority of glioblastomas. A second common abnormality is allelic loss on 19q; loss of heterozygosity at this locus is observed in up to 40% of anaplastic astrocytomas and glioblastomas (von Deimling *et al.*, 1992c). This region has been suggested to contain a tumor suppressor gene important for progression of astrocytomas (Seizinger *et al.*, 1991). This gene may be unique to gliomas and is involved in all three of the major cerebral glial tumors (ie. astrocytoma, oligodendroglioma, and oligoastrocytoma) (Seizinger *et al.*, 1991).

Glioblastomas are the most malignant form of astrocytomas and display their own complement of genetic abberations. As mentioned above, the majority, if not all, glioblastomas have alterations in the p16-CDK4-cyclin D1-Rb pathway of cell cycle control. In addition, chromosome 10 is lost frequently in glioblastomas, occurring in 60 to 95% of glioblastomas but only rarely in anaplastic astrocytomas (James *et al.*, 1988; Fults *et al.*, 1992; Rasheed *et al.*, 1992; von Deimling *et al.*, 1992b; Bello *et al.*, 1994; von Deimling *et al.*, 1995). Attempts to identify possible tumor suppressor genes for glioma on chromosome 10 have been hampered by the fact that, in most cases, the entire chromosome is lost but evidence suggests that loss of heterozygosity involves at least three distinct regions (Rasheed *et al.*, 1992; Karlbom *et al.*, 1993). Recently, Steck *et al.* (1997) have identified one candidate tumor suppressor gene, *MMAC1*, at chromosome 10q23.3.

Since one of the hallmarks of glioblastoma is marked vascular endothelial cell proliferation, the upregulation of a number of growth factors and their receptors has been investigated (Louis and Cavenee, 1997). A paracrine mechanism has been suggested whereby growth factors such as VEGF and PDGF are frequently overexpressed in tumor cells while their receptors, flk-1 and flt-1 for VEGF and the PDGF- β receptor for PDGF, are upregulated on endothelial cells (Hermanson *et al.*, 1992; Plate *et al.*, 1992). VEGF and its receptors, in particular, appear to play a key role in the angiogenesis of

glioblastoma (Plate et al., 1992; Plate et al., 1994; Plate and Risau, 1995).

The most frequently amplified oncogene in highly malignant astrocytic tumors is the epidermal growth factor receptor (*EGFR*) gene, being amplified in approximately 40% of glioblastomas but in few anaplastic astrocytomas (Ekstrand *et al.*, 1991; von Deimling *et al.*, 1992b; Wong *et al.*, 1992; Kordek *et al.*, 1995; Schwechheimer *et al.*, 1995; James and Olson, 1996). EGFR overexpression is almost always accompanied by a loss of chromosome 10 in glioblastomas, suggesting that these alterations may be integral to malignant progression in some astrocytic tumors (Ekstrand *et al.*, 1991; von Deimling *et al.*, 1992b; Nishikawa *et al.*, 1994; von Deimling *et al.*, 1995). Genes less commonly amplified in glioblastomas include N-*myc*, *gli*, PDGF- α receptor, c-*myc*, *myb*, K-*ras*, c-*fos*, *CDK4*, and *mdm*-2 (Orian *et al.*, 1992; Collins, 1995; Louis and Cavenee, 1997).

Subsets of glioblastoma multiforme can be classified on the basis of molecular genetic analysis (von Deimling *et al.*, 1993; van Meyel *et al.*, 1994; Kordek *et al.*, 1995; von Deimling *et al.*, 1995; Louis and Cavenee, 1997). For instance, loss of chromosome 17p and *p53* mutations occur in tumors with overexpression of the PDGF- α receptor (Hermanson *et al.*, 1996) and loss of chromosome 10 often is found in association with EGFR overexpression (von Deimling *et al.*, 1992b). However, loss of chromosome 17p almost never occurs in glioblastomas with EGFR overexpression: approximately one-third of glioblastomas display *p53*/chromosome 17p alterations, one-third have EGFR amplification, and one-third do not exhibit either change (von Deimling *et al.*, 1993). Of significant interest are the findings that glioblastomas with loss of chromosome 17p occur in younger patients and are hypothesized to arise via stepwise progression while those characterized by EGFR overexpression occur in older patients and are typically *de novo* tumors (von Deimling *et al.*, 1993; van Meyel *et al.*, 1994; Rasheed *et al.*, 1994; von Deimling *et al.*, 1995).

The most common molecular change found in oligodendrogliomas and oligoastrocytomas is the allelic loss of chromosomes 1p and 19q, affecting 40 to 80% of these tumors (Seizinger *et al.*, 1991; von Deimling *et al.*, 1992c; Reifenberger *et al.*,

1994; Kraus *et al.*, 1995). These regions are lost in low grade as well as anaplastic oligodendrogliomas and oligoastrocytomas at almost equal frequency, suggesting that these loci are important early in oligodendroglial tumorigenesis. Chromosome 1p and 19q losses are closely associated, indicating that tumor suppressor genes located in these regions might be involved in biologically distinct pathways, each of which must be altered for tumorigenesis (Reifenberger *et al.*, 1994; Kraus *et al.*, 1995). In contrast to astrocytomas, oligodendrogliomas rarely harbor p53 mutations (Reifenberger *et al.*, 1994) but often accumulate wild type p53 protein (Pavelic *et al.*, 1994).

Ependymomas are a clinically diverse group of gliomas. Progression to tumors characteristic of glioblastoma are rare and have not been studied by molecular biology techniques (Louis and Cavenee, 1997). The most common genetic alteration in ependymomas is chromosome 22q loss (James *et al.*, 1990; Ransom *et al.*, 1992). In contrast to astrocytomas, mutations in the p53 gene do not occur (Ohgaki *et al.*, 1991).

1.2.6 Treatment

1.2.6.1 Surgery

No other modality of cancer treatment can reduce tumor bulk as quickly as surgery. Well circumscribed tumors such as ependymomas and juvenile pilocytic astrocytomas often are amenable to gross total resection. Unfortunately, the infiltrative nature of most other gliomas precludes complete resection of the tumor in most instances and thus, a high rate of recurrence exists. When surgical cure is not possible, the main goals of surgery become tumor size reduction and consequent decompression of the brain (Levin *et al.*, 1997). Numerous retrospective studies have supported the concept that an aggressive resection has a positive impact on survival as well as quality of life (Berger *et al.*, 1996). In addition to the treatment goals of surgery, surgical biopsies are crucial for pathologic evaluation and treatment recommendations for all gliomas.

1.2.6.2 Radiation Therapy

Postsurgery external beam radiation for malignant gliomas has proven effective

in prolonging patient survival in several prospective randomized controlled trials and has become the standard therapy for glial tumors (Berger *et al.*, 1996). The treatment of low grade astrocytomas and oligodendrogliomas with radiation has been more controversial. There have been no randomized studies comparing the outcome of patients treated by surgery alone with those who received surgery plus radiation therapy and thus, treatment decisions are based on information obtained from retrospective reports (Berger *et al.*, 1996). It is agreed generally that patients with neurologic impairment, tumor progression, or malignant transformation should receive radiation therapy. The techniques of interstitial brachytherapy (Loeffler *et al.*, 1990) and stereotactic radiosurgery (Loeffler *et al.*, 1992) have been used to deliver intense focal radiation to the tumor. Initially, these methods appeared promising but have since been questioned due to the finding that most of the increase in patient survival could be traced to patient selection bias (Curran, Jr. *et al.*, 1992; Florell *et al.*, 1992).

1.2.6.3 Chemotherapy

Despite intensive research, attempts to develop effective chemotherapy regimens for gliomas have met with limited success. Early studies of the Brain Tumor Study Group (BTSG) found that a greater proportion of patients with malignant glioma survived 18 months or longer when chemotherapy was added to surgery and radiotherapy (Walker *et al.*, 1980; Green *et al.*, 1983). Although median survival did not change, 15% of patients receiving adjuvant chemotherapy were long-term survivors compared to 5% following radiation alone. Stenning *et al.* (1987) obtained complementary results from a meta-analysis of eight randomized trials reported between 1976 and 1985. They also described a significant, although limited, benefit to the addition of chemotherapy to radiation therapy; the 1-year and 2-year survival rates were increased 9 and 3.5 percent respectively. These results were interpreted to mean that a small subset of patients benefit from adjuvant chemotherapy but the question of which patients benefit remained.

More recently, some groups have investigated possible factors responsible for improved patient prognosis in response to adjuvant chemotherapy. Nelson *et al.* (1988),

in a re-evaluation of a randomized trial by the Radiation Therapy Oncology Group (RTOG) and Eastern Cooperative Oncology Group (ECOG), reported that adjuvant BCNU enhanced overall and proportionate survival at two years in patients with anaplastic glioma and glioblastoma. Patients aged 40-60 survived significantly longer with BCNU treatment; those aged less than 40 appeared to benefit; and those older than 60 showed no improvement in survival. A possible role for patient age in prognosis of glioma has been supported by others (Chang *et al.*, 1983; Grant *et al.*, 1995). In a meta-analysis of 16 randomized trials, Fine *et al.* (1993) concluded that adjuvant chemotherapy enhanced proportionate survival at one and two years independent of tumor grade. Finally, DeAngelis *et al.* (1996) re-examined two randomized trials by the BTSG and noted that adjuvant BCNU improved proportionate survival at 18 months across all major prognostic factors, including age less than 65, intermediate or high tumor grade, symptoms, and performance status levels. In conclusion, these studies suggested that prognostic factors for survival, with the possible exception of younger age, do not predict chemotherapeutic benefit.

Studies had largely ignored the possibility that specific histological subtypes of glioma might differ in their response to chemotherapy. Significant progress was made when Cairncross and Macdonald (1988) reported that anaplastic oligodendrogliomas responded to a chemotherapy regimen of procarbazine, CCNU, and vincristine (PCV). Subsequently, Glass *et al.* (1992) and Kim *et al.* (1996) observed that mixed oligoastrocytomas responded frequently to PCV. In addition, Mason *et al.* (1996) noted that symptomatic low-grade oligodendrogliomas often responded to PCV. At the current time, it is not known whether adjuvant PCV chemotherapy will result in prolonged patient survival but a randomized controlled trial for anaplastic pure and mixed oligodendrogliomas is being carried out by the RTOG in an attempt to answer this question.

1.2.7 Summary

Gliomas are the most common brain tumors of adults. The three main subtypes

of primary glial tumors are astrocytomas, oligodendrogliomas, and ependymomas. These tumors are a significant cause of cancer death; in 1997, the estimated ratio of deaths to newly diagnosed cases of brain tumors in Canada is 0.64, outranked only by cancers of the esophagus, pancreas, lung, and multiple myeloma (National Cancer Institute of Canada, 1997). Although there is little evidence for well established environmental or behavioral risk factors for glioma, there does appear to be a small population that may be predisposed to glioma due to inherited susceptibility. More is known about the molecular biology of brain tumors. Genetic alterations have been identified that can be linked to a specific pathologic subtype or malignancy stage of glioma. Finally, due to the poor prognosis of glial tumors, much emphasis has been placed on treatment improvement. Surgery can reduce tumor size and improve quality of life through the resultant decompression of the brain but the invasiveness of gliomas generally prevents total surgical resection. Conventional radiation and chemotherapy can offer temporary benefit but are seldom curative. Present and future studies designed to investigate why some gliomas benefit from treatment while others are refractory will lead to improved treatment regimens and patient prognosis.

1.3 EXPERIMENTALLY INDUCED BRAIN TUMORS

Since 1966, alkylating agents, principally the N-nitroso compounds, have been used to induce central nervous system (CNS) tumors in rodents (Druckrey *et al.*, 1966). The features of this experimental system as reviewed by Lantos (1986) are as follows: 1) the incidence, distribution, latency, and histology of these tumors are influenced by the species, age, and sex of the animals and by the dose and mode of application of the carcinogen; 2) the neoplasms induced by nitrosoureas are not distributed at random in the nervous system, but develop at certain favoured sites; most notably, the periventricular areas and the subcortical white matter in the cerebral hemispheres; 3) tumor development is often multicentric and more than one lesion can occur in one animal; 4) histologically, the cerebral tumors are most often gliomas; 5) the latency period is long, with cerebral gliomas developing in 245 \pm 80 days. It was found that, although MNU had to be administered to adult animals in small repeated doses in order
to produce a high tumor yield, ENU could induce an almost 100% incidence of neural neoplasms in the offspring of mothers injected during the second half of gestation and in neonatal rats using a single dose (Druckrey *et al.*, 1966). Consequently, most studies to date have utilized the transplacental administration of ENU in order to chemically induce gliomas in rat.

Although the pathology of experimental neurogenic tumors induced by transplacental administration of ENU has demonstrated a wide range of morphological variability (Wechsler et al., 1969), it is interesting to note that the cerebral glial tumors were more often mixed and pure oligodendrogliomas than astrocytomas (Wechsler et al., 1969; Grossi-Paoletti et al., 1970; Koestner et al., 1971; Lantos, 1972; Druckrey, 1973; Lantos and Cox, 1976; Schiffer et al., 1978; Usuki et al., 1992). These findings led to the hypothesis that the two most susceptible targets in the brain to carcinogenic action were the primitive cells of the subependymal plate and oligodendroglial cells (Lantos, 1972). To investigate this hypothesis further, Burger et al. (1988) removed the forebrains from transplacentally exposed fetal rats one day after treatment and transplanted them into the brains of adult animals. Unexpectedly, the histological spectrum of neoplasms was reduced to a single tumor type, oligodendrogliomas. They reasoned the selective induction of oligodendrogliomas indicated that neoplastic transformation in the nervous system could occur in a differentiated glial cell or in a precursor cell committed to oligodendrocytic differentiation, and that transformation of a pluripotential stem cell was not necessary.

1.4 GLIAL CELL LINEAGES IN THE MAMMALIAN BRAIN

1.4.1 Glial Cells In Vitro

In 1980, McCarthy and de Vellis developed a method for separating rat astrocytes from O2A progenitors and their derivatives based on differential adhesion to tissue culture plates. In the years since, much study has gone into the development, differentiation, and cellular lineage of these glial cells in culture (Abney *et al.*, 1983; Raff *et al.*, 1983b; Raff *et al.*, 1983a; Eccleston and Silberberg, 1984; Raff *et al.*, 1984b; Raff et al., 1984a; Miller et al., 1985; Raff et al., 1985; Williams et al., 1985; ffrench Constant and Raff, 1986; Kennedy and Fok Seang, 1986; Lillien et al., 1988; Bogler et al., 1990; Lillien and Raff, 1990; Agresti et al., 1991; Dutly and Schwab, 1991; Knapp, 1991; Madarasz et al., 1991; Mozell and McMorris, 1991; Wolswijk et al., 1991; Louis et al., 1992; Wolswijk and Noble, 1992; McKinnon et al., 1993) and the subject has been reviewed extensively (Miller et al., 1989; Raff, 1989; Cameron and Rakic, 1991; Goldman and Vaysse, 1991; Noble, 1991; Noble et al., 1991; Skoff and Knapp, 1991; Linskey and Gilbert, 1995). Two studies in particular advanced the ability to grow pure cultures of individual glial subtypes in culture: 1) Raff et al. (1983b) demonstrated that serum deprivation promoted oligodendrocytic differentiation in cultures of O2A progenitor cells and 2) Bogler et al. (1990) found that O2A progenitors proliferated in the presence of PDGF and basic fibroblast growth factor (bFGF) and differentiated into pure oligodendrocytes when bFGF was removed from the medium. With this knowledge, glial cell culture techniques were further refined by Barres et al. (1992) and it is now possible to culture large numbers of pure populations of astrocytes.

Two major branches of the mammalian glial cell lineage can be distinguished by the presence or absence of a surface marker detected by an antibody called A2B5 (Raff, 1989). A2B5 negative glial progenitors give rise to astrocytes (called type 1 astrocytes by some). A2B5 positive precursor cells, called oligodendrocyte or O2A progenitors, give rise to oligodendrocytes. In tissue culture, O2A progenitors also give rise to a second astrocytic cell type called type 2 astrocytes. The existence of type 2 astrocytes *in vivo* has been debated extensively and results remain inconclusive (Miller and Raff, 1984; Miller *et al.*, 1985; Miller *et al.*, 1989; Cameron and Rakic, 1991; Goldman and Vaysse, 1991; Noble, 1991; Skoff and Knapp, 1991; Espinosa de los Monteros *et al.*, 1993); thus, these cells are not discussed further. Astrocytes, oligodendrocyte progenitors, and oligodendrocytes have distinct morphologies and can be detected in tissue sections and cell culture using immunochemical techniques. Astrocytes are A2B5 negative, glial fibrillary acidic protein (GFAP) positive, and galactocerebroside (GC) negative.

oligodendrocyte progenitors, and oligodendrocytes from rat brain.

Oligodendrocyte progenitors are A2B5 positive, GFAP negative, and GC negative. Oligodendrocytes are A2B5 negative, GFAP negative, and GC positive [Note: immature oligodendrocytes express A2B5 but this marker is lost as oligodendrocytes differentiate fully.]

1.4.2 Glial Cells In Vivo

Neuroglia in the brain function as supportive cells and comprise approximately one half of the mammalian brain by mass (Marieb and Mallatt, 1997). Each glial cell type has its own specialized role. The most abundant cells are the astrocytes, stellate cells of diverse morphology and function. The main role of these cells is to stabilize the neuronal microenvironment: they remove neurotransmitters (Marieb and Mallatt, 1997) and potassium ions (Newman, 1985; Ransom and Sontheimer, 1992) from the extracellular space following neuronal activity; remove neurotoxic molecules (Eddleston and Mucke, 1993); participate in carbohydrate and lipid metabolism (Bignami and Dahl, 1994); and synthesize neurotrophic factors and cytokines (Stockli et al., 1991; Yoshida and Gage, 1992; Lin et al., 1993). In addition, Janzer and Raff (1987) have provided evidence that astrocytes are responsible for inducing capillary and venule endothelial cells in the CNS to form the blood-brain barrier. Oligodendrocytes are the myelin producing cells of the CNS and each oligodendrocyte can form myelin sheaths around multiple axons (Marieb and Mallatt, 1997). These insulating layers of myelin facilitate rapid action potentials and the conservation of metabolic energy in neurons (Morell and Norton, 1980). Ependymal cells form a single layer of ciliated epithelium that lines the ventricles of the brain and the central canal of the spinal cord; specialized ependymal cells in the choroid plexus regulate the production of cerebrospinal fluid (Del Bigio. 1995; Marieb and Mallatt, 1997). These cells constitute a fairly permeable interface between the cerebrospinal fluid and the tissue fluid that bathes the cells of the CNS and function as a protective and regulatory barrier (Del Bigio, 1995; Marieb and Mallatt, 1997). Finally, microglia are phagocytic cells that migrate through the CNS and remove foreign and degenerated material (Marieb and Mallatt, 1997). As microglia are not fixed glial cells of the brain and do not contribute to the formation of the gliomas discussed

1.5 MECHANISMS OF RESISTANCE TO BCNU

1.5.1 Introduction

The nitrosoureas (BCNU, CCNU, and methyl-CCNU being the most common) are lipid-soluble drugs capable of penetrating into the CNS for treatment of intracranial tumors. Of these, BCNU has been the most frequently studied chemotherapy agent in the treatment of glial tumors (Chang *et al.*, 1983; Walker *et al.*, 1978; Walker *et al.*, 1980). Other chemotherapeutic agents studied either alone or in combination displayed similar or less effective activity and often were more toxic than BCNU alone (Chang *et al.*, 1983; Green *et al.*, 1983; Leibel *et al.*, 1991; Nelson *et al.*, 1988; Shapiro *et al.*, 1989). For these reasons, the combination of conventional radiotherapy and BCNU became a standard treatment regimen for gliomas (Deutsch *et al.*, 1989). One exception to this finding has been the combination of procarbazine, CCNU, and vincristine (PCV) which in one study resulted in an improvement of outcome for patients with anaplastic gliomas (Levin *et al.*, 1990).

BCNU is highly unstable in aqueous medium, especially in the presence of plasma proteins, and its decomposition results in a large number of products (reviewed in Wiencke and Wiemels, 1995). The first stage of decomposition results in two products (Montgomery *et al.*, 1967). One of these is chloroethyldiazohydride, which gives rise to the reactive chloroethyl carbonium ion that is thought to be responsible for the alkylation of DNA. In addition, further degradation of chloroethyldiazohydride can lead to significant amounts of several volatile products, including acetaldehyde and chloroethanol. The second product from the first stage of decomposition is chloroethyl isocyanate. It is believed that protein modifications are the result of the carbamoylating activity of this compound. Metabolic deactivation of the chloroethylnitrosoureas is through either the cytochrome P-450 enzymes or by the more prevalent glutathione-*S*-transferases.

The main anti-tumor activity of BCNU is considered by most to be the formation

of DNA interstrand crosslinks (Colvin *et al.*, 1976; Erickson *et al.*, 1978; Erickson *et al.*, 1980; Bodell *et al.*, 1984; Beith *et al.*, 1997) and the initial, rapidly formed DNA modification O⁶-chloroethyl guanine is thought to be necessary for the formation of these crosslinks (Erickson *et al.*, 1980). After the initial chloroethylation at the O⁶ position of guanine, an intramolecular rearrangement occurs leading to an ethane bridge between the guanine and its partnered cytosine (Ludlum, 1990). The resultant crosslink has been identified as $1-[N^3-deoxycytidyl], 2-[N^1-deoxyguanosinyl]-ethane (Tong$ *et al.*, 1982).

1.5.2 O⁶-Methylguanine-DNA Methyltransferase

One of the best-studied mechanisms of nitrosourea resistance is the DNA repair protein O⁶-methylguanine-DNA methyltransferase (MGMT), also called O⁶-alkylguanine-DNA alkyltransferase (AGT) (reviewed extensively by Yarosh, 1985; Lindahi *et al.*, 1988; Pegg, 1990a; Pegg, 1990b). This protein removes adducts from the O⁶-position of guanine, one type of structural alteration formed when alkylating agents react with double stranded DNA. Of particular interest to BCNU resistance, MGMT repairs the O⁶chloroethyl adducts produced by the chloroethylnitrosoureas, preventing the ultimate cytotoxic lesions, guanine-cytosine interstrand DNA crosslinks (refer to section 1.5.1). MGMT acts by covalent transfer of the alkyl group from the modified base to a cysteine residue within the active site of the protein, restoring the integrity of the DNA and irreversibly inactivating the protein. Due to the stoichiometry of the reaction, the repair capacity of the cell depends on the number of active MGMT molecules present at the time of cytotoxic insult.

The cytotoxic and carcinogenic properties of the nitrosoureas appear related to MGMT activity: cells rendered MGMT positive by DNA transfection or transgenic expression become resistant to the nitrosoureas (Ding *et al.*, 1985; Kaina *et al.*, 1991; Dumenco *et al.*, 1993; Fritz *et al.*, 1993); resistant cells depleted of MGMT by pretreatment with O⁶-methylguanine or O⁶-benzylguanine are sensitized to the nitrosoureas (Dolan *et al.*, 1985; Yarosh *et al.*, 1986; Dolan *et al.*, 1990a; Dolan *et al.*, 1991; Mitchell *et al.*, 1992; Baer *et al.*, 1993; Felker *et al.*, 1993;

Sarkar *et al.*, 1993; Marathi *et al.*, 1994; Wedge and Newlands, 1996); and cells susceptible to neoplastic transformation by nitrosoureas express low levels of MGMT (Swenberg *et al.*, 1982; Wani *et al.*, 1992).

MGMT is the mechanism of drug resistance in human glial tumors and cells that has received the most attention to date (Wiestler et al., 1984; Bodell et al., 1986; Aida and Bodell, 1987; Schold, Jr. et al., 1989; Frosina et al., 1990; Citron et al., 1991; Ostrowski et al., 1991; Friedman et al., 1992b; Hotta et al., 1993; Mineura et al., 1993; Silber et al., 1993; Hotta et al., 1994; Bobola et al., 1995; Citron et al., 1995; Belanich et al., 1996; Mineura et al., 1996a; Mineura et al., 1996b; Beith et al., 1997). Most studies agree that one of the mechanisms responsible for resistance of gliomas to the chloroethylnitrosoureas is increased repair of O^6 -alkylguanine products in DNA by MGMT (Bodell et al., 1986; Aida and Bodell, 1987; Friedman et al., 1992b; Mineura et al., 1993; Hotta et al., 1994; Mineura et al., 1996a; Beith et al., 1997), although some studies provide evidence that BCNU resistance is multifactorial (Hotta et al., 1993; Bobola et al., 1995) and that MGMT makes only a modest contribution to resistance (Bobola et al., 1995). Additional studies have demonstrated a shorter time to treatment failure and death in patients with high MGMT activity, supporting the concept that MGMT correlates with resistance to the nitrosoureas (Belanich et al., 1996). A wide range of MGMT activity has been found in brain tumor samples (Frosina et al., 1990; Hotta et al., 1994; Citron et al., 1995); one study suggested that activity is higher in low grade gliomas than in high grade tumors (Mineura et al., 1996b), one suggested the reverse (Frosina et al., 1990), while still others have found no significant difference (Mineura et al., 1993; Hotta et al., 1994). In addition, Wiestler et al. (1984) and Frosina et al. (1990) found lower levels of MGMT activity in oligodendrogliomas than astrocytomas but this also has not been a universal finding (Mineura et al., 1993; Mineura et al., 1996a; Silber et al., 1993). Low MGMT activity has been detected in normal human brain tissue (Grafstrom et al., 1984; Wiestler et al., 1984; Silber et al., 1993; Silber et al., 1996) but MGMT activity has not been studied in purified populations of glial cells.

1.5.3 Glutathione and Glutathione-S-Transferases

Tumor cell resistance to alkylating agents has been associated with changes in cellular metabolic systems that facilitate detoxification of cytotoxic drugs (Beck and Dalton, 1997), such as cytochrome P-450 enzymes (Hill *et al.*, 1975; Grant and Ironside, 1995) and the glutathione-glutathione-*S*-transferase pathway. Of these systems, the majority of investigations of nitrosourea resistance have concentrated on the effect of alterations in levels of glutathione and its associated glutathione-*S*-transferases (GSTs). The tripeptide glutathione plays a role in detoxification by non-enzymatic conjugation of alkylating agents and other electrophilic molecules with its thiolate moiety (Russo *et al.*, 1986a). This conjugation of drug with glutathione results in a more water soluble and less toxic product (Beck and Dalton, 1997). GSTs are a family of enzymes that catalyze the conjugation of these drugs, thereby enhancing the rate of detoxification. GST isoenzymes are broadly grouped by their isoelectric points into three classes: GST- π , acidic; GST- μ , neutral; and GST- α , basic (Mannervik *et al.*, 1992). Each isoenzyme class exhibits different but potentially overlapping substrate specificities.

Levels of glutathione and GSTs have been implicated in resistance to a variety of alkylating agents including cyclophosphamide (McGown and Fox, 1986; Friedman *et al.*, 1992a), nitrogen mustards (Ahmad *et al.*, 1987; Evans *et al.*, 1987), cisplatin (Teicher *et al.*, 1987; Moore *et al.*, 1989), melphalan (Friedman *et al.*, 1989) and the chloroethylnitrosoureas (Ali-Osman *et al.*, 1989; Ali-Osman, 1989; Smith *et al.*, 1989; Ali Osman *et al.*, 1990; Hara *et al.*, 1993). Ali-Osman *et al.* (1989) have reported increased resistance to nitrosoureas *in vitro* following the addition of glutathione and Russo *et al.* (1986b) were able to show an increased resistance to melphalan after increasing glutathione levels with oxothiazolidine-4-carboxylate (OTZ), an intracellular cysteine delivery system. Others have demonstrated decreased resistance to alkylating agents upon reduction of glutathione levels with buthionine sulfoximine (BSO), an inhibitor of γ -glutamyl cysteine synthetase (Green *et al.*, 1984; Somfai-Relle *et al.*, 1984b; Friedman *et al.*, 1989). In addition, inhibition of the GST enzymes with ethacrynic acid or piriprost has been found to reverse resistance to

alkylating agents (Tew et al., 1988).

One interesting property of the chloroethylnitrosoureas in regard to glutathione is the ability to inactivate the enzyme glutathione reductase through the specific carbamoylation of the enzyme by the decomposition products, 2-chloroethyl isocyanate and cyclohexyl isocyanate (Babson and Reed, 1978). Babson *et al.* (1981) also were able to demonstrate that this effect resulted in a 70% decrease in glutathione levels. In contrast, Deneke *et al.* (1992) found that although BCNU treatment was effective at reducing glutathione reductase activity, a concentration-dependent increase in glutathione levels was observed after 24 hr of exposure and that this increase correlated with increases in cysteine uptake. It was suggested that in addition to the ability of BCNU to inactivate glutathione reductase, BCNU also may stimulate cysteine uptake and subsequent glutathione synthesis and that these two effects are perhaps totally independent of each other (Deneke *et al.*, 1992).

Glutathione and GST levels have been studied in human glioma tissues and cell lines and support for a role in resistance to alkylating agents has been found (Ali-Osman *et al.*, 1989; Friedman *et al.*, 1989; Ali Osman *et al.*, 1990). Studies by Strange *et al.* (1992) and Grant *et al.* (1995) demonstrated that the expression of GST- π was the predominant subclass expressed in astrocytic tumor cells, although GST- α and GST- μ also were expressed in some tumors. The localization of the placental form of GST- π , expressed normally during development, also has been demonstrated in glioma cell lines (Hara *et al.*, 1993). These findings are of particular interest since Ali-Osman *et al.* (1990) showed that only the expression of the GST- π subclass correlated significantly with BCNU resistance in human astrocytoma cell lines. Expression of GSTs also have been studied in normal human brain (Carder *et al.*, 1990; Strange *et al.*, 1992) and GST- μ subclasses being detected less frequently. Carder *et al.* (1990) also demonstrated expression of GST- π in normal human astrocytes.

1.5.4 Nucleotide-Excision Repair Proteins

A system that has received less attention as a mechanism of resistance to the chloroethylnitrosoureas involves nucleotide-excision repair (NER). It is known that interstrand crosslinks are removed from human DNA and based on evidence from bacterial systems, it is thought that these crosslinks are eliminated by the combined actions of the excision repair and recombination systems (Chaney and Sancar, 1996). NER removes bulky lesions from DNA in the form of 27- to 29-nucleotide-long oligomers by: incising the damaged strand on both sides of the lesion; excising the segment of damaged DNA with an excinuclease; filling the gap with DNA replication proteins; and ligating the repair patch (Sancar, 1994; Sancar, 1995; Chaney and Sancar, 1996). In addition, NER plays a back-up role for the repair of non-bulky lesions, including O^6 -methylguanine residues (Chaney and Sancar, 1996). It is not surprising, therefore, that evidence suggests the NER pathway is involved in repair of at least some of the cytotoxic lesions formed by the chloroethylnitrosoureas (Bronstein *et al.*, 1992; Wu *et al.*, 1992; Cappelli *et al.*, 1995); a direct effect on resistance to the chloroethylnitrosoureas is less well defined, however.

The investigation into a possible role for NER in glial tumors has been limited to studies of the excision repair cross-complementing (ERCC) proteins. ERCC1, a member of the incision complex, and ERCC2, a member of the helicase complex, are both mapped to chromosome 19q, a genomic region of common abnormality in glioma (refer to section 1.2.5). Abnormalities in copy number of ERCC1 and ERCC2 have been demonstrated in glial tumors (Liang *et al.*, 1995). In addition, although mRNA levels of ERCC1 and ERCC2 are tightly coordinated in non-malignant brain tissue, it has been suggested that this coordination is lost in malignant brain tissues (Dabholkar *et al.*, 1995) and that progressive loss of coordination correlated with increased malignancy (Dabholkar *et al.*, 1996). The role that these changes may play in resistance of glial tumors to the chloroethylnitrosoureas has not been examined.

Although the base mismatch repair system has been shown to remove some of the bases modified by the chloroethylnitrosoureas, these lesions are generally not considered cytotoxic and therefore, it is not likely that this system contributes substantially to resistance to these agents (Cappelli *et al.*, 1995). In addition to being able to repair DNA damage, acquired resistance to monofunctional alkylating agents such as MNU and *N*-methyl-*N*-nitro-*N*-nitrosoguanidine (MNNG) has been linked to their ability to tolerate the potentially cytotoxic methylated base O^6 -methylguanine in DNA (Karran and Bignami, 1992). This tolerance is imparted to cells defective for the DNA mismatch repair pathway (Branch *et al.*, 1993; Kat *et al.*, 1993; Hawn *et al.*, 1995). Since the O^6 -chloroethyl adducts produced by the chloroethylnitrosoureas go on to form guanine-cytosine interstrand DNA crosslinks (refer to section 1.5.1), tolerance of the original adduct is not a viable method of resistance. Therefore, this method of resistance will not be discussed further.

1.6 ADDITIONAL MECHANISMS OF DRUG RESISTANCE RELEVANT TO GLIAL TUMORS

1.6.1 Introduction

In addition to the classical chloronitrosoureas, a number of other chemotherapeutic agents have been used in the treatment of glial tumors. These agents often are used in various combination therapies and include procarbazine (Cairncross and Macdonald, 1988; Macdonald *et al.*, 1990; Cairncross and Macdonald, 1991; Cairncross *et al.*, 1992; Glass *et al.*, 1992; Kim *et al.*, 1996; Mason *et al.*, 1996), cisplatin (Boiardi *et al.*, 1991; Garfield *et al.*, 1991; Dropcho *et al.*, 1992; Mortimer *et al.*, 1992; Spence *et al.*, 1992; Yung *et al.*, 1992; Nakagawa *et al.*, 1994; Hiesiger *et al.*, 1995), melphalan (Brown *et al.*, 1990), thiotepa (Saarinen *et al.*, 1990), diaziquone (Cairncross and Macdonald, 1988; Cairncross and Macdonald, 1991), dacarbazine (Chang *et al.*, 1983; Nelson *et al.*, 1988), temozolomide (Newlands *et al.*, 1992; O'Reilly *et al.*, 1993; Newlands *et al.*, 1996), vincristine (Cairncross *et al.*, 1992; Glass *et al.*, 1992; Kim *et al.*, 1996), and etoposide (Boiardi *et al.*, 1991; Nakagawa *et al.*, 1994).

Although many of these are alkylating agents and their efficacy would be influenced by many of the same mechanisms of resistance already discussed (refer to sections 1.5.2 through 1.5.5), some belong to other families of drugs. For this reason, it is useful to consider alternate mechanisms that may play a role in the resistance of glial tumors to these agents.

1.6.2 Multidrug Resistance (P-glycoprotein)

Simultaneous resistance of cancers to multiple unrelated chemotherapeutic agents is a common occurrence and has been attributed to a phenomenon known as multidrug resistance (MDR). The best characterized MDR mechanism is P-glycoprotein, a protein encoded by the *MDR1* gene. P-glycoprotein is a member of the ATP-binding cassette (ABC) superfamily (Higgins, 1992) and acts as a transmembrane, energy-dependent drug-efflux pump (Endicott and Ling, 1989; Gottesman and Pastan, 1993). The drugs most often involved in MDR are vinca alkaloids (eg. vincristine), anthracyclines (eg. doxorubicin), and epipodophyllotoxins (eg. etoposide); elevated levels of P-glycoprotein have been associated with resistance to these agents in many systems (Endicott and Ling, 1989; Gottesman and Pastan, 1993). P-glycoprotein has been detected mainly on capillary endothelial cells within human brain tumors (Nabors *et al.*, 1991; Tanaka *et al.*, 1994) and implicated in the resistance, at least in part, of glial tumors to vincristine, etoposide, and doxorubicin (Feun *et al.*, 1994; Kiwit *et al.*, 1994; Abe *et al.*, 1995; Walther *et al.*, 1995).

1.6.3 Multidrug Resistance-Associated Protein

A second, and more recently characterized, mechanism of MDR is the multidrug resistance-associated protein (MRP) (Cole *et al.*, 1992). MRP is also a member of the ABC superfamily and exhibits approximately 15% amino acid identity with P-glycoprotein (refer to section 1.6.2) (Cole *et al.*, 1992; Grant *et al.*, 1994). MRP, like P-glycoprotein, is an energy-dependent drug pump that extrudes drugs from the cell and is capable of causing resistance to a similar, although not identical, spectrum of drugs

(Grant et al., 1994; Zaman et al., 1994; Koike et al., 1996; Utz et al., 1996). The main functional property of MRP that distinguishes it from P-glycoprotein is the ability of MRP to export lipophilic compounds that have been conjugated to glutathione, suggesting a role for MRP as a glutathione S-conjugate carrier (Leier et al., 1994; Muller et al., 1994; Versantvoort et al., 1995; Zaman et al., 1995; Hollo et al., 1996; Loe et al., 1996). Very few studies have been performed to investigate a possible role for MRP in human glial tumors but preliminary evidence suggests that low level MRP expression is more common in glioma than MDR gene expression (Feun et al., 1994). In addition, MRP has been implicated in the resistance, at least in part, of glial tumors to vincristine, etoposide, and doxorubicin (Abe et al., 1994; Abe et al., 1995).

1.6.4 Metallothionein

Although the physiologic role of metallothionein (MT) in resistance to the cytotoxic effects of anticancer drugs remains unclear and controversial, increased levels of MT have been associated with resistance to cisplatin and some alkylating agents (Kelley *et al.*, 1988; Kaina *et al.*, 1990; Lohrer *et al.*, 1990; Doz *et al.*, 1993). MTs are a family of low molecular weight, intracellular proteins characterized by an abundance of thiol groups (30% cysteine) and are, in fact, the principal source of thiol groups in eukaryotic cells (Kagi and Schaffer, 1988; Doz *et al.*, 1993). Originally, it was thought that MT and cytotoxic agents, including cisplatin and alkylating agents, interacted through simple covalent binding by the thiol groups (Zelazowski *et al.*, 1984); this mechanism would prevent the active molecules from reaching their DNA target in the cell. More recently, however, it has been suggested that the effect of MT on these agents may be more complex and the physiologic role of MT requires more study (Doz *et al.*, 1993). As with other tumor types, increased MT levels have been implicated in resistance of glial tumors to cisplatin (Phillips, 1991; Feun *et al.*, 1994).

1.6.5 Topoisomerase II

The DNA topoisomerases relieve the torsional stress on DNA during replication,

transcription, and cell division (Liu, 1989). Type II topoisomerases are necessary for the replication of DNA; they permit the creation of a double-strand break in the DNA, rotation of the two strands, and resealing of the helix (Liu, 1989; Tannock, 1992). Anthracyclines (eg. doxorubicin), epipodophyllotoxins (eg. etoposide), and aminoacridines (eg. amsacrine) all block topoisomerase II strand religation and stabilize DNA-protein complexes, resulting in cytotoxic lesions (Morrow and Cowan, 1990; Beck and Dalton, 1997). Resistance to these topoisomerase inhibiting drugs generally occurs in one of two ways: 1) topoisomerase levels are downregulated and the amount of stabilized complex formed is thus insufficient to initiate cell death; or 2) mutations of the topoisomerase allow it to resist stabilization by the drugs (Asano *et al.*, 1996b). Decreased topoisomerase II levels have been implicated in the resistance of glial tumors to etoposide (Mousseau *et al.*, 1993; Feun *et al.*, 1994; Asano *et al.*, 1996a; Asano *et al.*, 1996b).

In contrast, inhibition of topoisomerase II has been suggested to play a role in sensitizing brain tumor cells to DNA crosslinking anticancer agents (Ali Osman *et al.*, 1993). It was postulated that the topographical changes in DNA accompanying the unwinding of DNA by topoisomerase II would make the DNA segments more open and thus, would affect not only the extent and site of DNA damage but also the accessibility of the damaged DNA to repair proteins. The possibility of enhanced repair in the presence of topoisomerase II suggested a potential mechanism of increased resistance to the DNA crosslinking drugs.

1.7 THE p53 TUMOR SUPPRESSOR

1.7.1 Introduction

The p53 protein was first discovered in 1979 as part of a stable oligomeric complex with the SV40 large T-antigen that could be immunoprecipitated from extracts of SV40-transformed cells (Lane and Crawford, 1979; Linzer and Levine, 1979). Since p53 was expressed at higher levels in tumor cells than in normal cells (Benchimol *et al.*, 1982) and some cancer patients presented with circulating antibodies directed against the

protein (Crawford *et al.*, 1982), it was thought to be necessary for cellular transformation. Original p53 cDNA clones were found to immortalize cultured cells when co-transfected with other oncogenes such as *ras* and *E1A* and thus, p53 became regarded as an oncogene (Eliyahu *et al.*, 1984; Jenkins *et al.*, 1984; Parada *et al.*, 1984; Wolf *et al.*, 1984). This concept changed dramatically in 1989 when it was found that the clones used in previous experiments were actually mutants of p53 (Hinds *et al.*, 1989). Subsequently, it was determined that p53 inhibited the growth of tumor cells (Baker *et al.*, 1990; Chen *et al.*, 1990; Diller *et al.*, 1990; Mercer *et al.*, 1989; Finlay *et al.*, 1989). The *p53* gene was then reclassified properly as an anti-oncogene or tumor suppressor gene.

Since these initial findings, the role of p53 in human tumors has been studied extensively. The p53 gene is now considered the most frequently mutated gene in many common malignancies, including tumors of the colon, breast, lung, esophagus, liver, and brain (Nigro et al., 1989; Hollstein et al., 1991; Levine et al., 1991; Vogelstein and Kinzler, 1992). Approximately one half of these adult cancers contain a p53 mutation, of which more than 80% are missense mutations clustered within evolutionarily conserved regions (Hollstein *et al.*, 1991; Hollstein *et al.*, 1996). Although mutations and allelic losses are the most common mechanisms for the inactivation of p53, the loss of normal p53 function also can result from the formation of protein complexes with viral oncoproteins (Levine, 1990), dislocation of p53 between cell compartments (Moll et al., 1992), and inactivation and degradation through the overexpression of mdm-2 (Momand et al., 1992; Haupt et al., 1997; Kubbutat et al., 1997). In addition to the loss of p53 function, mutations also exist that result in a gain of oncogenic function (Dittmer et al., 1993; Hsiao et al., 1994). It is this potential to transform cells by two mechanisms that may provide an explanation for the high frequency of p53 mutations in human cancers (Greenblatt et al., 1994).

Mutation frequencies also have been studied extensively in human glial neoplasms. Chromosome 17p deletions and p53 gene mutations are found frequently in astrocytic gliomas of all malignancy stages (James et al., 1988; el Azouzi et al., 1989; James et al., 1989; Fults et al., 1989; Nigro et al., 1989; Chung et al., 1991; Hayashi et al., 1991; Mashiyama et al., 1991; Frankel et al., 1992; Fults et al., 1992; Saxena et al., 1992; Sidransky et al., 1992; von Deimling et al., 1992a; Aka et al., 1993; del Arco et al., 1993; Hunter et al., 1993; Louis et al., 1993; Newcomb et al., 1993; Wu et al., 1993; Bello et al., 1994; Koga et al., 1994; Kraus et al., 1994; Louis, 1994; Rasheed et al., 1994). In contrast, oligodendrogliomas rarely harbor p53 mutations (Reifenberger et al., 1994) but often accumulate wild type p53 protein (Pavelic et al., 1994). Elucidating the consequences of such differential genetic changes in gliomas will not only enhance understanding of brain tumor biology but also may have implications for diagnosis, prognosis, and treatment.

1.7.2 The *p53* gene

Interestingly, p53 is found in vertebrates only (Soussi *et al.*, 1987). The *p53* gene is expressed, at least at low levels, in all tissues (Rogel *et al.*, 1985) and appears to be highly conserved between species (Soussi *et al.*, 1987). The human *p53* gene is located on the short arm of chromosome 17 at position 17p13.1 and encompasses 16 to 20 kb of DNA (McBride *et al.*, 1986; Miller *et al.*, 1986). It is comprised of 11 exons, the first of which is located 8 to 10 kb away from exons 2 through 11 and is non-coding. Transcription of the human *p53* gene is controlled by two different promoters (Reisman *et al.*, 1988) and results in a transcript of approximately 2.8 kb in length (Matlashewski *et al.*, 1984).

The mouse p53 gene displays 81% homology to that of the human, has a similar genomic structure but spans approximately 12.5 kb of genomic DNA (Bienz *et al.*, 1984). It is localized to chromosome 11 and transcription results in a transcript approximately 2.0 kb in length. In addition to the gene found on chromosome 11, a processed, non-functional p53 pseudogene is located on chromosome 14 (Zakut-Houri *et al.*, 1983). Alternatively spliced mRNA also has been documented in mouse cells (Arai *et al.*, 1986; Han and Kulesz Martin, 1992; Kulesz Martin *et al.*, 1994; Wu *et al.*, 1994;

Will *et al.*, 1995). This variant form of p53 results in a protein shorter by nine amino acids and different by 17 amino acids at the C-terminus. Although a potential homologous alternative splice site is present in the human p53 gene, the presence of alternatively spliced human mRNA remains uncertain (Will *et al.*, 1995).

1.7.3 The p53 protein

Human p53 is a 393 amino acid nuclear phosphoprotein within which multiple functional domains can be found. The N-terminus contains an acidic transcriptional activation domain within amino acids 1-43 (Unger, 1992) and central residues 100-300 comprise a sequence specific DNA binding domain (Bargonetti *et al.*, 1993; Halazonetis *et al.*, 1993; Pavletich *et al.*, 1993; Wang *et al.*, 1993). The C-terminus encompasses an oligomerization domain within amino acids 320-360 (Sturzbecher *et al.*, 1992; Pavletich *et al.*, 1993; Wang *et al.*, 1993; Wang *et al.*, 1994) and a region from residues 330-393 capable of non-specific nucleic acid binding, reannealing, and strand-transfer (Wang *et al.*, 1993; Bakalkin *et al.*, 1994; Jayaraman and Prives, 1995; Lee *et al.*, 1995; Reed *et al.*, 1995). In addition, three nuclear localization signal sequences have been documented in the C-terminus of p53 (Shaulsky *et al.*, 1991) and regions that may harbor negative autoregulatory domains also have been suggested (Hupp *et al.*, 1992; Mosner *et al.*, 1995).

Levels of wild type p53 are generally low due to a short protein half life. The turnover of p53 *in vivo* is mediated by the ubiquitin proteolysis system (Maki *et al.*, 1996) and enzymes capable of participating in the *in vitro* ubiquitination of p53 have been documented (Scheffner *et al.*, 1990; Scheffner *et al.*, 1993; Scheffner *et al.*, 1995). Levels of p53 increase dramatically following DNA damage and this increase is thought to be due to p53 protein stabilization (Maltzman and Czyzyk, 1984; Kastan *et al.*, 1991; Kuerbitz *et al.*, 1992; Khanna and Lavin, 1993; Lu and Lane, 1993; Hess *et al.*, 1994; Sakhi *et al.*, 1994; Yamaizumi and Sugano, 1994; Midgley *et al.*, 1995; Sugano *et al.*, 1995). Maki and Howley (1997) suggested that one possible mechanism for the stabilization of p53 following DNA damage is the repression of the ubiquitin-mediated

degradation of p53. They showed that UV and gamma radiation had differential effects on ubiquitinated p53 and suggested that the UV-induced stabilization of p53 is due to a specific loss of p53 ubiquitination. Although there are several mechanisms by which p53 ubiquitination could be lost following UV treatment, one strong and very intriguing possibility is that changes in the phosphorylation of p53 following UV irradiation could modify p53 in such a way that it can no longer be recognized by the ubiquitinconjugating enzymes and the ubiquitin protein ligase involved in its ubiquitination (Maki and Howley, 1997).

Phosphorylation of the p53 protein appears to play a key role in the posttranslational modification and regulation of p53 (Meek, 1994). p53 is a substrate for a number of serine kinases including CDC2 (Bischoff *et al.*, 1990), CDK2 (Wang and Prives, 1995), casein kinase II (Meek *et al.*, 1990), DNA dependent protein kinase (Lees-Miller *et al.*, 1990), protein kinase C (Baudier *et al.*, 1992), mitogen-activated protein kinase (Milne *et al.*, 1994), and JNK1 (Milne *et al.*, 1995). The ability of p53 to be phosphorylated by these kinases suggests a mechanism by which p53 may be regulated through intracellular signalling. In fact, mutations of these phosphorylation sites can influence DNA binding, transactivation, tumor suppression, and cell cycle progression (Hupp *et al.*, 1992; Fiscella *et al.*, 1993; Mayr *et al.*, 1995; Wang and Prives, 1995).

1.7.4 The Role of p53 in Response to DNA-Damaging Agents

1.7.4.1 Introduction

When DNA damage occurs in a cell, the alterations must be detected and signals sent to cellular control mechanisms in order that the cell can take measures to prevent the damage from being genetically fixed and propagated (Kastan *et al.*, 1995; Leonard *et al.*, 1995). The two principal options for a cell in response to DNA damage are: 1) for the cell to temporarily arrest its cell cycle to allow for repair of the damage; or 2) initiate a programmed cell death pathway (apoptosis). The p53 protein has been identified as the most significant player in both of these pathways and upregulation of p53 protein

in response to DNA damage has been well documented (Maltzman and Czyzyk, 1984; Khanna and Lavin, 1993; Lu and Lane, 1993; Hess *et al.*, 1994; Sakhi *et al.*, 1994; Yamaizumi and Sugano, 1994; Midgley *et al.*, 1995). The factors influencing whether cells undergo p53-dependent cell cycle arrest or apoptosis in response to DNA damage are probably multifaceted and likely include cell type, nature or extent of DNA damage, and growth factor environment (Yonish Rouach *et al.*, 1991; Collins *et al.*, 1992; Clarke *et al.*, 1993; Lowe *et al.*, 1993a; Lowe *et al.*, 1993b; Di Leonardo *et al.*, 1994; Canman *et al.*, 1995; Girinsky *et al.*, 1995). Consequently, p53 has been referred to as the "guardian of the genome" (Lane, 1992).

In addition to the cell cycle arrest and apoptotic responses initiated by p53 in response to DNA damage, recent studies have suggested that p53 may have a role in the actual detection of DNA damage. Nelson *et al.* (1994) demonstrated that DNA strand breaks, but not other DNA lesions, were necessary for induction of p53 protein in response to DNA damaging agents. Others have shown that the C-terminal domain of p53 can bind single-stranded DNA ends and catalyzes DNA renaturation and strand transfer (Bakalkin *et al.*, 1994; Jayaraman and Prives, 1995; Reed *et al.*, 1995). Lee *et al.* (1995) also demonstrated that the C-terminal domain of p53 recognized single-stranded regions caused by insertion and deletion lesions. These p53-DNA complexes were highly stable and displayed a half life of greater than two hours, suggesting that this interaction was specific. It was hypothesized that p53 may act to recruit other proteins to the lesion site, providing a signal for DNA damage.

1.7.4.2 Cell Cycle Arrest

One of the ways p53 plays a role in protecting cells and organisms from the deleterious effects of DNA damage is by regulating cell cycle progression. The requirement of a wild type p53 for the induction of cell cycle arrest in response to various kinds of DNA damage has been observed in many cell types (Kastan *et al.*, 1991; Kuerbitz *et al.*, 1992; Livingstone *et al.*, 1992; Yin *et al.*, 1992; O'Connor *et al.*, 1993; Slichenmyer *et al.*, 1993; Cross *et al.*, 1995; Fan *et al.*, 1995; Girinsky *et al.*, 1995;

The mechanism by which p53 is thought to exert its effects on cell cycle regulation is by activating the transcription of genes that influence cell cycle progression, such as WAF1 and GADD45 (Kastan *et al.*, 1995; Leonard *et al.*, 1995). The WAF1 gene encodes the potent CDK inhibitor, p21 (el Deiry *et al.*, 1993; Harper *et al.*, 1993; Xiong *et al.*, 1993; Noda *et al.*, 1994). The p21 protein binds the ternary kinase complex composed of a CDK, a cyclin, and proliferating cell nuclear antigen (PCNA); p21 acts as an assembling factor of the CDK complex but when the ratio of p21 to complex is more than one, p21 inhibits kinase activity (Zhang *et al.*, 1994). Thus, upregulation of p21 results in the inhibition of kinase activity and subsequent cell cycle arrest (Xiong *et al.*, 1993; Michieli *et al.*, 1994).

The GADD45 gene was originally identified as an mRNA species whose expression was elevated in cells that had been growth arrested or had suffered DNA damage (Fornace, Jr. *et al.*, 1988; Fornace, Jr. *et al.*, 1989). GADD45, like p21, has been found to bind PCNA and inhibit DNA synthesis, although the mechanism by which this occurs is not well understood (Smith *et al.*, 1994; Chen *et al.*, 1995; Bates and Vousden, 1996). Interestingly, the binding of GADD45 to PCNA also has been linked with the stimulation of DNA excision repair (Smith *et al.*, 1994). In addition, the binding of p21 to PCNA prevented the association of PCNA with DNA polymerase δ but not DNA polymerase ϵ , thereby inhibiting DNA replication while allowing DNA repair to proceed (Li *et al.*, 1994; Shivji *et al.*, 1994; Waga *et al.*, 1994). Thus, by inducing the expression of p21 and GADD45, p53 can potentially achieve the dual effect of arresting cell cycle progression and stimulating DNA repair.

1.7.4.3 Apoptosis

A second, and equally important, role for p53 in protecting cells and organisms

from the deleterious effects of DNA damage is through the induction of programmed cell death or apoptosis. Apoptosis is an active cell death pathway that is characterized morphologically by cell shrinkage, membrane blebbing, chromatin condensation, apoptotic body formation, and DNA degradation; dead cells are removed by macrophages without the induction of an inflammatory response (Sen and D'Incalci, 1992; Schwartzman and Cidlowski, 1993; Bortner *et al.*, 1995). In the event that a cell suffers substantial and irreparable DNA damage, p53 can trigger the apoptotic pathway to remove damaged cells and prevent the propagation of cells that have sustained mutation. In addition, some cells appear to be primed for apoptosis and will undergo programmed cell death in the presence of even minor DNA damage (Wang and Harris, 1996). The ability of p53 to induce apoptosis in response to various forms of DNA damage has been documented in many different cell types (Clarke *et al.*, 1993; Lowe *et al.*, 1993e; Malcomson *et al.*, 1995; Norimura *et al.*, 1996).

The mechanism by which p53 induces apoptosis in response to DNA damaging agents is still unclear, although activities of p53 have been identified that may play a role in this process (Wang and Harris, 1996). The apoptotic pathway in humans that has received the most study to date includes multiple members of the bcl-2 family (Farrow and Brown, 1996) and is characterized by the interaction of the bcl-2 and bax proteins (Korsmeyer et al., 1993). The relative amounts of bcl-2 and bax can dictate whether a cell will survive or undergo apoptosis in response to an environmental stimulus. Bcl-2 can participate in homo- and heterodimeric complexes with bax; bcl-2/bax heterodimers promote cell survival (Yin et al., 1995) but the relative overexpression of bax results in the formation of bax/bax homodimers and subsequently, the induction of cell death (Oltvai et al., 1993). p53 has been shown to induce the transcription of bax (Miyashita and Reed, 1995) and inhibit the expression of bcl-2 (Miyashita et al., 1994). It remains unclear, however, whether transcriptional activation of p53 target genes is required for p53-dependent apoptosis and thus, the ability of p53 to repress transcription of genes required for cell survival or participate directly as a component of the enzymatic machinery for apoptosis may provide alternative mechanisms for apoptotic induction

1.7.4.4 p53-Regulated Drug Resistance Mechanisms

An alternative role for p53 in response to DNA-damaging agents may be to modulate response by influencing directly the regulation of drug resistance and DNA repair mechanisms. This has been suggested recently for a number of proteins. The first specific drug resistance mechanism implicated to be regulated by p53 was the *MDR1* gene (Chin *et al.*, 1992). Wild type p53 has been shown to repress the activity of the *MDR1* promoter *in vitro*, while mutant p53 was found to stimulate promoter activity (Chin *et al.*, 1992; Zastawny *et al.*, 1993; Nguyen *et al.*, 1994). These findings were supported by de Kant *et al.* when they found that *MDR1* expression correlated with mutant p53 expression in colorectal cancer metastases (de Kant *et al.*, 1996). In contrast, wild type p53 stimulated transcription from a *MDR* promoter in a p53-negative cell line during co-transfection studies, whereas a mutant p53 could not stimulate transcription (Goldsmith *et al.*, 1995). Finally, Strauss and Haas (1995) demonstrated that *MDR1* transcription could only be stimulated by specific subtypes of p53 mutants.

More recently, p53 has been implicated to play a role in the regulation of the DNA repair protein MGMT (refer to section 1.5.2). Again, as with the *MDR* promoter, conflicting data has emerged. Harris *et al.* (1996) suggested that wild type p53 suppressed transcription of the MGMT gene whereas Rafferty *et al.* (1996) found that MGMT was induced in a p53 gene-dose-dependent manner in response to ionizing radiation.

In addition to being able to stimulate DNA excision repair through p21 and GADD45 as described in section 1.7.4.2, p53 also has been shown to function in the NER pathway (refer to section 1.5.4) by direct interaction with proteins of the repair complex. p53 can selectively bind proteins that are part of the basal transcription complex, including ERCC2, ERCC3, and p62 (Xiao *et al.*, 1994; Wang *et al.*, 1994; Wang *et al.*, 1995; Leveillard *et al.*, 1996; Wang and Harris, 1996), whereas some mutant p53 proteins lack this ability (Leveillard *et al.*, 1996). The p53 protein also has

been implicated in regulation of the human mismatch repair gene hMSH2 due to the ability of p53 to bind the hMSH2 promoter *in vitro* (Scherer *et al.*, 1996).

1.7.4.5 Summary

When DNA damage occurs in a cell, the alterations must be detected and signals sent to cellular control mechanisms in order that the cell can take measures to prevent the damage from being genetically fixed and propagated. The p53 protein seems to be a key player in the response of cells to DNA damage. The two main response pathways induced by p53 are cell cycle arrest and apoptosis. The factors influencing whether cells undergo p53-dependent cell cycle arrest or apoptosis in response to DNA damage, and growth factor environment. In addition to cell cycle arrest and apoptosis, p53 also may play a role in the regulation of MDR expression and DNA repair; as much of this data has produced conflicting results, the significance of these findings remains to be determined. However, since p53 is the most frequently mutated gene in many common human malignancies, these studies have important pathogenetic and therapeutic implications.

CHAPTER 2

MATERIALS AND METHODS

2.1 MATERIALS

2.1.1 Cell Culture Materials and Reagents

Surgical tools for animal dissections were purchased from Fine Science Tools Inc. (North Vancouver, British Columbia). All coverslips and plasticware used in these studies were precoated with 13.3 μ g/ml poly-L-lysine obtained from Sigma Chemical Company (St. Louis, Missouri, USA). Circular coverslips (12 mm) were acquired from Fisher Scientific (Whitby, Ontario). Nunc cell culture plasticware. Dulbecco's Modified Eagles Medium (DMEM), L15 medium, Hank's Balanced Salt Solution, fetal bovine serum (FBS), antibiotics, and growth factors were purchased from Gibco BRL Life Technologies (Burlington, Ontario). All other cell culture reagents were obtained from Sigma Chemical Company. BCNU (1,3-bis(2-chloroethyl)-1-nitrosourea) was obtained from Bristol Laboratories of Canada (Montreal, Quebec) and MNU (N-methyl-Nnitrosourea) from Sigma Chemical Company. O⁶-Benzylguanine (BG) was kindly provided by Dr. Robert C. Moschel of the National Cancer Institute - Frederick Cancer Research and Development Center (Frederick, Maryland, USA). Anti-galactocerebroside (GC) hybridoma cells (Ranscht et al., 1982) were kindly provided by Dr. Mark Noble of the Huntsman Cancer Institute (Salt Lake City, Utah, USA). The anti-rat neural antigen-2 (anti-RAN-2) (Bartlett et al., 1981) and A2B5 (Eisenbarth et al., 1979) antibody hybridomas were purchased from the American Type Culture Collection (ATCC; Rockville, Maryland, USA). Anti-GC antibody was obtained from Boehringer Mannheim (Laval, Quebec) and anti-glial fibrillary acidic protein (GFAP) antibody from Dimension Laboratories Inc. (Mississauga, Ontario). Rhodamine- and fluoresceinconjugated secondary antibodies, biotinylated anti-rabbit IgG and 7-amino-4methylcoumarin-3-acetic acid (AMCA) streptavidin also were purchased from Dimension Laboratories Inc.

2.1.2 Molecular Biology Reagents

Restriction and modifying enzymes were obtained from Gibco BRL Life Technologies or Promega Corporation (Madison, Wisconsin, USA). Radiolabelled nucleotides ($[\alpha^{-32}P]dCTP$ and $[\gamma^{-32}P]dATP$) were purchased from Amersham Canada Limited (Oakville, Ontario). All other chemicals were of molecular biology grade and obtained from BDH Incorporated (Toronto, Ontario), Fisher Scientific, Gibco BRL Life Technologies and Sigma Chemical Company.

2.1.3 Primers, Probes and Plasmids

PCR primers and synthetic oligonucleotide probes for Northern hybridization were purchased from the Molecular Biology Core Facility of the Medical Research Council of Canada Group in Fetal and Neonatal Health and Development (London Regional Cancer Centre, London, Ontario). All oligonucleotides were prepared using a Gene Assembler Plus (Pharmacia LKB Biotechnology, Baie d'Urfé, Quebec). Sequences of the PCR primers are listed in Figure 2.1B. Sequences for the oligonucleotide probes are as follows:

MGMT 5' TTTCTTACCA GCAATTAGCA GCCCTGGCAG GCAACCCCAA AGC 3'

- GST-µ 5' CGGCACGAAT CCGCTCCTCC TCTGTCTCTC CACACAGGTG 3'
- GST- π 5' GGTAACCACC TCCTCCTTCC AGCTCTGGCC CTGGTCAGCC 3'
- Bcl-2 5' CATTGGGTTG CTCTCAGGCT GGAAGGAGAA GATGCCAGG 3'

Bcl-x 5' GGCTCCATTC ACGGCCGGGC TATCCGCCAG GTGCCAGG 3'

cDNA templates for the synthesis of DNA probes were isolated from the appropriate plasmid by restriction endonuclease digestion, gel electrophoresis, and purification from low melting agarose (Gibco BRL Life Technologies). All plasmids were transformed into the *Escherichia coli* strain XL 1-Blue (Stratagene Cloning Systems, La Jolla, California, USA) unless otherwise noted. All cDNA inserts used for Northern hybridization were generously donated by other investigators. A 1.35 kb mouse p53 partial cDNA in pUC-18 designated pMO53 was a gift from Dr. Sam Benchimol (Ontario Cancer Institute, Toronto, Ontario). A derivative of pMO53, pECM53, has been described by Johnson *et al.* (1991). The hamster *MDR* cDNA in pUC-9 designated

pEX/172 (Ng et al., 1989) and the primers (A100 and B100) required to label the probe using a PCR protocol (refer to section 2.7.4) were kindly provided by Dr. Victor Ling (British Columbia Cancer Research Centre, Vancouver, British Columbia). The 1.05 kb mouse *GAPDH* fragment in pBR322 designated pM-GAP (Edwards et al., 1985; Edwards and Denhardt, 1985) was donated by Dr. Ann Chambers (London Regional Cancer Centre, London, Ontario). A 1.3 kb mouse *mdm*-2 cDNA in pBluescript SK – designated 11B (Fakharzadeh et al., 1991) was provided by Dr. Donna George (University of Pennsylvania, Philadelphia, Pennsylvania, USA). The plasmid p21-9C (Huppi et al., 1994), containing a 0.85 kb mouse *WAF1* cDNA in pGEM-4Z, was supplied by Dr. Konrad Huppi (National Cancer Institute, Bethesda, Maryland, USA). Dr. A. J. Fornace (National Cancer Institute, Bethesda, Maryland, USA) provided pXR45m (Papathanasiou *et al.*, 1991), a 1.2 kb *GADD45* cDNA isolated from Chinese hamster. The 0.94 kb mouse *bax* cDNA designated 70Z7 (Oltvai *et al.*, 1993) was donated by Dr. Stanley Korsmeyer (Washington University School of Medicine, St. Louis, Missouri, USA).

2.2 RAT PRIMARY CELL CULTURES

2.2.1 Isolation of Glial Cell Enriched Cultures

Two populations of glial cells, oligodendrocyte lineage enriched and astrocyte enriched, were isolated from newborn (1 to 3 day old) rat cerebrums using a method modified from McCarthy and de Vellis (1980). Cerebral hemispheres were isolated from unanesthetized Sprague-Dawley neonates and dissected free of meninges and blood vessels. The tissue was minced with a scalpel blade and treated with 0.025% trypsin for 30 min at 37°C. Trypsin inhibitor with 40 μ g/ml DNase was added and incubated for 5 min to stop the digestion of the tissue. The tissue was centrifuged in a Beckman Model TJ-6 centrifuge (Beckman Instruments, Inc., Fullerton, California, USA) for approximately 1 min at 100g, the pellet resuspended in 40 μ g/ml DNase and triturated in a 10 ml pipette to dissociate cells. The resultant single cell suspension was centrifuged at 100g for 5 min, resuspended in DMEM supplemented with 10% FBS, 2 mM L-glutamine, 50 μ g/ml transferrin, 5 μ g/ml insulin, 30 nM selenium, 30 nM tri-

iodothyronine and 50 U/ml penicillin/streptomycin, and seeded into polylysine-coated 80 cm² flasks (2X10⁷ cells/flask). After five days, the FBS concentration was reduced to 1% to enhance oligodendrocyte differentiation (Raff *et al.*, 1983b). At day nine, oligodendrocyte lineage cells were separated from astrocytes by overnight rotary shaking (180 rpm) on a Gyrotory Shaker Model G2 (New Brunswick Scientific Company, Inc., Edison, New Jersey, USA) in DMEM containing 10% FBS, L-glutamine and 5 mM leucine methyl ester. Suspended cells (ie. oligodendrocyte lineage enriched) were decanted, filtered (100 μ m pore nylon mesh; B&SH Thompson & Company Ltd., Scarborough, Ontario), plated for 30 min to allow contaminating astrocytes to readhere and decanted again; adherent cells (ie. astrocyte enriched) were grown separately on coverslips for immunochemical analysis (refer to section 2.2.1.1).

2.2.1.1 Immunocytochemistry

A2B5 antibody (undiluted hybridoma supernatant), anti-GC antibody (diluted 1:40) and anti-GFAP antibody (1:300) were used to ascertain the percentage of oligodendrocyte lineage cells (A2B5-positive or GC-positive) and astrocytes (GFAP-positive, A2B5-negative) in the cultures. A2B5 antibody or anti-GC was added to unfixed cells and anti-GFAP after pre-treatment with 4% paraformaldehyde/0.05% Nonidet P40 detergent. To double label, these steps were done in sequence. Primary antibodies were detected as follows: A2B5 with rhodamine-conjugated goat anti-mouse IgM (diluted 1:250); anti-GC with fluorescein-conjugated goat anti-mouse IgG (1:400). Cells were visualized with fluorescence optics on a Leitz Diavert inverted microscope (Ernst Leitz Canada Ltd., Midland, Ontario).

2.2.2 Isolation of Purified Oligodendrocyte Progenitors and Oligodendrocytes

Populations of purified oligodendrocyte progenitors and oligodendrocytes were isolated from 7 day old rat corpus callosa using an immunopanning purification method modified from Barres *et al.* (1992). In the protocol described by Barres *et al.*, cells were

isolated from the optic nerves of postnatal Sprague-Dawley rats and three different antibody-coated plates were used sequentially for panning. The procedure yielded approximately 17 000 cells per animal and resulted in greater than 99% purity of oligodendrocyte progenitor cells. The experiments performed in this thesis required a much larger yield of purified progenitor cells and thus, cells were isolated from the corpus callosa of the animals and only the anti-RAN-2 panning plates were used in order to reduce the amount of cell death that occurred during the panning process. This modified protocol gave a greater than 100-fold increase in cell numbers while still maintaining a greater than 95% purity of oligodendrocyte progenitor cells.

Panning plates were prepared one day in advance by incubating 10 ml Tris buffer solution (50 mM, pH 9.5) with 50 μ g of goat anti-mouse IgG (Zymed Laboratories, Inc., San Francisco, California, USA) in a 10 cm Petri dish overnight at 4°C. The next day, each dish was washed 3 times with phosphate buffered saline (PBS) and incubated with anti-RAN-2 hybridoma supernatant (1:4) in DMEM containing 0.8 mg/ml bovine serum albumin (BSA) for 1 hr at 37°C. The addition of BSA was required to block the nonspecific adherence of cells to the panning plates. The dishes were then washed 3 times with PBS and PBS was left on the dishes until required for panning. Twelve anti-RAN-2 plates were prepared for each prep of 12-18 rats (one litter).

Corpus callosa were isolated from 6-9 unanesthetized Sprague-Dawley neonates (half of a litter), minced finely and incubated in 750 μ l of a 667 units/ml solution of collagenase in L15 medium for 45 min at 37°C. The tissue was then centrifuged at 1000 rpm in a Beckman Model TJ-6 centrifuge for 5 min and resuspended in 3 ml of a 30 unit/ml papain solution in DMEM containing L-cysteine and 0.013% DNase. This suspension was incubated for 1 hr at 37°C, centrifuged, and resuspended in 1 ml of papain inhibitor solution consisting of 2 mg/ml ovomucoid and 1 mg/ml BSA in L15 medium. The tissue was then triturated sequentially through 23 and 27 gauge needles to yield a single cell suspension. Two of these preps were performed in parallel for each litter of 12-18 rats.

The corpus callosa cell suspension was resuspended in Bottenstein-Sato (B-S) medium (Bottenstein and Sato, 1979) as modified by Lillien *et al.* (1988). This defined

medium consisted of 5.6 mg/ml glucose, 5 μ g/ml insulin, 100 μ g/ml transferrin, 100 μ g/ml BSA, 0.06 ng/ml progesterone, 16 μ g/ml putrescine, 40 ng/ml selenium, 40 ng/ml thyroxine, and 30 ng/ml tri-iodothyronine in DMEM. In order to deplete astrocytes and meningeal cells, the suspension was divided evenly into six samples and incubated on anti-RAN-2 panning plates for 30 min at 37°C. This incubation also depletes microglia and macrophages which stick via their Fc receptors to the original IgG coating on the panning plate. The plates were agitated gently at 15 min to ensure access of the panning surface to all cells. The non-adherent cells were transferred to a second set of six anti-RAN-2 plates for an additional 30 min incubation. An anti-GC panning plate to deplete oligodendrocytes was not used in these experiments since differentiated oligodendrocytes tended to die during the isolation process and thus did not present a major contamination problem. All remaining cells were seeded into polylysine-coated 80 cm² flasks in B-S medium containing 0.5% FBS and 10 ng/ml each of PDGF-AA and bFGF.

Oligodendrocyte progenitors were grown in the presence of 10 ng/ml each of PDGF-AA and bFGF in order to promote cell proliferation and prevent differentiation. To obtain pure populations of oligodendrocytes, oligodendrocyte progenitors were first grown to the required cell number. Oligodendrocyte progenitors were then allowed to undergo differentiation into oligodendrocytes by removing bFGF from the medium. All medium was removed from the flasks and the progenitors were washed once with citrate saline buffer (15 mM sodium citrate, 130 mM KCl). Fresh B-S medium containing 0.5% FBS and 10 ng/ml PDGF-AA was added to the flasks and the cells were allowed to differentiate for approximately 4 days. The state of differentiation was monitored morphologically; oligodendrocytes displayed highly branched processes in contrast to the bipolar morphology displayed by oligodendrocyte progenitors. Since the oligodendrocytes began to undergo cell death shortly after reaching terminal differentiation, all experiments were completed within 1 week of the original change in morphology. In all cases, the requisite growth factors were replenished every 24 hr. Both cell types were grown on coverslips for various time intervals under the above mentioned conditions before being used for immunochemical analysis (refer to section 2.2.2.1). With this modified protocol, greater than 95% oligodendrocyte progenitor purity and greater than 90%

oligodendrocyte purity could be obtained.

2.2.2.1 Immunocytochemistry

A2B5 antibody (mouse IgM), anti-GC antibody (mouse IgG₃) and anti-GFAP antibody (rabbit IgG) were used to ascertain the percentage of oligodendrocyte precursors (A2B5-positive, GC- and GFAP-negative), oligodendrocytes (GC-positive, A2B5- and GFAP-negative) and astrocytes (GFAP-positive, A2B5- and GC-negative) in the pure cultures using a triple staining protocol. All dilutions and washes were carried out using Hank's staining solution (Hank's balanced salt solution without sodium bicarbonate, 5% calf serum, 4.76 g/L N-2-hydroxyethylpiperazine-N'2-ethanesulphonic acid (HEPES; MW 238.3), 5.2 g/L HEPES (MW 260.3), 0.05% sodium azide). Briefly, coverslips containing cells were washed once by dipping the coverslip in Hank's staining solution and cells were fixed with 4% paraformaldehyde for 15-20 min. After all subsequent incubations, the coverslips were washed by dipping in four changes of Hank's staining solution. A mixture of 1/2 GC-1/2 A2B5 hybridoma supernatants was added for 20 min, and then ice-cold methanol for 15 min at -20°C in order to permeabilize the cell membranes. A mixture of anti-GFAP (1:300), anti-mouse IgG₃-fluorescein (1:100) and anti-mouse IgM-rhodamine (1:100) was incubated for 20 min in the dark. All subsequent incubations were performed in the dark. Goat anti-rabbit biotin (1:100) was added for 20 min followed by AMCA streptavidin (1:50) for 40 min. After the final wash. coverslips were rinsed in distilled water, mounted onto slides with a drop of anti-fade (2.5% 1,4-diazabicyclo-[2.2.2] octane in glycerol) and sealed with nail polish. Cells were visualized with fluorescence optics on a Leitz Diavert inverted microscope.

2.3 MOUSE PRIMARY CELL CULTURES

2.3.1 Mouse Husbandry

Mice were housed, fed, marked and sacrificed according to guidelines established by the Canadian Council on Animal Care. All experimental protocols have been approved by the Animal Care Committee of the University of Western Ontario and copies of all approvals are included in Appendix I. The colony was housed in standard cages under conditions of a 12 hr light/dark cycle and controlled temperature $(22 \pm 2^{\circ}C)$. Mice were fed Purina rat chow and tap water *ad libitum*. The colony was tested periodically for circulating antibodies to *Mycoplasma pulmonis*, coronaviruses and Sendai virus (Murine Immunocomb, Orgenics Limited, Yavne, Israel); at no time was a positive test result obtained for this colony.

The colony of p53 deficient mice was established from a breeding pair of TSGp53 knockout mice purchased from GenPharm International (Taconic, Germantown, New York, USA). Homozygous wild type (+/+) p53 mice were bred to obtain wild type offspring. Homozygous knockout (-/-) and heterozygous (+/-) mice were generated by breeding male -/- animals with +/- females. Mice were differentiated by genotyping tail DNA using a PCR-based method provided by GenPharm International (Timme and Thompson, 1994).

2.3.2 Mouse *p53* Genotyping

Mouse DNA was isolated by incubating a 0.5 cm piece of tail tip in 750 μ l tail DNA extraction buffer (50 mM Tris-Cl, pH 8.0, 100 mM EDTA, 100 mM NaCl, 1% SDS, 0.5 mg/ml Proteinase K) for 16 hr at 55°C. Following vigorous mixing, 250 μ l 6 M NaCl were added, mixing was repeated and the sample was centrifuged 10 min at full speed in a Sorvall MC 12V microcentrifuge (Dupont Canada, Mississauga, Ontario). Avoiding the surface layer, 750 μ l of the supernatant was removed, added to 500 μ l isopropanol and the sample was mixed and centrifuged. The pellet of crude DNA was washed in 70% ethanol and resuspended in 200 μ l T.1E (10 mM Tris-Cl, 0.1 mM EDTA, pH 7.5) for use in PCR.

To discriminate +/+, +/-, and -/- mice, target *p53* sequences were amplified from 2 μ l of DNA (Figure 2.1). Two PCRs were required for each tail DNA sample. The first reaction targeted specifically the wild type allele by amplifying a 320 bp intron 4/exon 5 fragment deleted in the knockout allele (WT5 primer pair) as well as a 280 bp exon 7 fragment to control for amplification success (WT7 primer pair). The second reaction targeted specifically the knockout allele by detecting a 150 bp fragment Figure 2.1 Mouse p53 Genotyping by PCR. A schematic of the PCR approach demonstrating the positions of the PCR primers on the wild type and mutant p53 alleles (A), sequences of the primers (B), and the expected pattern of bands obtained by agarose gel electrophoresis (C). Exons are represented by boxes (A); pol II, RNA polymerase II; neo, neomycin; WT, wild type; HET, heterozygote; KN, knockout; bp, base pairs.



Wild Type:





В

WT 5U:	5' - GTGTTTCATTAGTTCCCCACCTTGAC - 3'
WT 5L:	5' - ATGGGAGGCTGCCAGTCCTAACCC - 3'
WT 7U:	5' - CCCTACTCTACAACTAAAACTGAA - 3'
WT 7L:	5' - AAGCTGGGGAAGAAACAGGCTAAC - 3'
KN 5U:	5' - GTGGGAGGGACAAAAGTTCGAGGCC - 3'
KN 5L:	5' - TTTACGGAGCCCTGGCGCTCGATGT - 3'
KN 8U:	5' - CTAGTTTACACACAGTCAGGATGG - 3'
KN 8L:	5' - AAGAGGTGACTTTGGGGTGAAGCTC - 3'

С

Y	VT	HET	KN	
a	b	a b	a b	
		<u> </u>		320 bp
—			<u> </u>	280 bp
				150 bp

consisting of sequences from both the endogenous p53 and the construct used to disrupt p53 (KN5 primer pair) as well as a 280 bp control fragment (KN8 primer pair). Each 25 μ l reaction contained four oligonucleotide primers (1 μ M each), 50 mM KCl, 10 mM Tris-Cl (pH 9.0), 0.1% Triton X-100, 1 mM MgCl₂, four dNTPs (0.2 mM each) and 1 unit *Taq* DNA polymerase. An initial 7 min denaturation at 87°C was followed by 30 cycles of denaturation (92°C, 1 min), annealing (55°C, 2 min) and extension (72°C, 6 min) with a final 10 min extension at 72°C in a Perkin Elmer DNA Thermal Cycler (Perkin-Elmer Corporation, Norwalk, Connecticut, USA). PCR products were electrophoresed through a 4% agarose horizontal slab gel (Horizon 11.14, Gibco BRL Life Technologies) comprised of NuSieve GTG:Seakem GTG (3:1) (FMC Bioproducts, Rockland, Maine, USA) and visualized with ethidium bromide.

2.3.3 Isolation of Astrocytes and Fibroblasts

All primary +/+, +/- and -/- cell cultures were isolated from the TSG-p53 colony of mice. Enriched cultures of astrocytes were prepared from 4 to 6 day old mouse cerebrums using a method adapted from McCarthy and de Vellis (1980). Cerebral hemispheres were isolated from unanesthetized neonates, dissected free of meninges and blood vessels, and a single cell suspension was prepared by a combination of enzymatic (250 μ g/ml trypsin, 40 μ g/ml DNase) and mechanical dissociation. Fibroblast cultures were established from the cellular outgrowth of minced livers. All brains and livers were isolated separately and the resulting primary cultures were seeded into individual cell culture plates until p53 genotype could be resolved by PCR (section 2.3.2); once determined, cells of identical genotype were pooled from littermates (2-6 mice) for each independent experiment. All cultures were grown in DMEM containing 10% heat-inactivated FBS and 50 U/ml penicillin/streptomycin.

2.4 CELL PROLIFERATION ASSAY

The proliferation rates of all rat and mouse primary glial cells were determined by performing cell counts over a 1 week period. Briefly, cells were seeded onto polylysine coated 24-well plates at various densities. Each day for 7 days, cells from three individual wells were harvested with 0.05% trypsin and 0.53 mM EDTA and counted with a haemocytometer. The average cell count was plotted and the exponential growth phase determined. Approximate cell doubling times during exponential growth were calculated by averaging times obtained from at least three independent cell isolations.

2.5 CYTOTOXICITY ASSAY

BCNU was dissolved in absolute alcohol and diluted to 3.3 mg/ml with sterile water. MNU was prepared at a concentration of 6.6 mg/ml in sterile water. Both drugs were stored at -80°C and subsequent dilutions were made with culture medium at the time of cell treatment. BG stock solution was made in filter-sterilized dimethylsulfoxide (DMSO) at a concentration of 100 mM and stored at -80°C. At time of use, the BG was diluted to 1 mM in PBS and subsequent dilutions were made with culture medium. Twenty-five μ M DL-buthionine-[S,R]-sulfoximine (BSO) and 5 mM (-)-2-oxo-4-thiazolidine-carboxylic acid (OTZ) were prepared in medium at the time of treatment. Irradiation was performed at a distance of 31 cm from an Eldorado 6 ⁶⁰Co source (Theratronics, Kanata, Ontario) with an average dose rate of 240 cGy/min.

To test for cytotoxicity, 96 well plates were seeded with cells and incubated at 37° C. Due to differences in cell proliferation rates (refer to section 2.4), all rat cells were seeded at 10^{4} cells/well, wild type and heterozygous *p53* mouse cells at 7.5×10^{3} cells/well, and knockout *p53* mouse cells at 5×10^{3} cells/well. Forty-eight hr later, cells were treated and incubated for an additional 96 hr, at which time cytotoxicity was measured using the MTT (3-[4,5-dimethylthiazol-2-yl]-2,5-diphenyltetrazolium bromide) assay. BCNU and MNU treatments were performed for 2 hr. Pretreatment with modulators consisted of a 2 hr incubation for BG, a 24 hr incubation for BSO and 4 hr for OTZ. Irradiation was performed for the length of time required to achieve the desired dose (range of approximately 12.5 sec to 8.3 min).

The toxic effects of the above treatments were compared using the MTT assay as modified for chemosensitivity testing as described by Cole (1986). Ninety-six hr after treatment, 100 μ l of medium was removed, 25 μ l MTT solution (2 mg/ml in PBS) added and the plates were incubated for 2 hr at 37°C. To solubilize formazan crystals, 0.04 M hydrochloric acid in isopropanol was added to each well and thoroughly mixed. Plates were kept at 37°C for 1 hr and cell viability quantified by measuring light absorbance (570 nm) in a Bio-Rad Model 3550 automated microplate reader (Bio-Rad Laboratories, Richmond, California, USA). To control for differential growth rates, dose response curves were normalized by expressing absorbance values relative to non-treated control values. Cytotoxicity assay results were analyzed using the two-tailed *t*-test for independent means.

2.6 MGMT ASSAYS

Extracts were prepared from harvested rat cell pellets and rat tissues for the measurement of MGMT activity. Samples were washed in PBS, suspended in a buffer containing 70 mM HEPES, 1 mM dithiothreitol (DTT), 1 mM EDTA and 5% glycerol, and sonicated on ice for 30 sec (setting 30) using a 3mm tapered microtip attached to a Vibra-cell VC375 Cell Disrupter (Sonics & Materials Incorporated, Danbury, Connecticut, USA). Samples were frozen in liquid nitrogen and stored at -80°C until they were shipped on dry ice to Dr. Daniel Yarosh (Applied Genetics Incorporated, Freeport, New York, USA). MGMT activity was measured in the laboratory of Dr. Yarosh using a DNA adduct removal assay described by Myrnes *et al.* (1984). Extracts were incubated with calf thymus DNA containing O⁶-[³H]methylguanine for 30 min at 37°C. After the addition of 5% TCA and heating to 80°C for 30 min, tritiated methyl proteins were recovered by filtration through glass fibre filters (Whatman GF/C, Baxter Corporation, Mississauga, Ontario). Filters were washed with 5% TCA/ethanol and radioactivity counted in a xylene based cocktail containing 0.5% diphenyl oxazole and 0.02% 1-4-bis-2(5-phenyl oxazolyl) benzene. Activity was expressed as molecules of MGMT per cell.

Mouse primary cultures were harvested, washed once with citrate saline, and the cell pellet frozen in liquid nitrogen. Samples were stored at -80°C until they were shipped on dry ice to the laboratory of Dr. Anthony Pegg (The Milton S. Hershey Medical Center, Hershey, Pennsylvania, USA) for the measurement of MGMT activity. MGMT activity was determined by incubation of cell extracts with [³H]methylated calf

thymus DNA for 30 minutes at 37°C followed by acid hydrolysis of the substrate and separation of free bases on HPLC as described by Dolan *et al.* (1986; 1990a). The activity was expressed as fmol O⁶-methylguanine removed per mg protein. [³H]methylated DNA substrate was obtained by reaction of calf thymus DNA with 5.9 mCi/mmol N-[*methyl-*³H]N-nitrosourea (Amersham, Inc.). All MGMT results were analyzed using the two-tailed *t*-test for independent means.

2.7 NORTHERN BLOT ANALYSIS

2.7.1 RNA Isolation and Selection of mRNA

All glassware and solutions used in the isolation of RNA were autoclaved for 1 hr and solutions were treated with a 0.1% diethylpyrocarbonate (DEPC) solution where appropriate prior to use. Total RNA was isolated from cultured cells using TRIzol Reagent (Gibco BRL Life Technologies) according to manufacturer's instructions. Briefly, T80 flasks containing cells growing in monolayer were washed once in a citrate saline buffer. Four ml of TRIzol Reagent was added directly to the flask and cells were lysed by passing the suspension through a 10 ml pipette. After an incubation of at least 5 min at room temperature, the suspension was transferred to a glass Corex tube (Corning Costar Corporation, Richmond Hill, Ontario), 0.2 ml of chloroform was added per 1 ml of TRIzol Reagent, and the sample was vortexed for 15 sec. Following a 2 min incubation at room temperature, the sample was centrifuged at 12 000 rpm (Beckman Model J2-21 centrifuge) for 15 min at 4°C and the aqueous phase was transferred to a new Corex tube. RNA was precipitated from the aqueous phase by the addition of 0.5 ml of isopropanol per 1 ml of TRIzol Reagent used in the initial lysis. After a 10 min incubation at room temperature, the sample was centrifuged at 12 000 rpm for 10 min. The RNA pellet was then washed with 75% ethanol, dried and resuspended in DEPCtreated water.

Selection of rat mRNA was performed using Oligo(dT) Cellulose Columns (Gibco BRL Life Technologies). Columns were first equilibrated with 4 ml binding buffer (10 mM Tris-HCl (pH 7.5), 1 mM EDTA, 0.3 M NaCl, 0.1% SDS) in 1 ml aliquots. RNA pellets were resuspended in 3 ml binding buffer, heated at 70°C for 5 min and chilled
on ice for 5 min. Samples were loaded onto the equilibrated columns under gravity flow and washed with 4 ml of binding buffer to elute non-messenger RNA. mRNA was eluted with 1.5 ml elution buffer (10 mM tris-HCl (pH 7.5), 1 mM EDTA, 0.1% SDS) and the entire selection procedure was repeated a second time. The resultant mRNA was precipitated, resuspended in DEPC-treated water and quantitated by spectrophotometry (Beckman DU-65 Spectrophotometer, Beckman Instruments, Inc.). Samples were stored at -80°C for future use.

Selection of mouse mRNA was performed using the PolyATract mRNA Isolation System III (Promega Corporation). This system utilizes a biotinylated oligo(dT) primer and streptavidin coated paramagnetic particles to capture mRNA species at high efficiency. Briefly, 500 μ l of total RNA was heated to 65°C for 10 min and then annealed with 3 μ l biotinylated-oligo(dT) probe and 13 μ l 20X SSC (3 M NaCl, 0.3 M sodium citrate, pH 7.0). The sample was mixed gently and incubated at room temperature until completely cooled. The streptavidin-paramagnetic particles were resuspended and washed with 0.5X SSC according to manufacturer's instructions. The entire annealing reaction was added to the washed particles and incubated at room temperature for 10 min. The paramagnetic particles were captured using the supplied magnetic separation stand. The particles were washed with 0.1X SSC to remove contaminating non-messenger RNA and the mRNA was eluted with RNase-free water. Samples of mRNA were quantitated by spectrophotometry and stored at -80°C.

2.7.2 Gel Electrophoresis of mRNA

RNA was analyzed as described by Sambrook *et al.* (1989). The desired quantity of RNA was aliquotted, dried (Speed-Vac Model RH40-11, Savant Instruments, Hicksville, New York, USA), and rehydrated in 2.25 μ l of DEPC-treated water. The RNA was then resuspended in a formamide/formaldehyde mix comprised of 65% deionized formamide and 22% deionized formaldehyde in MOPS solution (0.25 M 3 [Nmorpholino] propane-sulphonic acid (pH 8.0), 0.62 mM EDTA, 6.2 mM sodium acetate). The samples were heated to 55°C for 15 min, removed to ice and glyoxal loading buffer (50% glycerol, 1 mM EDTA, 0.4% bromophenol blue, 0.4% xylene cyanol in MOPS solution) was added. Samples and molecular weight RNA markers (0.24-9.5 kb, Gibco BRL Life Technologies) were loaded immediately onto a 1.1% agarose horizontal slab gel containing formaldehyde and electrophoresed at 50 mA for 3 to 4 hr.

2.7.3 Northern Transfer

Formaldehyde gels used to resolve mRNA for transfer were rinsed briefly in distilled water. The lane containing the RNA ladder was removed and stained in $1 \mu g/ml$ ethidium bromide in order to visualize the RNA standards. Transfer of the mRNA was carried out by capillary action onto Gene Screen Plus membranes (Dupont Canada) in 10X SSC buffer using a method modified from Alwine *et al.* (1977). Transfer proceeded overnight, at which time the blot was removed, rinsed briefly in 2X SSC and baked for 2 hr at 80°C in a Gravity Convection Oven (Economy Model 16EG, Canlab, Baxter Corporation, Mississauga, Ontario). The membrane was stored in sealed plastic until hybridization was performed.

2.7.4 Radioactive Labelling of Hybridization Probes

Labelling the 5' ends of oligonucleotide probes was carried out in a reaction mixture containing 5 pmol probe, kinase buffer (0.1 M Tris-HCl (pH 9.5), 10 mM MgCl₂), 10 mM DTT, 40 μ Ci [γ -³²P]dATP and 5-10 units T4 polynucleotide kinase (Gibco BRL Life Technologies). The reaction was incubated at 37°C for 30 min and was stopped by the addition EDTA to 80 mM final concentration.

A PCR-based method was used to label the pEX/172 (*MDR*) cDNA probe (Ng *et al.*, 1989). Each 50 μ l reaction contained 100 pg pEX/172, primers A-100 (5' GTCCAGGGCTTCCTGGAC 3') and B-100 (5' CCATGGAGAAATAGATGC 3') at a concentration of 1 μ M, dATP, dGTP, dTTP (each at 0.25 mM), 0.5 μ M non-radioactive dCTP, 70 μ Ci [α -³²P]dCTP, 50 mM KCl, 10 mM Tris-Cl (pH 8.3), 1.5 mM MgCl₂, 0.01% gelatin and 2.5 units of *Taq* DNA polymerase. The mixture was subjected to 25 cycles of denaturation (94°C, 30 sec), annealing (48°C, 30 sec) and extension (72°C, 1 min).

Labelling of all other cDNA probes was performed using the Gibco Random

Primers DNA Labelling System (Gibco BRL Life Technologies) according to manufacturer's instructions. The reaction mixture containing 25 ng probe, dATP, dGTP, dTTP (0.02 mM each), random primers buffer (0.2 M HEPES, 50 mM Tris-HCl, 5 mM MgCl₂, 10 mM 2-mercaptoethanol, 0.4 mg/ml BSA, 5.4 OD₂₆₀ units/ml oligodeoxyribonucleotide primers, pH 6.8), 50 μ Ci [α -³²P]dCTP and 3 units Klenow Fragment was incubated at 25°C for 1 hr. The reaction was stopped by the addition of stop buffer (Na₂EDTA to 20 mM final concentration).

Following all labelling reactions, unincorporated isotope was removed by spin chromatography on a column of 3 ml of Sephadex G-50 (Pharmacia LKB Biotechnology). The G-50 gel filtration matrix retains the unincorporated isotope and allows the labelled probe to pass through. As the percentage of incorporation of radionucleotide remaining after chromatography has been found to be greater then 90%, the degree to which the probes were labelled was determined by measuring directly an aliquot in a scintillation counter (LKB Wallac 1211 Rackbeta, Pharmacia LKB Biotechnology).

2.7.5 Hybridization of Blots

Prior to using a blot for the first time, Gene Screen Plus membranes were dampened with RNase-free water, wrapped in plastic wrap, and exposed to ultraviolet light for 5 min in an ultraviolet viewing cabinet (Spectroline Model CC-80, Spectronics Corporation, Westbury, New York, USA) in order to cross-link the bound RNA to the membrane. The membranes were then prehybridized to block non-specific binding at 42°C in 5-10 ml prehybridization buffer (50% formamide, 10% dextran sulphate, 1% SDS, 1 M NaCl, 100 μ g/ml herring sperm DNA) in a rotating bottle hybridization oven (VWR Scientific Model 2710, Toronto, Ontario). After 1-4 hr, radiolabelled probe (10⁶ cpm/ml) was added to the prehybridization solution and incubated with constant rotation for 16-20 hr at 42°C. Following hybridization, blots were washed according to the protocol appropriate for the particular probe. For 5'-end labelled probes, the blots were washed briefly 4 times in 2X SSC/0.1% SDS at room temperature, once for 30 min in 2X SSC/0.1% SDS at 65°C, once for 5 min in 2X SSC/0.1% SDS at room temperature, and once briefly in 2X SSC at room temperature. For random primed probes, a more

stringent protocol was used. These blots were washed twice for 5 min in 2X SSC at room temperature, twice for 30 min in 2X SSC/1.0% SDS at 60°C, and twice for 30 min in 0.1X SSC/0.1% SDS at room temperature. Following the washes, membranes were blotted dry on Whatman paper (Baxter Corporation), wrapped in plastic wrap, and placed on a Storage Phosphor Screen (Molecular Dynamics, Inc., Sunnyvale, California, USA).

For further probing of the same membrane, each probe was removed from the blot by vigorous washing in a boiling solution of 0.01X SSC/0.01% SDS for 2-3 min. This was repeated twice with fresh solution and the blot was exposed to a Storage Phosphor Screen to ensure the sufficient removal of the probe. Membranes were then placed in prehybridization solution and hybridized with the next probe as described.

2.7.6 PhosphorImage and Densitometry Analysis

Northern hybridization images were scanned off of the Storage Phosphor Screens using a Personal Densitometer SI Model 375A (Molecular Dynamics, Inc.). Images were quantitated using ImageQuant software (Molecular Dynamics, Inc.). All images used for quantitation were below saturation and corrected appropriately for background. Quantified signal intensities for each probe were normalized to a poly-dT probe for rat samples and a mouse glyceraldehyde-3-phosphate dehydrogenase (GADPH) probe for mouse samples. Images were assembled into composites using CorelDraw software (Corel Corporation, Ottawa, Ontario) and signal intensity data was analyzed using Microsoft Excel software (Microsoft Canada, Inc., Mississauga, Ontario).

2.8 FLOW CYTOMETRY

Nuclear DNA was stained with propidium iodide using the CycleTEST PLUS DNA Reagent Kit (Becton Dickinson, San Jose, California, USA) following manufacturer's instructions. Briefly, cells were harvested at the appropriate times, washed once in citrate saline, and resuspended in Solution A (250 μ l per sample). Solution A contained trypsin in a detergent buffer and was responsible for the enzymatic disaggregation of the cells and digestion of the cell membranes and cytoskeleton. This suspension was mixed gently and incubated at room temperature for 10 min. Solution B (200 μ l) was added, mixed gently, and incubated for an additional 10 min. Solution B was comprised of trypsin inhibitor and ribonuclease A in a stabilizing buffer and resulted in the inhibition of the trypsin and digestion of cellular RNA. Lastly, ice-cold Solution C (200 μ l) was added, mixed gently and incubated on ice for 10 min in the dark. Solution C contained the propidium iodide in a stabilizing buffer and stoichiometrically bound the DNA at a final concentration of at least 125 μ g/ml. Samples were filtered through a 74 μ m mesh (B&SH Thompson & Company Ltd.) directly into a 12X75 mm tube and the tubes were stored on ice in the dark until flow analysis was performed (not greater than 30 min). Fifty thousand events per sample were analyzed with a Coulter EPICS XL flow cytometer (Coulter Electronics, Hialeah, Florida, USA). Cell distribution was determined after correction for background aggregates and debris using Multicycle software (Phoenix Flow Systems, San Diego, California, USA). Flow cytometry results were analyzed using the two-tailed *t*-test for independent means.

2.9 DETECTION OF APOPTOSIS

2.9.1 ApopTagTM In Situ Apoptosis Detection Kit

To test for the presence of apoptotic cells, +/+, +/-, and -/-p53 astrocytes were seeded onto coverslips in 24-well plates at a cell density of approximately 10⁴ cells/coverslip. At the desired time after BCNU treatment, cells were assayed with the ApopTagTM In Situ Apoptosis Detection Kit (Oncor, Gaithersburg, Maryland, USA) according to manufacturer's instructions. Briefly, cells were washed once in PBS, fixed for 10 min in 4% neutral buffered formalin, and quenched in 2% hydrogen peroxide in PBS for 5 min at room temperature. Cells were then exposed to the supplied equilibration buffer for 10-15 sec at room temperature. Without washing, coverslips were immediately incubated with reaction buffer containing terminal deoxynucleotidyl transferase (TdT) enzyme in a humidified chamber at 37°C for 1 hr. Negative controls consisted of identical cells assayed in the absence of the TdT enzyme (incubated with reaction buffer alone). The enzyme reaction was stopped by incubating the cells in the stop/wash buffer at 37°C for 30 min. Coverslips were washed 3 times in PBS and treated with antidigoxigenin-peroxidase at room temperature for 30 min. Cells were then washed again and exposed to 0.05% DAB substrate in PBS to which 0.02% fresh hydrogen peroxide had been added. Color development was monitored with a microscope and once the desired level of color was obtained, the coverslips were placed in distilled water and washed for at least 5 min. Cells were counterstained in 0.5% methyl green in 0.1M sodium acetate (pH 4.0) at room temperature for 10 min and washed in distilled water. Finally, coverslips were washed in 100% butanol, dehydrated in 3 changes of xylene and mounted on slides with Permount mounting medium (Fisher Scientific).

2.9.2 Cell Death Detection ELISA

In an attempt to achieve a more quantitative determination of the amount of apoptosis occurring in response to BCNU treatment than that provided by the ApopTagTM *In Situ* Apoptosis Detection Kit, the commercially available Cell Death Detection ELISA kit from Boehringer Mannheim was used. This photometric enzyme immunoassay allows for the qualitative and quantitative *in vitro* determination of cytoplasmic histone-associated DNA fragments (nucleosomes) after induced cell death.

Cells were seeded onto 35 mm plates at a cell density of 2.5×10^5 and, 24 hr later, treated as desired. At the time of sample harvest, the medium was removed from the plate and centrifuged in a Beckman Model TJ-6 centrifuge (1500 rpm for 5 min) to collect any cellular fraction. Five hundred μ l cold isolation buffer was added to the plate and cells were scraped into the buffer and collected in an Eppendorf tube. This solution was added to the cell pellet collected from the medium and placed at 4°C for 30 min with occasional inversion of the Eppendorf tube to mix gently. The sample was centrifuged at 4°C for 10 min in a Sorvall MC 12V microcentrifuge at full speed. Four hundred μ l of the supernatant was removed to a new tube and the sample was frozen at -20°C until the ELISA could be performed.

The ELISA was performed according to manufacturer's instructions. Wells were coated with 100 μ l of an anti-histone antibody solution for 1 hr at room temperature followed by 200 μ l incubation buffer for 30 min. The wells were rinsed 3 times with washing solution and then incubated with 100 μ l of sample solution (10X dilution of

original sample in incubation buffer) for 90 min at room temperature. Wells were washed again and 100 μ l of an anti-DNA-peroxidase conjugate solution was added for 90 min. After a final wash, 100 μ l of an ABTS (2,2'-azino-di-[3-ethylbenzthiazoline sulfonate(6)]) substrate solution was added with shaking (250 rpm) on a Gyrotory Shaker Model G2 until colour development was sufficient for analysis. The contents of the wells were homogenized by tapping carefully at the edges of the plate and measured at 405 nm in a Bio-Rad Model 3550 automated microplate reader. Incubation buffer was used as a blank and non-treated cells as negative controls. All samples were measured in triplicate. Results were expressed as enrichment factors which were calculated by expressing the average absorbance of the sample as a fraction of the average absorbance of the controls.

CHAPTER 3

THE ROLE OF O⁶-METHYLGUANINE-DNA METHYLTRANSFERASE IN THE RESPONSE OF GLIAL LINEAGE CELLS TO BCNU

3.1 INTRODUCTION

Oligodendrogliomas (Cairncross and Macdonald, 1988; Macdonald *et al.*, 1990; Cairncross *et al.*, 1994; Cairncross, 1994; Mason *et al.*, 1996) and mixed oligoastrocytomas (Glass *et al.*, 1992; Kim *et al.*, 1996) respond predictably to nitrosourea-based chemotherapy, especially PCV (procarbazine, CCNU (lomustine) and vincristine). In contrast, astrocytomas are relatively chemoresistant (refer to section 1.2.6.3 for details). Moreover, experimental studies have demonstrated that tumors of oligodendroglial lineage are preferentially induced in rats following transplacental exposure to ethylnitrosourea (Druckrey, 1973; Burger *et al.*, 1988) (refer to section 1.3 for details). The results of these clinical and experimental studies suggest not only that oligodendroglial tumors are chemosensitive but that the normal cell giving rise to oligodendrogliomas (ie. cell of origin) may be sensitive to the transforming effects of the nitrosoureas. The biochemical basis for this apparent vulnerability to the nitrosoureas is not known.

The nitrosoureas alkylate DNA at O⁶-guanine and repair at this site is mediated by MGMT (refer to section 1.5.2 for details). The cytotoxic and carcinogenic properties of the nitrosoureas appear related to MGMT activity; cells rendered MGMT positive by DNA transfection or transgenic expression become resistant to the nitrosoureas, resistant cells depleted of MGMT by pretreatment with O⁶-methylguanine or O⁶-benzylguanine are sensitized to the nitrosoureas and cells susceptible to neoplastic transformation by nitrosoureas express low levels of MGMT (refer to section 1.5.2 for details). In addition, procarbazine, the other major component of the PCV regimen, also produces cytotoxicity by alkyl substitution at O⁶-guanine (Schold, Jr. *et al.*, 1989). The purposes of this study were 1) to explore whether the differential response to BCNU displayed by gliomas was reproducible in normal rat glial lineage cells and 2) to ascertain whether this difference correlated with MGMT activity in these cells.

3.2 RESULTS

3.2.1 Immunochemical characterization of rat glial cell enriched cultures

Isolation of glial cell enriched cultures from neonatal rats was performed as described in section 2.2.1. The cellular composition of the cultures was determined by immunochemistry as shown in Figure 3.1 and described in section 2.2.1.1. Oligodendrocyte lineage cultures in these studies were approximately 80% pure. They contained 40% GC-positive, A2B5-negative, GFAP-negative oligodendrocytes, 10% A2B5-positive, GFAP-negative, GC-negative oligodendrocyte progenitors, and 30% A2B5-positive, GFAP-positive, GC-negative type 2 astrocytes. They also contained 5-10% GFAP-positive, A2B5-negative, GC-negative type 1 astrocytes and 10-15% unidentified cells. Cultures enriched for astrocytes (type 1; GFAP-positive, A2B5-negative, GC-negative) were at least 90% pure. Contaminants included small numbers of oligodendrocyte lineage and unidentified cells.

3.2.2 Effect of BCNU concentration on rat glial cell viability

To determine the effect of BCNU on glial cells, cultures of oligodendrocyte lineage cells and astrocytes were exposed to increasing concentrations of drug and cell viability was measured using the MTT assay (refer to section 2.5 for details). Oligodendrocyte lineage cells were found to be significantly more sensitive to BCNU than astrocytes (p=0.002 at 80 μ g/ml; Figure 3.2). The 50% inhibitory concentration (IC₅₀) was 40 μ g/ml for oligodendrocyte lineage cells and 102 μ g/ml for astrocytes, a 2.5-fold difference.

3.2.3 MGMT activity in rat glial cell cultures

MGMT activity was measured in oligodendrocyte lineage cells and astrocyte cultures in order to investigate possible differences in MGMT levels between the two

Figure 3.1 Immunochemical characterization of rat glial cell enriched cultures. This figure demonstrates the double labelling of a typical culture before the oligodendrocyte lineage cells have been separated from the astrocytes (ie. both cell types grown on the same coverslip). Oligodendrocyte lineage cells are stained with A2B5 and rhodamine-conjugated goat anti-mouse IgM (A). Astrocytes are stained with anti-GFAP and fluoresceinconjugated goat anti-rabbit IgG (B). Bar, 50 μ m.



Figure 3.2 Effect of BCNU concentration on rat glial cell viability. Oligodendrocyte lineage cells were found to be significantly more sensitive to BCNU than astrocytes. Circles represent average values for astrocyte enriched cultures (n=18) and squares depict oligodendrocyte lineage enriched cultures (n=6). The absorbance of untreated cells was normalized to a value of 1.0. Bars depict the standard error of the mean and the double asterisks show the level of statistical significance (p=0.002).



glial cell lineages (refer to section 2.6 for details). Oligodendrocyte lineage cells had low MGMT activity, approximately $5X10^3$ molecules per cell (Figure 3.3). MGMT levels in cells of the oligodendrocyte lineage were low relative to other rat cells (Grafstrom *et al.*, 1984), and approximately one-third of the 1.45X10⁴ value found for astrocytes (p < 0.02).

3.3 DISCUSSION

One intriguing development in neuro-oncology in recent years has been the observation that aggressive oligodendrogliomas (Cairncross and Macdonald, 1988; Macdonald *et al.*, 1990) and anaplastic mixed gliomas with an unequivocal oligodendroglial element (Glass *et al.*, 1992; Kim *et al.*, 1996) respond predictably to a combination chemotherapy regimen containing procarbazine, lomustine (CCNU) and vincristine, called PCV. Since oligodendrogliomas appear less vascular and less permeable than other gliomas (Paleologos *et al.*, 1992), superior drug delivery would seem an unlikely explanation for their chemosensitive nature. Presumably oligodendroglioma cells are inherently susceptible to the cytotoxic effects of PCV, but why? The answer to this question is undoubtedly important as it may lead to better therapies for oligodendrogliomas and to new treatment strategies for anaplastic astrocytomas and glioblastomas.

The observation that oligodendrogliomas are preferentially induced in rats following transplacental exposure to ethylnitrosourea suggested that developing oligodendrocytes have difficulty repairing nitrosourea-induced DNA alkylation and that oligodendrogliomas might be chemosensitive if this biochemical vulnerability were retained in tumors of the oligodendrocyte lineage. Since MGMT repair of DNA alkylation confers resistance to nitrosoureas it could be reasoned further that oligodendrocyte lineage cells and oligodendrogliomas might have low levels of MGMT activity. MGMT activity in human oligodendrocyte lineage cells has not been studied, but the results presented in section 3.2.3 demonstrated that low levels of MGMT activity are exhibited by rat oligodendrocyte lineage cells. In addition, several groups have measured low levels in human oligodendrogliomas (Wiestler *et al.*, 1984; Frosina *et al.*, Figure 3.3 MGMT activity in rat glial cell cultures expressed as molecules/cell. MGMT activity in cells of the oligodendrocyte lineage was approximately one-third of that found for astrocytes. The average values are shown for astrocyte enriched cultures (AEC; n=6) and oligodendrocyte lineage enriched cultures (OLEC; n=5). Error bars depict the standard error of the mean and the asterisk the level of statistical significance (p<0.02).



1990; Mineura et al., 1993; Silber et al., 1993; Citron et al., 1995; Mineura et al., 1996a) and we have found low MGMT activity in PCV-responsive human oligodendrogliomas (Nutt et al., 1995). Allowing for the fact that rat tissues and cells have a several-fold lower level of MGMT than human tissues and cells (Grafstrom et al., 1984), both rat oligodendrocyte lineage cells and human oligodendrogliomas displayed low levels of MGMT activity relative to other rat and human tissues, respectively. The significance of this finding in relation to the clinical observation that oligodendrogliomas in humans appear more sensitive to PCV than anaplastic astrocytomas and glioblastomas, although intriguing, requires further study. Wiestler et al. (1984) and Frosina et al. (1990) found lower levels of MGMT activity in oligodendrogliomas than astrocytomas but this has not been a universal finding (Mineura et al., 1993; Mineura et al., 1996a; Silber et al., 1993).

It is unlikely that the response of human cancers to chemotherapy is determined by a single biochemical process, such as MGMT. In addition to PCV and BCNU (Cairneross and Macdonald, 1988), oligodendrogliomas have been found to respond to a number of additional DNA damaging agents including diaziquone (Cairncross and Macdonald, 1988; Cairneross and Macdonald, 1991), melphalan (Brown et al., 1990), thiotepa (Saarinen et al., 1990), dacarbazine (New, 1995), and regimens containing cisplatin (Yung et al., 1992; Peterson et al., 1996). These findings raise the possibility that additional DNA repair mechanisms may be inefficient in oligodendroglial tumors. Studies of MGMT and other mechanisms of drug resistance in glial neoplasms, especially uncommon ones like oligodendrogliomas, are seriously limited by tissue availability. Moreover, contamination of samples by necrotic debris and non-tumor elements potentially complicate the interpretation of results. The observation that human oligodendrogliomas and rat oligodendrocyte lineage cells have similarly low MGMT activity may be noteworthy in this regard. If mechanisms of drug resistance in glial tumors reflect the biochemical properties of their cells of origin, then the results presented in this study suggest that normal glial cells may serve as a laboratory substitute for human gliomas.

CHAPTER 4

DIFFERENTIAL EXPRESSION OF DRUG RESISTANCE GENES IN CULTURES OF PURIFIED GLIAL CELLS

4.1 INTRODUCTION

Human oligodendrogliomas and mixed oligoastrocytomas respond predictably to nitrosourea-based chemotherapy, whereas astrocytomas are relatively chemoresistant (refer to sections 1.2.6.3 and 3.1 for details). One possible explanation for this difference in response is that oligodendroglioma cells are inherently susceptible to the cytotoxic effects of these alkylating agents. If this were true, one might hypothesize that mechanisms of drug resistance in glial tumors would reflect the biochemical properties of their cell of origin. In Chapter 3, it was shown that rat oligodendrocyte lineage cells were more sensitive to the cytotoxic effects of BCNU than astrocytes. In addition, levels of the DNA repair protein MGMT were reduced in oligodendrocyte lineage cells in comparison to astrocytes (Chapter 3; Nutt *et al.*, 1995). These results, although performed in rat and precursory in nature, provided support for this hypothesis.

It is unlikely that the response of human gliomas to chemotherapy is determined by a single biochemical process, such as MGMT. Additional mechanisms of resistance have been implicated for BCNU such as glutathione, GSTs, and nucleotide excision repair (refer to section 1.5). In addition to the classical chloronitrosoureas, a number of other chemotherapeutic agents have been used in the treatment of glial tumors. These agents often are used in various combination therapies and include procarbazine, cisplatin, melphalan, thiotepa, diaziquone, temozolomide, vincristine, and etoposide (refer to section 1.6.1). Although many of these are alkylating agents and their efficacy would be influenced by many of the same mechanisms of resistance as BCNU, some belong to other drug families. It is useful, therefore, to consider alternate mechanisms that may play a role in the resistance of glial tumors to chemotherapeutic agents. The purposes of this study were 1) to explore the possibility of the differential expression of drug resistance genes in cultures of purified rat glial cells; 2) to investigate the response of purified rat astrocytes, oligodendrocyte progenitors, and oligodendrocytes to treatment with the bifunctional alkylating agent, BCNU, and the monofunctional alkylating agent, MNU; 3) to examine the effect of MGMT modulation on resistance to BCNU and MNU in these cells; and 4) to study the effect of glutathione modulation on resistance to BCNU in purified astrocytes.

4.2 RESULTS

4.2.1 Immunochemical characterization of cultures of purified rat glial cells.

Isolation of cultures of purified rat glial cells from neonatal rats was performed as described in sections 2.2.1 (astrocytes) and 2.2.2 (oligodendrocyte progenitors and oligodendrocytes). The cellular composition of the cultures was determined by immunochemistry as described in section 2.2.2.1 and representative cultures are shown in Figure 4.1. Astrocyte cultures (GFAP-positive, A2B5-negative, GC-negative) were at least 90% pure. Oligodendrocyte progenitors (O2A cells; A2B5-positive, GFAP-negative, GC-negative) were cultured to a purity of greater than 95% and oligodendrocytes (GCpositive, A2B5-negative, GFAP-negative) to a purity greater than 90%.

4.2.2 Differential expression of drug resistance genes in purified rat glial cells.

In order to explore the possibility of the differential expression of drug resistance genes in cultures of purified rat glial cells, RNA was isolated from rat tissues (liver and brain) and purified glial cells (astrocytes, oligodendrocyte progenitors and oligodendrocytes), poly-A selected, and subjected to Northern analysis (refer to section 2.7). The Northern blots were probed sequentially for transcripts encoding MGMT, GST- μ , GST- π , p53, MDR, and MT (Figure 4.2). Hybridization signals were normalized to a poly-dT probe and comparative results are shown in Figures 4.3 and 4.4. A differential pattern of mRNA expression was observed for each of the six probes. Levels of *MGMT*, *GST*- π , and *p53* mRNA were highest in oligodendrocyte progenitors: progenitors displayed an approximate 2-fold increase in *MGMT* expression over astrocytes and a 5Figure 4.1 Immunochemical characterization of purified rat glial cell cultures. This figure demonstrates astrocytes, oligodendrocyte progenitors, and oligodendrocytes purified from rat using the procedures described in sections 2.2.1 (astrocytes) and 2.2.2 (progenitors and oligodendrocytes). Astrocytes are stained with anti-GFAP and fluorescein-conjugated goat anti-rabbit IgG (A). Oligodendrocyte progenitors (O2A cells) are stained with A2B5 and rhodamine-conjugated goat anti-mouse IgM (B). Oligodendrocytes are stained with anti-GC and fluorescein-conjugated goat anti-mouse IgG₃ (C). Bar, 25 μ m.



Figure 4.2 Differential expression of drug resistance genes in cultures of purified glial cells. RNA was isolated from rat tissues (liver and brain) and purified cells (astrocytes, progenitors, and oligodendrocytes), poly-A selected, and subjected to Northern analysis. Quantitated signal intensities for each probe were normalized to poly-dT as described in section 2.7.6 and comparative results are shown in Figures 4.3 and 4.4. Oligo, oligodendrocyte.

	Liver Brain Astrocyte Oligo Progenitor Oligodendrocyte
MGMT	
GST µ	
GST π	
p53	
MDR	
MT	

Figure 4.3 Densitometry analysis of *MGMT* and *GST* mRNA expression in purified rat glial cells. A differential pattern of mRNA expression was observed for each of the three probes. Quantified signal intensities from Figure 4.2 were normalized to poly-dT as described in section 2.7.6 and values for astrocytes were arbitrarily set to 1.0. L, liver; B, brain; A, astrocytes; OP, oligodendrocyte progenitors; O, oligodendrocytes.



Figure 4.4 Densitometry analysis of *p53*, *MDR* and *MT* mRNA expression in purified rat glial cells. A differential pattern of mRNA expression was observed for each of the three probes. Quantified signal intensities from Figure 4.2 were normalized to poly-dT as described in section 2.7.6 and values for astrocytes were arbitrarily set to 1.0. L, liver; B, brain; A, astrocytes; OP, oligodendrocyte progenitors; O, oligodendrocytes.



p53

MDR

fold increase over oligodendrocytes; GST- π expression was approximately half that of progenitors in both astrocytes and oligodendrocytes; and a slight increase in p53 expression was seen in progenitors over astrocytes while an approximate 2-fold increase was displayed over oligodendrocytes. Oligodendrocytes displayed the highest expression of GST- μ and MT mRNAs: GST- μ levels were only slightly lower in astrocytes, whereas oligodendrocyte progenitors expressed approximately half of that observed in oligodendrocytes; and MT mRNA levels were slightly lower in both astrocytes and oligodendrocyte progenitors. Only MDR expression was highest in astrocytes, with oligodendrocyte progenitors displaying a slightly lower level of expression and oligodendrocytes exhibiting approximately one-fifth of the astrocyte levels. In general, expression of the drug resistance genes tended to be lower in tissue samples than in samples from the cultured cells, with the notable exception of MT; MT expression was at least 10-fold higher in liver tissue than the levels displayed by the cultured glial cells.

Two mRNA transcripts were detected for each of p53 and MDR. Although two transcripts have not been demonstrated previously in rat tissue, alternatively spliced forms of p53 have been documented in mouse tissues (Han and Kulesz Martin, 1992; Will *et al.*, 1995). Two transcripts have been detected previously for MDR in rat and were shown to correspond to the MDR1a and MDR1b genes (larger and smaller transcripts respectively; Gant *et al.*, 1992).

4.2.3 Effect of BCNU and MNU treatment on the viability of purified rat glial cells.

Since a differential expression of drug resistance genes was found in cultures of purified rat glial cells, the response of purified rat astrocytes, oligodendrocyte progenitors, and oligodendrocytes to treatment with BCNU and MNU was investigated. Cells were isolated from rats, treated with increasing concentrations of either BCNU or MNU, and cell viability tested using the MTT assay (as described in sections 2.2.1, 2.2.2, and 2.5). Oligodendrocyte progenitors exhibited a trend toward increased resistance to BCNU when compared to astrocytes but this difference did not reach statistical significance (Figure 4.5). Both astrocytes and oligodendrocyte progenitors

Figure 4.5 Effect of BCNU treatment on purified rat astrocytes, oligodendrocyte and oligodendrocytes. Oligodendrocyte progenitors progenitors, exhibited a trend toward increased resistance to BCNU when compared to astrocytes but this difference did not reach statistical significance. Both astrocytes and oligodendrocyte progenitors displayed greater resistance to BCNU than did oligodendrocytes. The absorbance of untreated cells was normalized to a value of 1.0. Symbols represent the mean of at least three independent experiments, each performed in triplicate. Circles represent astrocytes; triangles, oligodendrocyte progenitors; squares, oligodendrocytes. Bars depict the standard error of the mean and asterisks the level of statistical significance (**, P < 0.005).



displayed greater resistance to BCNU than did oligodendrocytes. All three cell types were extremely resistant to MNU although astrocytes and oligodendrocyte progenitors displayed a significantly greater resistance than oligodendrocytes at a concentration of 320 μ g/ml (Figure 4.6).

In order to culture pure populations of glial cells successfully, each of the three glial cell types had to be grown in a defined, but distinct, cell culture medium (refer to sections 2.2.1 and 2.2.2). Due to these differences, the effect of medium on the viability of rat astrocytes in response to BCNU was determined. Astrocytes displayed a slight increase in resistance to BCNU when grown in the alternate B-S medium + 0.5% FBS supplemented with either PDGF and FGF or PDGF alone in comparison to the usual medium of DMEM + 10% FBS (Figure 4.7).

4.2.4 Effect of BG pretreatment on the response of purified rat glial cells to BCNU and MNU.

Since the expression of *MGMT* mRNA levels correlated with resistance to BCNU and MNU in the purified glial cells, the effect of BG pretreatment on the response of these cells to BCNU and MNU was determined. Cells were pretreated with 25 μ M BG for two hours, exposed to increasing concentrations of either BCNU or MNU, and cell viability was determined using the MTT assay (refer to section 2.5 for details). BG appeared to increase the sensitivity to BCNU in all three cell types (Figure 4.8). In contrast, BG did not increase sensitivity to MNU in any of the cells (Figure 4.9).

4.2.5 Effect of BSO and OTZ pretreatment on the response of astrocytes to BCNU.

Although the levels of GST mRNAs did not appear to correlate directly with resistance to BCNU, modulation of glutathione has been demonstrated to alter resistance to a number of alkylating agents, including BCNU (refer to section 1.5.3). For this reason, the effect of BSO and OTZ pretreatment on the response of astrocytes was examined. Astrocytes were pretreated with either 25 μ M BSO for 24 hours or 5 mM OTZ for four hours, exposed to increasing concentrations of BCNU, and cell viability was determined using the MTT assay (refer to section 2.5 for details). BSO pretreatment

Figure 4.6 Effect of MNU treatment on purified rat astrocytes, oligodendrocyte progenitors, and oligodendrocytes. All three cells types were extremely resistant to MNU although astrocytes and oligodendrocyte progenitors displayed a significantly greater resistance than oligodendrocytes at 320 μ g/ml. The absorbance of untreated cells was normalized to a value of 1.0. Symbols represent the mean of at least three independent experiments, each performed in triplicate. Circles represent astrocytes; triangles, oligodendrocyte progenitors; squares, oligodendrocytes. Bars depict the standard error of the mean and asterisks the level of statistical significance (**, P < 0.005).



Figure 4.7 The effect of cell culture medium on the viability of rat astrocytes in response to BCNU. Astrocytes displayed a slight increase in resistance to BCNU when grown in the alternate cell culture media. The absorbance of untreated cells was normalized to a value of 1.0. Symbols represent the mean of at least two independent experiments, each performed in triplicate. Circles represent astrocytes grown in DMEM + 10% FBS; squares, B-S medium + 0.5% FBS, PDGF, and bFGF; triangles, B-S medium + 0.5% FBS and PDGF. Bars depict the standard error of the mean and asterisks the level of statistical significance (*, P < 0.05).



Figure 4.8 Effect of BG pretreatment on the response of purified rat glial cells to BCNU. Astrocytes (A), oligodendrocyte progenitors (B), and oligodendrocytes (C) were pretreated with 25 μ M BG for two hours and then exposed to varying concentrations of BCNU. BG appeared to increase slightly the sensitivity to BCNU in all three cell types. Symbols represent the mean of at least two independent experiments, each performed in triplicate. Circles represent cells treated with BCNU alone; squares, 25 μ M BG pretreatment. Bars depict the standard error of the mean and asterisks the level of statistical significance (*, P < 0.05; **, P < 0.005).


Figure 4.9 Effect of BG pretreatment on the response of purified rat glial cells to MNU. Astrocytes (A), oligodendrocyte progenitors (B), and oligodendrocytes (C) were pretreated with 25 μ M BG for two hours and then exposed to varying concentrations of MNU. BG did not significantly increase sensitivity to MNU in any of the three cells types. Symbols represent the mean of at least two independent experiments, each performed in triplicate. Circles represent cells treated with MNU alone; squares, 25 μ M BG pretreatment. Bars depict the standard error of the mean.



Α

В

С

MNU Concentration (µg/ml)

increased the sensitivity of astrocytes to BCNU but OTZ pretreatment displayed no effect (Figure 4.10).

4.3 DISCUSSION

One of the disadvantages with the cell culture methodology utilized in Chapter 3 was that the oligodendrocyte lineage cell population was not pure. Since multiple cell types were present in the cultures, characteristics displayed by the oligodendrocyte lineage cells could not be attributed to a particular glial cell type. Due to the refinement of glial cell culture techniques by Barres *et al.* (1992), it became possible to culture large numbers of pure populations of astrocytes, oligodendrocyte progenitors, and oligodendrocytes from rat brain. All experiments presented in this chapter were performed on these pure populations of glial cells.

As discussed previously, it is unlikely that the response of human gliomas to chemotherapy is determined by a single biochemical process. Furthermore, if mechanisms of drug resistance in some subtypes of glial tumors reflect the biochemical properties of their cell of origin, then it would follow that a differential expression of drug resistance genes would be seen in subtypes of normal glial cells. This hypothesis was supported by experiments performed on pure cultures of rat glial cells. RNA was isolated from purified rat astrocytes, oligodendrocyte progenitors and oligodendrocytes, poly-A selected, and subjected to Northern analysis. Northern blots were probed sequentially for transcripts encoding MGMT, GST- μ , GST- π , p53, MDR, and MT. A differential pattern of mRNA expression was observed for each of the six probes. Rat oligodendrocytes expressed lower levels of *MGMT*, *p53*, and *MDR* than astrocytes. These findings demonstrated that more than one mechanism of drug resistance was deficient in oligodendrocytes and the possibility that some of these same mechanisms may be inefficient in oligodendroglial tumors is raised.

Since a differential expression of drug resistance genes was found in cultures of purified rat glial cells, the response of purified rat astrocytes, oligodendrocyte progenitors, and oligodendrocytes to treatment with BCNU and MNU was investigated. Both astrocytes and oligodendrocyte progenitors displayed greater resistance to BCNU Figure 4.10 Effect of BSO and OTZ pretreatment on the response of rat astrocytes to BCNU. Astrocytes were pretreated with either 25 μ M BSO for 24 hours (A) or 5 mM OTZ for four hours (B) and then exposed to varying concentrations of BCNU. BSO pretreatment increased significantly the sensitivity of astrocytes to BCNU but OTZ pretreatment displayed no effect. Symbols represent the mean of at least two independent experiments, each performed in triplicate. Circles represent astrocytes treated with BCNU alone; squares, BSO or OTZ pretreatment. Bars depict the standard error of the mean and asterisks the level of statistical significance (*, P < 0.05; **, P < 0.005).





than did oligodendrocytes. All three cell types were extremely resistant to MNU although astrocytes and oligodendrocyte progenitors displayed a significantly greater resistance than oligodendrocytes at a concentration of 320 μ g/ml. Thus, it was demonstrated that subtypes of pure populations of rat glial cells displayed a differential response to both the bifunctional alkylating agent BCNU and the monofunctional alkylating agent MNU. As each of the three glial cell types were grown in a defined, but distinct, cell culture medium, the possibility that these differences in response to cytotoxic drug was due to dissimilar medium could not be excluded. Since astrocytes were the only cell type that could survive in each of the three media, the effect of cell culture medium on the resistance of astrocytes to BCNU was determined. Astrocytes displayed a slight increase in resistance to BCNU when grown in the alternate B-S medium + 0.5% FBS supplemented with either PDGF and bFGF (oligodendrocyte progenitor medium) or PDGF alone (oligodendrocyte medium) in comparison to the usual medium of DMEM + 10% FBS. There was still no significant difference, however, between the response of astrocytes and that of oligodendrocyte progenitors to BCNU and astrocytes remained significantly more resistant than oligodendrocytes in each of the three media. Therefore, it was concluded that the differential response of astrocytes, oligodendrocyte progenitors, and oligodendrocytes to cytotoxic drugs was not a result of differences in cell culture conditions and was, in fact, due to properties of the individual cell types.

The two mechanisms of drug resistance that have received the most study in relation to BCNU are MGMT and the GSTs (refer to sections 1.5.2 and 1.5.3 respectively). Since MGMT protein activity has been found to correlate with resistance to BCNU in cultures of mixed glial cells (refer to Chapter 3; Nutt *et al.*, 1995) and expression of *MGMT* mRNA levels correlated with resistance to BCNU and MNU in purified glial cells, the effect of pretreatment with the modulator, BG, on the response of these cells to BCNU and MNU was determined. BG has been shown to cause the depletion of MGMT activity (refer to section 1.5.2) and appeared to increase the sensitivity to BCNU, in all three cell types. In contrast, BG did not increase sensitivity to MNU in any of the cells. Thus, it was concluded that MGMT contributed, at least in part, to the resistance of rat glial cells to BCNU. Although the results obtained for MNU

seemed contradictory to those for BCNU, the extreme level of resistance displayed by glial cells to MNU may have precluded any modulation at the drug concentrations used; it is possible that even in the absence of MGMT, glial cells remain resistant to the cytotoxic effects of MNU at these concentrations.

Although the levels of *GST* mRNAs did not appear to correlate directly with resistance to BCNU, modulation of glutathione has been demonstrated to alter resistance to a number of alkylating agents, including BCNU (refer to section 1.5.3). For this reason, the effect of BSO and OTZ pretreatment on the response of astrocytes was examined. BSO has been shown to cause depletion of cellular glutathione levels and OTZ has been shown to increase these levels (refer to section 1.5.3). Interestingly, BSO pretreatment increased the sensitivity of astrocytes to BCNU but OTZ pretreatment displayed no effect. Thus, it appeared that the glutathione-GST detoxification system also played a role in the resistance of astrocytes to BCNU. The finding that depletion of glutathione increased sensitivity to BCNU but the induction of glutathione did not enhance resistance was of particular interest; since GSTs are a family of enzymes that catalyze the conjugation of cytotoxic drugs to glutathione, thereby enhancing the rate of detoxification, these results indicated that the levels of GSTs, not glutathione, may actually be the rate limiting step for detoxification in glial cells.

In summary, it was determined that rat glial cell subtypes exhibit differential expression of a number of drug resistance genes. In addition, a differential response to the alkylating agent BCNU was demonstrated. The increased resistance of astrocytes in comparison to oligodendrocytes was modulated, at least in part, by both BG and BSO, suggesting a role for both MGMT and the GSTs in the resistance of astrocytes to BCNU. Thus, the response of normal rat glial cells to BCNU appeared to be dependent on more than one mechanism of drug resistance. If mechanisms of drug resistance in some subtypes of glial tumors reflect the biochemical properties of their cell of origin, then these experiments provide additional support for the hypothesis that oligodendroglioma cells may be inherently susceptible to the cytotoxic effects of alkylating agents.

CHAPTER 5

THE EFFECT OF p53 STATUS ON THE VIABILITY AND CELL CYCLE RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU

5.1 INTRODUCTION

The p53 protein plays a role in protecting cells and organisms from the deleterious effects of DNA damage by regulating cell cycle progression and by triggering programmed cell death (refer to sections 1.7.4.2 and 1.7.4.3 for details). A p53dependent G₁ arrest in some cell types in response to certain kinds of DNA damage is protective (Kastan et al., 1991; Kuerbitz et al., 1992); it prevents the propagation of mutations by allowing additional time in the cell cycle for DNA repair. In other cell types in response to DNA damage, p53-dependent apoptosis is protective to the organism (Clarke et al., 1993; Lowe et al., 1993a); it prevents the propagation of abnormal cells by facilitating their self-destruction. These normal functions for p53 may influence the response of human cancers to DNA-damaging therapies in opposite ways; p53 could confer resistance if G₁ arrest permits therapeutic DNA damage to be repaired, but also sensitivity if damaged cancer cells apoptose. The situation is further complicated because p53 often is abnormal in cancer cells; dysfunctional p53 could confer sensitivity if G₁ arrest cannot occur and treatment-induced DNA damage cannot be repaired, but also resistance if damaged cancer cells are unable to apoptose. Failure to apoptose seems to predominate in cancers with p53 abnormalities; abrogation of a functional p53 often is associated with treatment resistance (Lowe et al., 1993a; O'Connor et al., 1993; McIlwrath et al., 1994; Nabeya et al., 1995) and shorter survival (Chang et al., 1995). This may not be true for all tumor types, however. Wild type p53 may confer resistance to treatment in some cancer cells (Biard et al., 1994; Petty et al., 1994; Fan et al., 1995; Vikhanskaya et al., 1995) and in evolving astrocytic gliomas, wild type p53 has been associated with shorter patient survival (van Meyel et al., 1994). The effect of p53 status

on the response of astrocytomas to chemotherapy has not been studied.

The p53 protein has been shown to activate transcription of the WAF1 and GADD45 genes in response to some forms of DNA damage (refer to section 1.7.4.2), resulting in the induction of a cell cycle arrest. p53 also has been demonstrated to activate transcription of the mdm-2 gene (Barak et al., 1993). Mdm-2 is a cellular inhibitor of p53 in that it binds the transactivation domain of p53 and downregulates its ability to activate transcription (Momand et al., 1992; Meltzer, 1994; Kussie et al., 1996). In addition, binding by mdm-2 can promote the rapid degradation of p53 (Haupt et al., 1997; Kubbutat et al., 1997). Thus, the potential exists for a tightly regulated cell cycle response to DNA damage; the activation of p53 can upregulate genes known to affect a cell cycle response as well as mdm-2, resulting in its own inactivation and degradation through an autoregulatory loop.

The purposes of this study were 1) to examine the effect of p53 status on the viability of mouse astrocytes exposed to BCNU, the standard cytotoxic drug for astrocytic gliomas; 2) to determine if this effect was drug-specific or cell-specific; 3) to explore the effect of BCNU treatment on the expression of three p53-regulated genes (mdm-2, WAF1, and GADD45) that may play a role in the cell cycle response to DNA damage; and 4) to examine the effect of p53 status on the cell cycle of astrocytes in response to DNA damaging treatments (ie. BCNU and radiation).

5.2 RESULTS

5.2.1 Effect of p53 on the viability of mouse astrocytes exposed to BCNU.

First, the effect of p53 on the viability of BCNU-treated astrocytes was determined. Astrocytes were isolated from mice with distinct p53 genotypes (+/+, +/-, -/-), treated with increasing concentrations of BCNU and cell viability tested using the MTT assay (as described in sections 2.3.3 and 2.5). Homozygous wild type (+/+) p53 astrocytes displayed significantly greater resistance to BCNU than knockout (-/-) astrocytes (Figure 5.1). Heterozygous (+/-) astrocytes demonstrated an intermediate level of resistance and were significantly more resistant to BCNU than -/-

Figure 5.1 Effect of p53 status on the viability of mouse astrocytes treated with BCNU. Homozygous wild type (+/+) p53 astrocytes displayed significantly greater resistance to BCNU than homozygous knockout (-/-) astrocytes. Heterozygous (+/-) astrocytes displayed an intermediate level of resistance. The absorbance of untreated cells was normalized to a value of 1.0. Symbols represent the mean of at least five independent experiments, each performed in triplicate. Circles represent +/+ cells; triangles, +/- cells; squares, -/- cells. Bars depict the standard error of the mean and asterisks the level of statistical significance (*, P < 0.05; **, P < 0.005).



astrocytes (Figure 5.1). Hence, wild type p53 appeared to protect mouse astrocytes from the cytotoxic effects of BCNU in a gene-dose-dependent fashion.

5.2.2 Effect of p53 on the viability of mouse astrocytes and fibroblasts in response to cytotoxic treatment.

To test whether the p53 genotype/BCNU response pattern was BCNU-specific or astrocyte-specific, the effect of p53 on the viability of astrocytes treated with MNU and radiation, and that of fibroblasts treated with BCNU and radiation was determined. The p53 genotype had no discernible effect on the response of astrocytes to MNU over the range of concentrations tested (Figure 5.2A). However, in all other situations -/- cells were more resistant than +/+ cells: -/- astrocytes exhibited greater resistance to radiation than +/+ astrocytes (Figure 5.2B); -/- fibroblasts exhibited greater resistance to BCNU than +/+ fibroblasts (Figure 5.3A) and -/- fibroblasts were more resistant than +/+ fibroblasts to radiation at lower doses, although not at higher doses (Figure 5.3B). Heterozygous astrocytes and fibroblasts displayed an intermediate level of resistance to all toxic agents (data not shown). In summary, mouse astrocytes appeared to have a unique response to BCNU; wild type p53 rendered astrocytes resistant to this drug while in no other situation tested were +/+ cells more resistant than -/- cells.

5.2.3 Effect of BCNU treatment on p53-regulated gene expression.

In order to investigate whether BCNU treatment could influence the expression of p53-regulated genes, +/+ and -/- astrocytes and fibroblasts were exposed to 40 μ g/ml BCNU for 2 hr and samples were taken at defined times for Northern analysis (refer to section 2.7). The Northern blots were probed sequentially for transcripts encoding mdm-2, p21 (WAF1) and GADD45 (Figures 5.4 - 5.7). Hybridization signals were normalized to the *MGAP* signal from the same lane. Expression of *GADD45* remained relatively constant over the 24 hr time period in response to BCNU in all four cell types; densitometry was performed (refer to section 2.7.6) and any differences in signal intensities were not greater than 50% and were within the range of variation displayed by the control samples.. There was evidence, however, of a slight induction Figure 5.2 Effect of p53 status on the viability of mouse astrocytes in response to cytotoxic treatment. There was no significant difference in the level of resistance exhibited by wild type (+/+) p53 nor knockout (-/-) astrocytes to MNU over the range of concentrations used (A). Knockout astrocytes displayed greater resistance to radiation than +/+ astrocytes (B). The absorbance of untreated cells was normalized to a value of 1.0. Symbols represent the mean of at least five independent experiments, each performed in triplicate. Circles represent +/+ cells; squares, -/- cells. Bars depict the standard error of the mean and asterisks the level of statistical significance (*, P < 0.05; **, P < 0.005). Gy, grays.



ω

101

Figure 5.3 Effect of p53 status on the viability of mouse fibroblasts in response to cytotoxic treatment. Knockout fibroblasts (-/-) demonstrated greater resistance to BCNU than wild type (+/+) p53 fibroblasts (A). In response to radiation, -/- fibroblasts displayed greater resistance than +/+fibroblasts at lower doses, but no significant difference at higher doses (B). The absorbance of untreated cells was normalized to a value of 1.0. Symbols represent the mean of at least five independent experiments, each performed in triplicate. Circles represent +/+ cells; squares, -/cells. Bars depict the standard error of the mean and asterisks the level of statistical significance (*, P < 0.05; **, P < 0.005). Gy, grays.



Figure 5.4 Effect of BCNU treatment on the mRNA expression of p53 and p53regulated genes in wild type p53 mouse astrocytes. Cells were treated with 40 μ g/ml BCNU for two hours. The first treated mRNA sample (t=0) was harvested at the end of the drug incubation and successive samples were taken at the specified times (in hours). Quantified signal intensities for each probe were normalized to *MGAP* and comparative results for *mdm-2* and *WAF1* are shown in Figures 5.8 and 5.9 respectively. Cont, control; MGAP, mouse glyceraldehyde-3-phosphate dehydrogenase.



Figure 5.5 Effect of BCNU treatment on the mRNA expression of p53-regulated genes in knockout p53 mouse astrocytes. Cells were treated with 40 μ g/ml BCNU for two hours. The first treated sample (t=0) was harvested at the end of the drug incubation and successive samples were taken at the specified times (in hours). Quantified signal intensities for each probe were normalized to MGAP and comparative results for mdm-2 and WAF1 are shown in Figures 5.8 and 5.9 respectively. Cont, control; MGAP, mouse glyceraldehyde-3-phosphate dehydrogenase.



Figure 5.6 Effect of BCNU treatment on the mRNA expression of p53 and p53regulated genes in wild type p53 mouse fibroblasts. Cells were treated with 40 μ g/ml BCNU for two hours. The first treated sample (t=0) was harvested at the end of the drug incubation and successive samples were taken at the specified times (in hours). Quantified signal intensities for each probe were normalized to *MGAP* and comparative results for *mdm-2* and WAF1 are shown in Figures 5.8 and 5.9 respectively. Cont, control; MGAP, mouse glyceraldehyde-3-phosphate dehydrogenase.



•

Figure 5.7 Effect of BCNU treatment on the mRNA expression of p53-regulated genes in knockout p53 mouse fibroblasts. Cells were treated with 40 μ g/ml BCNU for two hours. The first treated sample (t=0) was harvested at the end of the drug incubation and successive samples were taken at the specified times (in hours). Quantified signal intensities for each probe were normalized to MGAP and comparative results for mdm-2 and WAF1 are shown in Figures 5.8 and 5.9 respectively. Cont, control; MGAP, mouse glyceraldehyde-3-phosphate dehydrogenase.



in *mdm*-2 and a 2- to 5-fold induction in *WAF1* expression in response to BCNU (summarized in Figures 5.8 and 5.9 respectively).

The effect of BCNU treatment on the expression of p53 mRNA also was determined. Wild type astrocytes and fibroblasts were exposed to 40 μ g/ml BCNU for 2 hr and samples were taken at defined times for Northern analysis (refer to section 2.7). There was no evidence of p53 mRNA induction; expression remained constant over a 24 hr time period in response to BCNU in both cell types (Figures 5.4 and 5.6).

5.2.4 Cell cycle distribution of mouse astrocytes treated with BCNU or radiation.

Since changes in the expression of the cell cycle *WAF1* gene were found in mouse astrocytes in response to BCNU (refer to section 5.2.3), the effect of p53 on cell cycle distribution in response to cytotoxic treatment was investigated. Astrocytes were isolated from mice with distinct p53 genotypes (+/+, +/-, -/-), treated with either 20 μ g/ml BCNU or 2 Gy radiation, and the effect on cell cycle distribution was determined by flow cytometry (refer to sections 2.3.3 and 2.8 for details). A p53 gene-dose-dependent cell cycle pattern was observed in non-treated, control astrocytes; heterozygous (+/-) astrocytes consistently displayed a response intermediate to wild type (+/+) and knockout (-/-) astrocytes. This pattern remained relatively constant following treatment for both BCNU-treated and irradiated astrocytes (Figure 5.10). This apparent p53 gene-dose-dependent effect likely reflects differences in growth rates of the cells; +/+, +/- and -/- astrocytes have doubling times of 60-65, 50 and 40 hours respectively (refer to section 2.4). The cell cycle response to treatment with BCNU was not, however, the same as that observed in response to radiation (Figures 5.11 and 5.12).

Whereas the induction of p21 often is consistent with a cell cycle arrest in G_1 (Di Leonardo *et al.*, 1994; Chen *et al.*, 1995), a G_1 arrest was not observed in +/+ astrocytes in response to BCNU (Figure 5.11A); instead, the percentage of cells in G_1 decreased and a G_2 arrest was evident 48 hours post BCNU treatment. Knockout astrocytes exhibited a similar G_2 arrest pattern (Figure 5.11B). These findings suggest that the cell cycle modification that occurred in astrocytes following treatment with

Figure 5.8 Densitometry analysis of changes in *mdm-2* mRNA expression in response to BCNU treatment. A slight induction in *mdm-2* mRNA expression was seen in response to BCNU in all four cell types. Quantified signal intensities from Figures 5.4 to 5.7 were normalized to *MGAP* as described in section 2.7.6 and the lowest control value for each individual cell type (panels A through D) was arbitrarily set to a value of 1.0. Comparative results are given for wild type (A) and knockout (B) astrocytes and wild type (C) and knockout (D) fibroblasts. Lanes are identified as: 1, control T=0; 2, BCNU T=0; 3, BCNU T=4; 4, BCNU T=8; 5, BCNU T=12; 6, BCNU T=24; 7, control T=24; T represents time in hours.



С

.

114

Figure 5.9 Densitometry analysis of changes in WAF1 mRNA expression in response to BCNU treatment. A slight induction in WAF1 mRNA expression was seen in response to BCNU in all four cell types. Quantified signal intensities from Figures 5.5 to 5.8 were normalized to MGAP as described in section 2.7.6 and the lowest control value for each individual cell type (panels A through D) was arbitrarily set to a value of 1.0. Comparative results are given for wild type (A) and knockout (B) astrocytes and wild type (C) and knockout (D) fibroblasts. Lanes are identified as: 1, control T=0; 2, BCNU T=0; 3, BCNU T=4; 4, BCNU T=8; 5, BCNU T=12; 6, BCNU T=24; 7, control T=24; T represents time in hours.



С

116

Figure 5.10 Summary of cell cycle distribution of mouse astrocytes treated with 20 μ g/ml BCNU for 2 hrs (A) or 2 Gy radiation (B). A *p53* gene-dosedependent cell cycle pattern was observed in non-treated, control astrocytes; heterozygous (+/-) astrocytes consistently displayed a response intermediate to wild type (+/+) and knockout (-/-) astrocytes. This pattern remained relatively constant following treatment for both BCNU-treated and irradiated astrocytes. Bars depict the mean of at least four independent experiments (\pm SEM). Detailed statistical examination of the cell cycle responses of +/+ and -/- astrocytes are depicted in Figures 6.2 and 6.3.





A



Figure 5.11 Cell cycle distribution of mouse astrocytes treated with 20 μ g/ml BCNU for 2 hrs. Homozygous wild type p53 astrocytes showed a decrease in G₁ and an increase in G₂ at 48 hours (A); a similar trend was observed for homozygous knockout astrocytes (B). Bars depict the mean of at least four independent experiments (\pm SEM) and asterisks significant change from non-treated astrocytes (*, P < 0.05; **, P < 0.005).



Figure 5.12 Cell cycle distribution of mouse astrocytes treated with 2 Gy radiation. Irradiated wild type p53 astrocytes accumulated in G₁; maximum G₁ accumulation occurred at 24 hours (A). The percentage of cells in S phase was significantly lower than non-treated astrocytes at all three time points, with the lowest value at 12 hours. There were no statistically significant changes in the cell cycle distribution of knockout astrocytes in response to radiation (B). Bars depict the mean of at least four independent experiments (\pm SEM) and asterisks significant change from non-treated astrocytes (*, P < 0.05; **, P < 0.005).


BCNU was a p53-independent response. The G_1 decrease appeared at 48 hours in +/+ astrocytes compared to 24 hours in -/- cells and a ripple effect was observed in the percentage of -/- astrocytes in S phase. The earlier G_1 decrease and the detection of S phase changes in -/- cells might be explained by the faster rate of growth of -/- as compared to +/+ astrocytes.

To test whether this p53 genotype/cell cycle pattern seen in astrocytes was specific to BCNU, I studied the effect of p53 on cell cycle changes in irradiated astrocytes. In contrast to BCNU, +/+ astrocytes exhibited a G₁ arrest in response to radiation which peaked 24 hours post-treatment; the percentage of cells in G₁ was returning to control levels 48 hours post-radiation (Figure 5.12A). At 12 hours, G₁ arrest was accompanied by a decrease in the percentage of astrocytes in S phase and a transient G₂ arrest. There were no cell cycle changes seen in -/- astrocytes were dependent on a functional p53. In summary, BCNU and radiation had dissimilar effects on the cell cycle of astrocytes; while the response to BCNU was comparable in both +/+ and -/- cells and therefore p53-independent, the response to radiation was clearly p53-dependent.

5.3 DISCUSSION

The p53 protein plays a key role in protecting cells and organisms from the deleterious effects of DNA damage. The functional significance of this role is especially important in the resistance of human cancers to DNA-damaging treatments since the p53 gene is now considered the most frequently mutated gene in many common malignancies, including glial tumors. The effect of p53 status on the response of astrocytomas to chemotherapy has not been studied. Malignant glial tumors and cell lines often harbor multiple genetic defects (refer to section 1.2.5), thus impeding the examination of the effects of an isolated gene or protein. In order to examine the specific effects of p53 on drug resistance, I chose to utilize cells isolated from TSG-p53 mice (refer to section 2.3.1); this experimental system allowed me to examine the consequences of p53 gene dosage on drug resistance on an otherwise normal cellular background.

First, the response of normal and p53 deficient mouse astrocytes to the alkylating

agent 1,3-bis(2-chloroethyl)-1-nitrosourea (BCNU), a clinically useful DNA-damaging drug to which some human astrocytomas are resistant and some are relatively sensitive, was determined. Astrocyte cultures were isolated from the cerebrums of +/+, +/-, and -/-p53 neonatal mice and treated with increasing concentrations of BCNU. Wild type p53 astrocytes were significantly more resistant to BCNU than -/-p53 astrocytes, with +/- astrocytes exhibiting an intermediate level of resistance. Hence, wild type p53 appeared to protect mouse astrocytes from the cytotoxic effects of BCNU in a gene-dose-dependent fashion. In addition, these results suggested that +/+ astrocytes were not undergoing p53-dependent apoptosis; one would expect +/+ astrocytes to be more sensitive to BCNU than -/- cells if they were preferentially undergoing programmed cell death.

The effect of p53 on the response of cells to toxic stimuli may be cell-specific or insult-specific (Lowe et al., 1993a; O'Connor et al., 1993; Biard et al., 1994; McIlwrath et al., 1994; Petty et al., 1994; Fan et al., 1995; Nabeya et al., 1995; Vikhanskaya et al., 1995) and sometimes there is no significant relationship between p53 status and treatment resistance (Brachman et al., 1993; Slichenmyer et al., 1993; Petty et al., 1994; Fan et al., 1995; Vikhanskaya et al., 1995; Vikhanskaya et al., 1993; Slichenmyer et al., 1993; Petty et al., 1994; Fan et al., 1995; Vikhanskaya et al., 1995). Therefore, to test whether the p53 genotype/BCNU response pattern was BCNU-specific or astrocyte-specific, the effect of p53 on the viability of astrocytes treated with MNU and radiation, and that of fibroblasts treated with BCNU and radiation was determined. Mouse astrocytes appeared to have a unique response to BCNU; wild type p53 rendered astrocytes resistant to this drug while in no other situation tested were +/+ cells more resistant than -/- cells.

Interestingly, astrocytes and fibroblasts displayed greater than anticipated resistance to radiation and small differential responses between +/+ and -/- cells. These results may reflect the inability of the MTT assay to distinguish between healthy, dividing cells and viable, non-dividing cells; the utilization of primary cell cultures necessitated use of the non-clonogenic MTT assay.

In light of its effects on the chemosensitivity of mouse astrocytes, might the status of the p53 gene in human astrocytomas influence tumor response to BCNU, the standard

drug therapy for this tumor type? Mutations of p53 are an early molecular event in the genesis of many astrocytomas (Chung *et al.*, 1991; Frankel *et al.*, 1992; Louis *et al.*, 1993; Van Meir *et al.*, 1994) but are less commonly observed in the tumors of older patients than in the tumors of younger patients (Chung *et al.*, 1991; Louis *et al.*, 1993). Older patients with high grade astrocytomas, often with normal p53, are reported to be less likely to benefit from treatment with BCNU than younger patients with histologically similar tumors (Chang *et al.*, 1983; Nelson *et al.*, 1988; Grant *et al.*, 1995), often harboring p53 mutations. In addition, Russell *et al.* (1995) studied glioma cell lines and xenografts and found that those containing normal p53 tended to be more resistant to procarbazine, another clinically useful alkylating agent, than glioma cell lines and xenografts with mutated p53. Their results in astrocytic glioma cell lines, coupled with my findings in primary cultures of mouse astrocytes, suggest that p53 status may influence the response of human cancers to chemotherapy in a cell- and drug-specific manner.

In order to investigate whether BCNU treatment could influence the expression of p53-regulated genes that may play a role in the cell cycle response to DNA damage, the mRNA expression of *mdm-2*, *WAF1*, and *GADD45* following treatment with BCNU was examined. There was evidence of a slight induction in *mdm-2* and a 2- to 5-fold induction in *WAF1* expression in response to BCNU but expression of *GADD45* remained constant over the 24 hr time period. Increased *mdm-2* expression could be detected in the first treated samples that were harvested immediately following the two hour BCNU incubation. Since mdm-2 acts in an autoregulatory loop with p53, these results provide evidence for a very tightly regulated p53 response. The induction of *WAF1* also was rapid in fibroblasts; increased levels could be found in the first samples following BCNU incubation. In comparison, astrocytes demonstrated a slightly delayed induction of *WAF1*; increased expression was observed by four hours following treatment and had returned to normal levels by 24 hours. There was no change in the expression of *p53* over the course of treatment as expected; any evidence of p53 induction in response to DNA damage has been shown to occur at the level of protein stabilization (refer to sections 1.7.3 and 1.7.4.1).

The relative delayed induction of WAF1 in astrocytes, combined with the almost immediate induction of *mdm-2* suggested that a role for a G_t arrest in response to BCNU in astrocytes may not be crucial. For this reason, the effect of p53 on the cell cycle of mouse astrocytes in response to DNA damaging treatments (ie. BCNU and radiation) was examined. Although astrocytes demonstrated the characteristic G₁ arrest in response to radiation, a G₂ arrest was observed in response to BCNU. G₂ arrest has been noted previously in various cell types in response to the nitrosoureas (Tobey, 1975; Tobey and Crissman, 1975), including BCNU (Barranco and Humphrey, 1971; Tobey, 1975; Tobey and Crissman, 1975; Nomura et al., 1978; Hoshino et al., 1981), and in response to some other alkylating agents (Tobey, 1975). Nomura et al (1978) demonstrated that BCNU-treated 9L rat glioma cells arrested in G_2 by 48 hours; G_2 arrest persisted for 24 hours and then returned to control levels. Transient delays in the G₁ and S phases lasting less than four hours in total also have been observed (Barranco and Humphrey, 1971; Tobey and Crissman, 1975). The design of my experiments would not have detected minor delays of this type. Lastly, leftward "skewing" of the G₂/M peak has been noted previously (Tobey and Crissman, 1975) and also was seen in my flow cytometry studies (data not shown); this finding has been interpreted by Tobey and Crissman (1975) as a late S/G_2 accumulation, not exclusively a G_2 arrest. In summary, the cell cycle changes found in BCNU-treated astrocytes appeared to be characteristic of the nitrosoureas (Tobey, 1975).

BCNU and radiation had dissimilar effects on the cell cycle of astrocytes; while the response to BCNU was comparable in both +/+ and -/-p53 cells, the response to radiation was p53-dependent. Mouse astrocytes were clearly able to undergo cell cycle arrest in response to BCNU but this response was not p53-dependent. Therefore, the differential resistance to BCNU of +/+, +/-, and -/- astrocytes could not be explained by a p53-dependent cell cycle arrest, G₁ or otherwise.

Wild type p53 rendered mouse astrocytes resistant to BCNU in a gene-dosedependent fashion yet resistance could not be explained by either p53-dependent apoptosis or differential cell cycle arrest. How then does p53 render +/+ astrocytes resistant to BCNU? Perhaps the slower rate of proliferation of +/+ as compared to -/- astrocytes contributes to drug resistance; slower cell cycling in the presence of p53 would allow more time for repair of DNA damage and as a consequence enhance genomic stability in +/+ astrocytes. If this were true, one might hypothesize that -/- astrocytes would become genetically unstable in response to BCNU. This hypothesis is supported by the findings of Tobey and Crissman (1975) that demonstrated DNA polyploidy in cells treated with BCNU. Since p53-independent apoptotic pathways have been documented (Clarke *et al.*, 1993; Lotem and Sachs, 1993; Selvakumaran *et al.*, 1994), one might reason further that damage in these unstable cells can be detected by an alternative DNA damage-sensing mechanism and these cells may be triggered subsequently to undergo p53-independent programmed cell death.

CHAPTER 6

THE EFFECT OF p53 STATUS ON THE APOPTOTIC RESPONSE OF MOUSE ASTROCYTES EXPOSED TO BCNU

6.1 INTRODUCTION

The p53 protein plays a role in protecting cells and organisms from the deleterious effects of DNA damage by regulating cell cycle progression and by triggering programmed cell death (refer to sections 1.7.4.2, 1.7.4.3, and Chapter 5). In Chapter 5, wild type p53 was demonstrated to render mouse astrocytes resistant to BCNU in a gene-dose-dependent fashion. These results suggested that +/+ astrocytes were not displaying a significant amount of p53-dependent apoptosis; one would expect +/+astrocytes to be more sensitive to BCNU than -/- cells if they were preferentially undergoing programmed cell death. In addition, using flow cytometry experiments to examine cell cycle response, it was found that both +/+ and -/- mouse astrocytes were clearly able to affect cell cycle arrest in response to BCNU and therefore, this response was not p53-dependent. Thus, it seemed that the resistance displayed by +/+astrocytes in response to BCNU could not be explained by either p53-dependent apoptosis or differential cell cycle arrest. As a result, I hypothesized that perhaps the slower rate of proliferation of +/+ as compared to -/- astrocytes contributed to drug resistance; slower cell cycling in the presence of p53 would allow more time for repair of DNA damage and as a consequence enhance genomic stability in +/+ astrocytes. If this were true, I reasoned further that damage in these unstable -/- cells may be detected by an alternative DNA damage-sensing mechanism and these cells may be triggered subsequently to undergo a p53-independent programmed cell death. It was my intent, in the studies summarized in this chapter, to investigate the possibility of increased programmed cell death in -/- astrocytes in response to BCNU.

The p53 protein has been shown to influence the expression of two members of the bcl-2 family involved in the regulation of apoptosis (refer to section 1.7.4.3); the transcription of *bax* has been shown to be induced by p53 (Miyashita and Reed, 1995) and inhibition of *bcl-2* expression has been documented (Miyashita *et al.*, 1994). Mechanisms of p53-independent apoptosis do, however, exist (Clarke *et al.*, 1993; Lotem and Sachs, 1993; Selvakumaran *et al.*, 1994) as well as a number of additional members of the bcl-2 apoptotic family (Craig, 1995). One member of this family that has received much attention is bcl-x. Two alternatively spliced forms of *bcl-x* exist; the long form, bcl-x_L, enhances viability whereas the short form, bcl-x_S, acts in a dominant negative fashion to inhibit the effects of bcl-x_L (Boise *et al.*, 1993; Craig, 1995).

The purposes of this study were 1) to examine the effect of p53 status on the ability of mouse astrocytes to undergo apoptosis when exposed to BCNU; 2) to perform a semi-quantitative analysis of apoptosis in mouse cells in response to BCNU and correlate with p53 status; and 3) to explore the effect of BCNU treatment on the expression of two p53-regulated genes (*bax* and *bcl-2*) and one additional cell death gene (*bcl-x*) that may play a role in the apoptotic response to DNA damage.

6.2 RESULTS

6.2.1 Percent BAD values in mouse astrocytes after BCNU treatment or radiation.

In order to investigate the effect of p53 status on the ability of mouse astrocytes to undergo apoptosis when exposed to BCNU, I began by examining in more detail the % BAD values taken from the original flow cytometry experiments performed in Chapter 5 (refer to section 5.2.4). Percent BAD is a flow cytometry measure of *background aggregates* and *debris* (BAD) calculated by the Multicycle software program (refer to section 2.8); the % BAD calculation also takes into consideration the presence of aneuploid and apoptotic cells. In this analysis, % BAD values were significantly higher in -/- control, non-treated astrocytes than +/+ control astrocytes; -/- control astrocytes displayed a slightly greater than 4-fold increase over +/+ cells (Table 6.1). In response to treatment with BCNU or radiation, % BAD values increased significantly over control values for BCNU treated -/- astrocytes only; BCNU treated -/- cells

	p53 Genotype		
Treatment	(+/+)	(+/-)	(-/-)
Control	4.52 ± 0.86	8.92 ± 1.11 ^b	19.3 ± 2.50^{b}
BCNU	9.10 ± 4.62	7.13 ± 1.71	34.6 ± 3.48°
Radiation	7.56 ± 2.21	9.45 ± 0.83	15.1 ± 3.24

 Table 6.1
 Percent BAD^a in Astrocyte Flow Cytometry Analysis

Percent BAD values are reported for cells 48 hours post-treatment. Values were calculated using the Multicycle software. Shown is the mean \pm SEM of at least four independent experiments. Statistically significant differences from non-treated control values are noted.

- ^a Percent background aggregates and debris
- ^b P < 0.005 in comparison with +/+ control
- ^c P < 0.005 in comparison with -/- control

6.2.3 Detection of apoptosis in mouse astrocytes after BCNU treatment.

Since there appeared to be evidence for programmed cell death in -/- astrocytes, this finding was investigated further by detecting immunohistochemically the presence of apoptotic astrocytes following treatment with BCNU. Astrocytes were seeded onto coverslips, treated with BCNU, and assayed at various times after treatment for the presence of apoptotic cells with the ApopTagTM In situ Apoptosis Detection Kit (refer to section 2.9.1 for details). Positive staining for apoptotic cells was found for all three *p53* genotypes (Figure 6.1). In addition, it appeared that -/- astrocytes may be more susceptible to apoptosis than +/+ astrocytes; -/- cells tended to display more positive staining than +/+ cells at equal BCNU doses and positive staining was evident in -/- astrocytes at lower BCNU concentrations than +/+ cells (data not shown).

As evidenced in Figure 6.1, most of the cells that had undergone apoptosis seemed to lift off their substrate in culture and therefore, were not represented in the final immunohistochemical staining. Although it appeared that -/- astrocytes may be more susceptible to apoptosis in response to BCNU than +/+ cells, the ApopTagTM Kit was unsuitable for true quantitation of apoptotic cells; it was possible that damaged, but not apoptotic, cells were being preferentially washed off the coverslips during staining. In an attempt to achieve a more quantitative determination of the amount of apoptosis occurring in response to cytotoxic treatment, the commercially available Cell Death Detection ELISA kit was used (refer to section 2.9.2); this colorimetric assay measured apoptosis within a cell culture well and therefore, did not require cells to remain adherent to their substrate after treatment. Wild type and -/- astrocytes were exposed to either 20 μ g/ml BCNU or 2 Gy radiation and fibroblasts (+/+ and -/-) to 20 μ g/ml BCNU; a semi-quantitative analysis was then performed using the Cell Death Detection ELISA kit in order to determine the amount of apoptosis induced by each treatment (refer to section 2.9.2). Astrocytes exposed to BCNU showed a marked increase in the presence of apoptotic cells in comparison to non-treated controls; enrichment factors of approximately 2 to 7 were demonstrated over the three time periods sampled for the two

Figure 6.1 Detection of apoptotic mouse astrocytes after treatment with BCNU. The presence of apoptotic cells (brown DAB staining) in response to BCNU was detected using the ApopTagTM In situ Apoptosis Detection Kit as described in section 2.9.1. Non-treated cells counterstained with 0.5% methyl green in 0.1M sodium acetate (pH 4.0) are shown in the left panels and cells treated with 160 μ g/ml BCNU are shown in the right panels after a four day recovery. Apoptosis was detected in cells of all three *p53* genotypes: wild type (A); heterozygous (B); and knockout (C). Bar, 50 μ m.



cell genotypes (Figure 6.2). Little, if any, enrichment of apoptosis was evident in astrocytes exposed to radiation and fibroblasts exposed to BCNU. Of particular interest, -/- astrocytes exhibited greater amounts of apoptosis in response to BCNU than did +/+ astrocytes. These findings of increased programmed cell death in -/- astrocytes in response to BCNU confirmed what had been suggested visually by the ApopTagTM data.

6.2.4 Effect of BCNU treatment on the expression of bax, bcl-2, and bcl-x.

In order to investigate whether BCNU treatment could influence the expression of two p53-regulated genes (*bax* and *bcl-2*) and one additional cell death gene (*bcl-x*), +/+ and -/- astrocytes and fibroblasts were exposed to 40 μ g/ml BCNU for 2 hr and samples were taken at defined times for Northern analysis (refer to section 2.7). One Northern blot was probed sequentially for transcripts encoding bax and bcl-x and a second blot was probed for transcripts encoding bcl-2 (Figures 6.3 - 6.6). Two transcripts were detected for each of the three genes, as previously described (Negrini *et al.*, 1987; Oltvai *et al.*, 1993; Frankowski *et al.*, 1995), although the larger *bcl-2* transcript was barely detectable in fibroblasts. Hybridization signals were normalized to the *MGAP* signal from the same lane. Expression of all three genes remained relatively constant over the 24 hr time period in response to BCNU in all four cell types; densitometry was performed (refer to section 2.7.6) and any differences in signal intensities were not greater than 50% and were within the range of variation displayed by the control samples.

6.3 DISCUSSION

Percent BAD is a flow cytometry measure of background aggregates and debris (BAD) calculated by the Multicycle software program (refer to section 2.8); the % BAD calculation also takes into consideration the presence of aneuploid and apoptotic cells. Although generally speaking a quality control measure, % BAD values may be an important indicator of genomic instability or apoptotic cell death. In order to investigate

Figure 6.2 Semi-quantitative analysis of apoptosis in mouse cells. An attempt to quantitate the extent of apoptosis occurring in response to cytotoxic treatment was made using the Cell Death Detection ELISA as described in section 2.9.2. Enrichment factors are expressed as multiples of 1.0, where 1.0 represents the amount of apoptosis occurring in non-treated, control cells. Astrocytes displayed an enrichment of apoptotic cells in response to 20 μ g/ml BCNU (A). Knockout astrocytes exhibited greater amounts of apoptosis in response to BCNU than did wild type astrocytes (statistically significant at 12 hours; *, *P* < 0.05). Astrocytes treated with 2 Gy radiation (B) and fibroblasts treated with 20 μ g/ml BCNU (C) did not display any substantial enrichment over non-treated, control cells. Bars depict the mean of 2 or 3 independent experiments each performed in triplicate (±SEM). WT, wild type; KN, knockout; 12, 24, 48, time in hours after cytotoxic treatment.





С

B



Figure 6.3 Effect of BCNU treatment on the mRNA expression of bax, bcl-x, and bcl-2 in wild type p53 mouse astrocytes. Cells were treated with 40 μ g/ml BCNU for two hours. The first treated mRNA sample (t=0) was harvested at the end of the drug incubation and successive samples were taken at the specified times (in hours). Panels A and B represent results obtained from two different Northern blots. Expression of all three genes remained relatively constant over the 24 hr time period in response to BCNU. Cont, control; MGAP, mouse glyceraldehyde-3-phosphate dehydrogenase.



Α



Figure 6.4 Effect of BCNU treatment on the mRNA expression of bax, bcl-x, and bcl-2 in knockout p53 mouse astrocytes. Cells were treated with 40 μ g/ml BCNU for two hours. The first treated mRNA sample (t=0) was harvested at the end of the drug incubation and successive samples were taken at the specified times (in hours). Panels A and B represent results obtained from two different Northern blots. Expression of all three genes remained relatively constant over the 24 hr time period in response to BCNU. Cont, control; MGAP, mouse glyceraldehyde-3-phosphate dehydrogenase.



Α



Figure 6.5 Effect of BCNU treatment on the mRNA expression of bax, bcl-x, and bcl-2 in wild type p53 mouse fibroblasts. Cells were treated with 40 μ g/ml BCNU for two hours. The first treated mRNA sample (t=0) was harvested at the end of the drug incubation and successive samples were taken at the specified times (in hours). Panels A and B represent results obtained from two different Northern blots. Expression of all three genes remained relatively constant over the 24 hr time period in response to BCNU. Cont, control; MGAP, mouse glyceraldehyde-3-phosphate dehydrogenase.



Figure 6.6 Effect of BCNU treatment on the mRNA expression of bax, bcl-x, and bcl-2 in knockout p53 mouse fibroblasts. Cells were treated with 40 μ g/ml BCNU for two hours. The first treated mRNA sample (t=0) was harvested at the end of the drug incubation and successive samples were taken at the specified times (in hours). Panels A and B represent results obtained from two different Northern blots. Expression of all three genes remained relatively constant over the 24 hr time period in response to BCNU. Cont, control; MGAP, mouse glyceraldehyde-3-phosphate dehydrogenase.



Α



the effect of p53 status on the apoptotic response of mouse astrocytes exposed to BCNU, I examined in more detail the % BAD values taken from the flow cytometry experiments performed in Chapter 5. In this analysis, % BAD values were significantly higher in -/- control astrocytes than +/+ control astrocytes. While this difference in % BAD could be due to enhanced aggregation of -/- cells, spontaneous apoptosis or emergence of aneuploid -/- astrocytes seem more likely explanations since the loss of functional p53 is associated with genomic instability (reviewed in Hartwell, 1992). Of greater interest in these flow cytometry experiments were the % BAD values after BCNU treatment; the % BAD increased significantly for -/- astrocytes only. Polyploidy or cell death induced by BCNU would lead to high % BAD values in astrocytes; Tobey and Crissman (Tobey and Crissman, 1975) observed polyploidy after treatment with BCNU and presented data suggesting that polyploid cells and cells arrested in G₂ were among the first to die after exposure to BCNU.

The possibility of increased programmed cell death in -/- astrocytes in response to BCNU was supported qualitatively by immunohistochemical staining with the ApopTagTM In Situ Apoptosis Detection Kit and confirmed quantitatively using the Cell Death Detection ELISA. These results demonstrated that p53 was not required for apoptosis in mouse astrocytes and suggested the possibility of p53-independent apoptosis as a preferred cell death pathway for these cells in response to BCNU. To investigate the role of potential effectors in this pathway, the effect of BCNU treatment on the expression of three genes known to be involved in the apoptotic response to DNA damage was examined. The expression of all three genes, bax, bcl-2, and bcl-x, remained constant over the 24 hr period after treatment with BCNU. Therefore, differential induction of these cell death genes in response to BCNU did not explain the increased survival of +/+ astrocytes. It is interesting to note, however, in one experiment that examined basal levels of bcl-2 mRNA expression (data not shown), levels of both transcripts were approximately 5-fold higher in +/+ than -/- astrocytes; this difference in basal levels was not seen for bax and bcl-x (data not shown). This finding is of particular interest in light of the fact that, although +/+ fibroblasts also displayed

a slightly higher level of expression of the smaller bcl-2 transcript (approximately 3.5-fold), the larger transcript was barely detectable in both +/+ and -/- cells. Perhaps this increased expression of bcl-2 in non-treated astrocytes is enough to protect +/+ cells from undergoing programmed cell death in response to BCNU and therefore induction of additional bcl-2 is not required; this finding also could explain the presence of greater amounts of apoptosis in non-treated -/- astrocytes. In addition, the bcl-2 family includes many proteins and is continually expanding as new members are isolated (Farrow and Brown, 1996). Whether any of these other proteins play a role in the protection of +/+ astrocytes in response to BCNU remains to be determined.

In summary, wild type p53 was found to render mouse astrocytes resistant to BCNU in a gene-dose-dependent fashion yet this resistance could not be explained by either differential cell cycle arrest or p53-dependent apoptosis. I hypothesized that perhaps the slower rate of proliferation of +/+ as compared to -/- astrocytes contributed to drug resistance; slower cell cycling in the presence of p53 would allow more time for repair of DNA damage and as a consequence enhance genomic stability in +/+ astrocytes. If this were true, I reasoned further that damage in these unstable -/- cells may be detected by an alternative DNA damage-sensing mechanism and these cells may be triggered subsequently to undergo a p53-independent programmed cell death. It has been demonstrated by others that the presence of wild type p53 does, in fact, enhance genomic stability (reviewed in Hartwell, 1992) and the studies presented here confirmed that -/- astrocytes could undergo apoptosis both in the absence of drug and at increased levels in response to BCNU treatment. The actual mechanism for increased apoptosis in -/- mouse astrocytes in response to BCNU remains unclear and awaits further study; increased basal bcl-2 expression in +/+ astrocytes provides one possible explanation for the differential response to BCNU, as does the potential for additional members of the bcl-2 family to alter the balance of cell survival/death.

CHAPTER 7

THE EFFECT OF p53 STATUS ON O⁶-METHYLGUANINE-DNA METHYLTRANSFERASE ACTIVITY IN MOUSE ASTROCYTES

7.1 INTRODUCTION

The p53 protein plays numerous roles in protecting cells and organisms from the deleterious effects of drug-induced DNA damage. As discussed in chapters 5 and 6, p53 may function to inhibit cell cycle progression, presumably allowing time for DNA repair, or induce apoptosis in response to DNA damage. In addition, p53 may regulate directly alternate drug resistance mechanisms such as MDR or DNA repair (Chin *et al.*, 1992; Zastawny *et al.*, 1993; Nguyen *et al.*, 1994; Xiao *et al.*, 1994; Wang *et al.*, 1994; Goldsmith *et al.*, 1995; Strauss and Haas, 1995; Wang *et al.*, 1995; Leveillard *et al.*, 1996; Scherer *et al.*, 1996; Wang and Harris, 1996) (refer to section 1.7.4.4). These many functions of p53 may influence the response of human cancers to DNA-damaging therapies in a very intricate manner.

It is known that a subset of astrocytomas respond to alkylating agents but why some astrocytomas are more sensitive than others is not well understood (Nelson *et al.*, 1988; Fine *et al.*, 1993; DeAngelis *et al.*, 1996). Since chromosome 17p deletions and mutations of the p53 tumor suppressor gene are fairly common in astrocytomas (James *et al.*, 1988; el Azouzi *et al.*, 1989; James *et al.*, 1989; Fults *et al.*, 1989; Nigro *et al.*, 1989; Chung *et al.*, 1991; Hayashi *et al.*, 1991; Mashiyama *et al.*, 1991; Frankel *et al.*, 1992; Fults *et al.*, 1992; Saxena *et al.*, 1992; Sidransky *et al.*, 1992; von Deimling *et al.*, 1992a; Aka *et al.*, 1993; del Arco *et al.*, 1993; Hunter *et al.*, 1993; Louis *et al.*, 1994; Kraus *et al.*, 1994; Louis, 1994; Rasheed *et al.*, 1994), the effect of p53 status on the viability of mouse astrocytes exposed to the clinically useful, bi-functional alkylating agent BCNU was examined. Wild type p53 rendered mouse astrocytes resistant to BCNU

in a gene-dose-dependent fashion, yet resistance could not be explained by either differential cell cycle arrest or p53-dependent apoptosis (refer to chapters 5 and 6; Nutt *et al.*, 1996).

Repair of alkyl adducts at the O⁶-guanine position of DNA is mediated by O⁶methylguanine-DNA methyltransferase (MGMT). Nitrosoureas alkylate DNA at this position and the cytotoxic and carcinogenic properties of these drugs appear related to MGMT activity (Swenberg *et al.*, 1982; Ding *et al.*, 1985; Dolan *et al.*, 1985; Dolan *et al.*, 1990b; Kaina *et al.*, 1991; Wani *et al.*, 1992) (refer to section 1.5.2). Russell *et al.* (1995) found that MGMT levels correlated negatively with response to procarbazine, another clinically useful alkylating agent, in primary human brain tumors and glioma cell lines when grown as xenografts. In addition, they found a strong trend toward an increased p53 mutation incidence in tumors with increased growth delay in procarbazinetreated xenografts and no MGMT activity. More recently, there have been studies to suggest that p53 may play a role in the regulation of MGMT activity (Harris *et al.*, 1996; Rafferty *et al.*, 1996) (refer to section 1.7.4.4). Since MGMT-mediated repair has been shown to play a role in resistance to alkylating agents, the effect of p53 status on MGMT activity in mouse astrocytes was examined.

The purposes of this study were 1) to determine the effect of p53 status on MGMT protein and mRNA levels; 2) to investigate whether depletion of MGMT protein with BG could sensitize mouse astrocytes to BCNU; and 3) to explore the possibility of p53-dependent changes in MGMT levels in response to BCNU treatment.

7.2 RESULTS

7.2.1 Effect of p53 status on MGMT protein and mRNA in mouse astrocytes and fibroblasts.

To determine the effect of p53 on MGMT activity, astrocytes and fibroblasts were isolated from mice with distinct p53 genotypes (+/+, +/-, -/-) and MGMT levels were measured in each cell type (as described in sections 2.3.3 and 2.6 respectively). Homozygous wild type p53 astrocytes displayed significantly greater MGMT activity than

Figure 7.1 Effect of p53 on MGMT activity in mouse astrocytes (A) and fibroblasts (B). Homozygous wild type (+/+) p53 astrocytes displayed significantly greater MGMT activity than homozygous knockout (-/-)astrocytes while heterozygous (+/-) cells exhibited intermediate activity. There was no significant difference in MGMT activity between +/+ and +/- fibroblasts but both displayed significantly higher activity than -/fibroblasts. Bars depict the mean of at least eight replicates from a minimum of two independent experiments (\pm SEM) and asterisks the level of statistical significance (*, P < 0.05; **, P < 0.005). WT, wild type; HET, heterozygous; KN, knockout.







-/- astrocytes while +/- cells exhibited intermediate activity (Figure 7.1A). Hence, p53 appeared to enhance MGMT activity in a gene-dose-dependent manner in astrocytes. In comparison, a gene-dose-dependent relationship was not evident in fibroblasts; there was no significant difference in MGMT activity between +/+ and +/- cells (Figure 7.1B). However, +/+ and +/- fibroblasts both displayed significantly greater activity than -/- fibroblasts (Figure 7.1B) suggesting that MGMT activity was enhanced in the presence of p53. Although +/+ levels were comparable in astrocytes and fibroblasts, -/- astrocytes exhibited half the MGMT activity of -/- fibroblasts.

The effect of p53 on MGMT mRNA expression also was examined. Astrocytes and fibroblasts were isolated and mRNA was analyzed as described in section 2.7. Northern blot procedures were performed twice and, consistent with the MGMT activity results, +/+ cells displayed higher levels of MGMT mRNA than -/- cells (data not shown). Knockout fibroblasts also displayed consistently higher MGMT mRNA levels than did -/- astrocytes (data not shown).

7.2.2 Effect of BG pretreatment on the response of mouse astrocytes and fibroblasts to BCNU.

Since MGMT activity correlated with response to BCNU in mouse astrocytes (as described in section 5.2.1), cells were pretreated with BG to determine if depletion of MGMT would increase sensitivity to BCNU. First, cells were exposed to BG and MGMT activity was measured to ensure that BG would deplete MGMT. Astrocytes and fibroblasts were treated with either 25 μ M or 50 μ M BG for two hours, cells were harvested and MGMT activity was measured (refer to section 2.6 for details). BG treatment significantly depleted MGMT activity in +/+ p53 astrocytes and fibroblasts and -/-p53 fibroblasts (Figure 7.2). There was no significant depletion of the already low MGMT activity displayed by -/- astrocytes; the basal MGMT level found in -/- astrocytes was not significantly different from those levels found in the more MGMT-positive cells following depletion of greater than 90% of MGMT activity.

To determine if depletion of MGMT would sensitize cells to BCNU, astrocytes and fibroblasts were pretreated with 25 μ M or 50 μ M BG for 2 hr, exposed to increasing Figure 7.2 Effect of BG treatment on MGMT activity in mouse astrocytes and fibroblasts. Cells were treated with either 25 or 50 μ M BG for two hours prior to the measurement of MGMT activity. BG treatment significantly depleted MGMT activity in wild type *p53* astrocytes and fibroblasts and knockout *p53* fibroblasts. There was no significant depletion of the already low MGMT activity displayed by knockout astrocytes. Control MGMT levels from Figure 7.1 are shown for comparison. Bars depict the mean of at least two replicates from a minimum of two independent experiments (±SEM) and asterisks the level of statistical significance (*, *P* < 0.05; **, *P* < 0.005). WT, wild type; HET, heterozygous; KN, knockout; A, astrocytes; F. fibroblasts.



concentrations of BCNU and cell viability tested using the MTT assay (refer to section 2.5 for details). BG did not significantly increase sensitivity to BCNU in any of the combinations tested. In addition to the +/+ astrocytes and fibroblasts and -/- astrocytes and fibroblasts (Figures 7.3 and 7.4 respectively), +/- cells also were tested and an increased sensitivity to BCNU was not evident (data not shown).

7.2.3 Effect of BCNU and BG treatment on MGMT levels.

Since there have been reports of MGMT induction in response to treatment with DNA-damaging agents (Margison *et al.*, 1985; Dexter *et al.*, 1989; Stammberger *et al.*, 1990; Fritz *et al.*, 1991; Stammberger and Tempel, 1991; Chan *et al.*, 1992; Wilson *et al.*, 1993; Grombacher *et al.*, 1996), cells were treated with BCNU to determine if there were any changes in MGMT levels. First, cells were examined for changes in *MGMT* mRNA expression. Wild type and -/- astrocytes and fibroblasts were exposed to either 40 μ g/ml BCNU or 25 μ M BG for 2 hr and samples were taken at defined times for Northern analysis (refer to section 2.7). There was no evidence of *MGMT* mRNA induction; expression remained constant over a 24 hr time period in response to both treatments in all four cell types (Figure 7.5).

In addition to exploring mRNA levels, cells also were examined for changes in MGMT activity following treatment with BCNU. There was no significant change in MGMT activity in +/+ and -/- astrocytes in response to BCNU (Figure 7.6). There was an indication of a slight, but non-significant, decrease in MGMT activity in +/+ astrocytes which may be evidence of the inactivation of some MGMT molecules following repair of BCNU-induced DNA damage. This conclusion is supported by the finding that a reduction in activity was not seen in response to treatment with 5 Gy radiation (Figure 7.6); radiation does not alkylate DNA at the O⁶ position of guanine. Furthermore, there was no evidence of a substantial induction of MGMT activity in fibroblasts (Figure 7.7). The same pattern of a slight decrease in MGMT activity in response to BCNU was seen for +/+ fibroblasts and although an increasing trend is seen in -/- fibroblasts, the levels of MGMT activity do not increase to the levels displayed

Figure 7.3 Effect of BG pretreatment on the response of wild type p53 (A) and knockout (B) mouse astrocytes to BCNU. Cells were pretreated with either 25 or 50 μ M BG for two hours and then exposed to varying concentrations of BCNU. BG did not significantly increase sensitivity to BCNU in any of the combinations tested. Symbols represent the mean of at least two independent experiments, each performed in triplicate. Circles represent cells treated with BCNU alone; triangles, 25 μ M BG pretreatment; squares, 50 μ M BG pretreatment. Bars depict the standard error of the mean.



Figure 7.4 Effect of BG pretreatment on the response of wild type p53 (A) and knockout (B) mouse fibroblasts to BCNU. Cells were pretreated with either 25 or 50 μ M BG for two hours and then exposed to varying concentrations of BCNU. BG did not significantly increase sensitivity to BCNU in any of the combinations tested. Symbols represent the mean of at least two independent experiments, each performed in triplicate. Circles represent cells treated with BCNU alone; triangles, 25 μ M BG pretreatment; squares, 50 μ M BG pretreatment. Bars depict the standard error of the mean.


Figure 7.5 Effect of BCNU and BG treatment on MGMT mRNA expression. Cells were exposed to either 40 μ g/ml BCNU or 25 μ M BG for two hours. The first treated mRNA sample (t=0) was harvested at the end of the incubation and succeeding samples were taken at the specified times (in hours). Quantified signal intensities for MGMT were normalized to MGAP in wild type p53 mouse astrocytes (A), knockout astrocytes (B), wild type fibroblasts (C), and knockout fibroblasts (D). MGMT mRNA expression remained constant throughout both treatments in all four cell types. Cont, control; MGAP, mouse glyceraldehyde-3-phosphate dehydrogenase.



Figure 7.6 Effect of BCNU treatment and radiation on MGMT activity in wild type p53 (A) and knockout (B) mouse astrocytes. Cells were treated with either 40 μ g/ml BCNU for two hours or 5 Gy radiation. MGMT activity was measured at successive time points as indicated (in hours). There was no significant change in MGMT activity in response to treatment. Bars depict the mean of at least four replicates from a minimum of two independent experiments (±SEM). Gy, grays; cont, control; rad, radiation.



В



Figure 7.7 Effect of BCNU treatment and radiation on MGMT activity in wild type p53 (A) and knockout (B) mouse fibroblasts. Cells were treated with either 40 μ g/ml BCNU for two hours or 5 Gy radiation. MGMT activity was measured at successive time points as indicated (in hours). Although there was some statistically significant changes in MGMT activity in response to treatment, the magnitude of the changes were quite small and there was no evidence of a substantial induction in MGMT activity. Bars depict the mean of at least four replicates from a minimum of two independent experiments (±SEM) and asterisks the level of statistical significance (*, P < 0.05; **, P < 0.005). Gy, grays; cont, control; rad, radiation.





 \mathbf{D}

 \triangleright

by +/+ cells. Heterozygous cells also were tested at the 24 and 48 time points and there was no change in MGMT activity (data not shown).

7.3 DISCUSSION

The p53 protein plays numerous roles in protecting cells and organisms from the deleterious effects of drug-induced DNA damage, one of which may be to regulate the expression of alternate drug resistance mechanisms. Since MGMT-mediated repair has been shown to play a role in resistance to alkylating agents and recently, there have been studies to suggest that p53 may be involved in the regulation of MGMT activity, the effect of p53 status on MGMT activity in mouse astrocytes was examined. Homozygous wild type p53 astrocytes displayed significantly greater MGMT activity than -/astrocytes while +/- cells exhibited intermediate activity. Hence, p53 appeared to enhance MGMT activity in a gene-dose-dependent manner in astrocytes. These results correlated well with the p53 gene-dose-dependent resistance to BCNU displayed by mouse astrocytes. A gene-dose-dependent relationship for MGMT activity was not evident in fibroblasts; although +/+ and +/- fibroblasts both displayed significantly greater activity than -/- fibroblasts, there was no significant difference in MGMT activity between +/+ and +/- cells. Of particular interest, +/+ levels were comparable in astrocytes and fibroblasts but -/- astrocytes exhibited half the MGMT activity of -/- fibroblasts; basal levels of MGMT activity in -/- astrocytes could be considered MGMT-negative as they were not significantly different from those levels found in the MGMT-positive cells following depletion of greater than 90% of MGMT activity. Perhaps MGMT was contributing to the resistance of +/+ mouse cells to BCNU and only -/- astrocytes were sensitive due to their apparent lack of MGMT activity. This would explain the differential response exhibited by -/- astrocytes and fibroblasts in response to BCNU; -/- fibroblasts did not appear to display the same sensitivity to BCNU as did -/- astrocytes (refer to section 5.2.2).

Since MGMT activity correlated with response to BCNU in mouse astrocytes, cells were pretreated with BG to determine if depletion of MGMT would increase sensitivity to BCNU. Although BG pretreatment was shown to deplete MGMT activity

in +/+ astrocytes and fibroblasts and -/- fibroblasts, BG did not significantly increase sensitivity to BCNU in any of the combinations tested. These results suggested that although p53 status could affect basal levels of MGMT in astrocytes and fibroblasts, MGMT was not playing a role in the resistance of these cells to BCNU.

MGMT has been shown to be induced in response to some DNA damaging treatments (Margison *et al.*, 1985; Dexter *et al.*, 1989; Stammberger *et al.*, 1990; Fritz *et al.*, 1991; Stammberger and Tempel, 1991; Chan *et al.*, 1992; Wilson *et al.*, 1993; Grombacher *et al.*, 1996); it was possible that MGMT activity was being induced in response to either BG or BCNU and therefore, any attempts to deplete MGMT were, in turn, being overcome by the cells. This possibility was excluded as induction of neither MGMT mRNA nor protein levels was found after BG and BCNU treatment.

In summary, p53 appeared to enhance MGMT activity in a gene-dose-dependent manner in astrocytes. Although these results correlated well with the p53 gene-dosedependent resistance to BCNU displayed by mouse astrocytes, depletion of MGMT by BG pretreatment did not significantly increase sensitivity to BCNU. This inability to increase sensitivity was not due to the subsequent induction of additional MGMT protein. Thus, although these results support a role for p53 in the regulation of MGMT, the p53dependent BCNU resistance displayed by mouse astrocytes appears to be mediated by a non-MGMT mechanism. The significance of these results are not lost however; Preuss *et al.* (1996) recently suggested that the protective effect of MGMT varied depending on the alkylating agent in question, with BCNU being less affected by MGMT modulation than other alkylating agents of this class. It would be of interest, therefore, to investigate whether modulation of the p53-dependent differential expression of MGMT demonstrated in this study could confer sensitivity to additional alkylating agents.

CHAPTER 8

SUMMARY AND CONCLUSIONS

Despite intensive research, attempts to develop effective chemotherapy regimens for gliomas have met with limited success. Significant progress was made when Cairncross and Macdonald (1988) reported that anaplastic oligodendrogliomas responded to a chemotherapy regimen of procarbazine, CCNU, and vincristine (PCV). It has been confirmed since that oligodendrogliomas and mixed oligoastrocytomas respond predictably to nitrosourea-based chemotherapy, especially PCV, whereas astrocytomas are relatively chemoresistant. Presumably oligodendroglioma cells are inherently susceptible to the cytotoxic effects of PCV, but why? In addition, a subset of astrocytomas have been observed to benefit from chemotherapy. What intrinsic properties of this subset of astrocytomas set them apart from the majority of astrocytic tumors? The answers to these questions are undoubtedly important as they may lead to better therapies for oligodendrogliomas and to new treatment strategies for anaplastic astrocytomas and glioblastomas.

In this thesis, two main approaches were taken to test the hypothesis that biochemical properties intrinsic to specific subtypes of glial tumors play a role in the resistance of these tumors to chemotherapeutic agents. The first approach was to examine biochemical properties inherent to the normal glial cell of origin. In cultures of semi-pure glial cells isolated from rat, oligodendrocyte lineage cells were found to be significantly more sensitive to BCNU than astrocytes. In addition, oligodendrocyte lineage cells had low MGMT activity, approximately one-third of the levels found for astrocytes. In cultures of purified rat glial cells, purified oligodendrocytes also were demonstrated to be more sensitive to BCNU than astrocytes, while oligodendrocyte progenitors and astrocytes displayed no significant difference in response to BCNU. When a panel of six drug resistance genes (*MGMT*, *GST*- μ , *GST*- π , *p53*, *MDR*, and *MT*) was probed for expression in these cells, a differential pattern of mRNA expression was observed for each of the six probes. The roles of MGMT and the glutathione-glutathione-S-transferase

detoxification pathway in response to BCNU were examined further. Depletion of both MGMT and glutathione were demonstrated to increase sensitivity to BCNU in astrocytes, suggesting a role for these two mechanisms in the resistance of rat glial cells to alkylating agents. The evidence provided from the differential expression of drug resistance genes suggested that more than one mechanism of resistance may be deficient in oligodendrocytes. In addition, numerous other mechanisms exist that may play a role in drug resistance in these cells but were not explored in this study (refer to sections 1.5 and 1.6). Thus, the response of normal rat glial cells to BCNU appeared to be dependent on more than one mechanism of drug resistance. If mechanisms of drug resistance in some subtypes of glial tumors reflect the biochemical properties of their cell of origin, then these experiments provide additional support for the hypothesis that oligodendroglioma cells may be inherently susceptible to the cytotoxic effects of alkylating agents.

The second approach of this thesis was to examine the impact of a specific genetic alteration, found commonly in astrocytic gliomas, on the response of otherwise normal astrocytes to BCNU. Mutations of the p53 gene, and subsequent loss of p53 function, are a common occurrence in astrocytic tumors. Therefore, the impact of the loss of p53 function on response of astrocytes to BCNU was examined using cells isolated from TSG-p53 mice (refer to section 2.3.1). Wild type p53 appeared to protect mouse astrocytes from the cytotoxic effects of BCNU in a gene-dose-dependent fashion. These results suggested that +/+ astrocytes were not undergoing p53-dependent apoptosis; one would expect +/+ astrocytes to be more sensitive to BCNU than -/- cells if they were preferentially undergoing programmed cell death. In addition, when the effect of p53 status on cell cycle response to BCNU treatment was examined, mouse astrocytes were shown to undergo cell cycle arrest in response to BCNU but this response was not p53-dependent. Thus, it seemed that the resistance displayed by +/+ astrocytes in response to BCNU could not be explained by either p53-dependent apoptosis or differential cell cycle arrest.

How then would p53 render +/+ astrocytes resistant to BCNU? It was hypothesized that perhaps the slower rate of proliferation of +/+ as compared to -/-

astrocytes contributed to drug resistance; slower cell cycling in the presence of p53 would allow more time for repair of DNA damage and as a consequence enhance genomic stability in +/+ astrocytes. If this were true, it was reasoned further that damage in these unstable -/- cells may be detected by an alternative DNA damage-sensing mechanism and these cells may be triggered subsequently to undergo a p53-independent programmed cell death. The studies presented here confirmed that -/- astrocytes could undergo apoptosis both in the absence of drug and at increased levels in response to BCNU treatment. The actual mechanism for increased apoptosis in -/- mouse astrocytes in response to BCNU remains unclear and awaits further study; increased basal *bcl-2* expression in +/+ astrocytes provides one possible explanation for the differential response to BCNU, as does the potential for additional members of the bcl-2 family to alter the balance of cell survival/death.

Finally, it has been suggested that p53 may regulate directly alternate drug resistance mechanisms. Since MGMT-mediated repair has been shown to play a role in resistance to alkylating agents and recently, there have been studies to suggest that p53 may be involved in the regulation of MGMT activity, the effect of p53 status on MGMT activity in mouse astrocytes was examined. The presence of p53 appeared to enhance MGMT activity in a gene-dose-dependent manner in astrocytes. Although these results correlated well with the p53 gene-dose-dependent resistance to BCNU displayed by mouse astrocytes, depletion of MGMT with BG did not sensitize these cells to BCNU and thus, the p53-dependent BCNU resistance displayed by mouse astrocytes appeared to be mediated by a non-MGMT mechanism. The finding that mouse astrocytes were not sensitized to BCNU following BG pretreatment was of particular interest since, in Chapter 4, rat astrocytes were found to be sensitized, at least to some degree, following the same treatment regimen. These results suggested that some inter-species variation may exist in regard to preferred mechanisms of choice for drug resistance. This variability has been documented previously for MGMT; Gerson et al. (1986) showed that, compared with human and rat tissues, all mouse tissues contained low levels of MGMT activity.

Why might some gliomas benefit from chemotherapy treatment whereas others

do not? Taken together, the results presented in this thesis suggest that properties inherent to the normal glial cell of origin and specific genetic alterations occurring during tumorigenesis both may play a role in the development of a resistant tumor. In conclusion, these results provide support for the hypothesis that biochemical properties intrinsic to specific subtypes of glial tumors play a role in the resistance of these tumors to chemotherapeutic agents. APPENDIX I

CERTIFICATION OF APPROVAL OF ANIMAL RESEARCH



MAN 20 1992

172

The UNIVERSITY of WESTERN ONTARIO

Council on Animal Care • Animal Care and Veterinary Services Director - Michele M. Bailey, D.V.M. Clinical Veterinarian - Susan H. Fussell, D.V.M.

January 15, 1992

Dear Dr. Cairncross:

Your "Application To Use Animals for Research or Teaching" entitled:

"DNA Methyltransferase and Glutathlone Transferase Activity in Rat Glial Cell Cultures"

has been approved by the University Council on Animal Care. This approval expires in one year on the last day of the month. The number for this project is <u># 92013-1</u>

- 1. This number must be indicated when ordering animals for this project.
- 2. Animals for other projects may not be ordered under this number.
- 3. <u>If no number appears on this approval please contact this office when grant approval is</u> received. If the application for funding is not successful and if you wish to proceed with the project, request that an internal scientific peer review be performed by your animal care committee.
- 4. Purchases of animals other than through this system must be cleared through the ACVS office. Health certificates will be required.

ANIMALS APPROVED

Rats - S.D. 60

REQUIREMENTS/COMMENTS

Woily

c.c. A. Parbtani P. Coakwell



The UNIVERSITY of WESTERN ONTARIO

Council on Animal Care • Animal Care and Veterinary Services Director • Michele M. Bailéy, D.V.M. Clinical Veterinarian • Susan H. Fussell, D.V.M.

September 11, 1992

Dear Dr. Cairncross:

A <u>MODIFICATION</u> to your "Application to Use Animals for Research or Teaching" entitled:

"DNA Methyltransferase and Blutathione Transferase Activity in Rat Glial Cell Cultures"

has been approved. The protocol <u>#92013-1</u> and expiry date of <u>January</u> <u>1993</u> remains unchanged.

APPROVED CHANGES INCLUDE:

----- 1. Name change to " "

X ----- 2. Animal use change(s)

> Within Year Modification to ADD 7 - 14 day old rats. 40 littco

----- 3. Procedural

----- 4. Other

X 5. Comments

Rats greater than 7 days old will be euthanized with CO.

c.c. A. Parbtani P. Coakwell

/Kaie

Health Sciences Centre • London, Ontario • Canada • N6A 5C1 • Telephone: (519) 661-3034 • Fax: (519) 661-3797



The UNIVERSITY of WESTERN ONTARIO

Council on Animal Care • Animal Care and Veterinary Services Director - Michele M. Bailey, D.V.M. Clinical Veterinarian - Susan H. Fussell, D.V.M.

March 3, 1993

Dear Dr. Cairncross:

Your "Application To Use Animals for Research or Teaching" entitled:

"DNA Methyltransferase and Glutathione Transferase Activity in Glial Cell Cultures"

has been approved by the University Council on Animal Care. This approval expires in one year on the last day of the month. The number for this project is <u># 93050-3</u>

- 1. This number must be indicated when ordering animals for this project.
- 2. Animals for other projects may not be ordered under this number.
- 3. <u>if no number appears on this approval please contact this office when grant approval is</u> <u>received.</u> If the application for funding is not successful and if you wish to proceed with the project, request that an internal scientific peer review be performed by your animal care <u>committee</u>.
- 4. Purchases of animals other than through this system must be cleared through the ACVS office. Health certificates will be required.

ANIMALS APPROVED

Rats - S.D. - 150 litters

REQUIREMENTS/COMMENTS

C.C. A. Parbtani P. Coakwell 11 -

Health Sciences Centre • London, Ontario • Canada • N6A 5C1 • Telephone: (519) 661-3034 • Fax: (519) 661-2028



MAR 20 1904

175

The UNIVERSITY of WESTERN ONTARIO

Council on Animal Care • Animal Care and Veterinary Services Director - Michele M. Bailey, D.V.M. Clinical Veterinarian - Susan H. Fussell, D.V.M.

March 18, 1994

Dear Dr. Cairncross:

Your "Application to Use Animals for Research or Teaching" entitled:

"DNA Methyltransferase and Glutathione Transferase Activity in Glial Cell Cultures"

has been approved by the University Council on Animal Care. This approval expires in one year on the last day of the month. The number for this project is <u># 94075-3</u>

- 1. This number must be indicated when ordering animals for this project.
- 2. Animals for other projects may not be ordered under this number.
- 3. If no number appears on this approval please contact this office when grant approval is received. If the application for funding is not successful and if you wish to proceed with the project, request that an internal scientific peer review be performed by your animal care committee.
- 4. Purchases of animals other than through this system must be cleared through the ACVS office. Health certificates will be required.

ANIMALS APPROVED

Rats - S.D. - 150 litters

REQUIREMENTS/COMMENTS

Please ensure that individual(s) performing procedures, as described in this protocol, are familiar with the contents of this document.

A. Parbtani C.C. P. Coakwell

Health Sciences Centre • London, Ontario • Canada • N6A 5C1 • Telephone: (519) 661-3034 • Fax: (519) 661-2028



The UNIVERSITY of WESTERN ONTARIO

Council on Animal Care • Animal Care and Veterinary Services Director • Michele M. Bailey, D.V.M. Clinical Veterinarian • Susan H. Fussell, D.V.M.

May 1, 1995

Dear Dr. Cairncross:

Your "Application to Use Animals for Research or Teaching" entitled:

"DNA Methltransferase and Glutathione Transferase Activity in Glial Cell Cultures Funding Agency: <u>LRCC Endowment Fund – A572</u>

has been approved by the University Council on Animal Care. This approval expires in one year on the last day of the month. The number for this project is <u># 95084-5</u>

- 1. This number must be indicated when ordering animals for this project.
- 2. Animals for other projects may not be ordered under this number.
- 3. If no number appears on this approval please contact this office when grant approval is received. If the application for funding is not successful and if you wish to proceed with the project, request that an internal scientific peer review be performed by your animal care committee.
- 4. Purchases of animals other than through this system must be cleared through the ACVS office. Health certificates will be required.

ANIMALS APPROVED

Rats - S.D. - 125 litters Mice - TSG-p53 (C57BI) - 150 litters

REQUIREMENTS/COMMENTS

Please ensure that individual(s) performing procedures, as described in this protocol, are familiar with the contents of this document.

- 1. SOP#320-01 "Euthanasia" applies to this protocol. If you do not already have a copy they are available from Mr., Paul Coakwell.
- 2. Inaccurate protocol numbers in the body of your protocol have been corrected.
- 3. Pups less than 7 days old must be decapitated with scissors.
- c.c. A. Parbtani P. Coakwell Office of Research Services

J. Marten



٩

•

April 15, 1996

• •

Dear Dr. Caimcnossi

Your "Application to Use Animals for Research or Teaching" entitled:

"Mechanisms of Drug Resistance in Glial Cells" Funding Agency: <u>LRCC Endowment Fund</u>

has been approved by the University Council on Animal Care. This approval expires in one year on the last day of the month. The number for this project is $\sqrt{\frac{95080-5}{5000-5}}$

1. This number must be indicated when ordering animals for this project.

- 2. Animals for other projects may not be ordered under this number.
- 3. If no number appears on this approval please contact this office when grant approval is received. If the application for funding is not successful and if you wish to proceed with the project, request that an internal scientific peer review be performed by your animal care committee.
- 4. Purchases of animals other than through this system must be cleared through the ACVS office. Health certificates will be required.

ANIMALS APPROVED

REQUIREMENTS/COMMENTS

Please ensure that individual(s) performing procedures, as described in this protocol, are familian with the contents of this document.

c.c. Approved Reneval – J. Cairneross, P. Coakwell Approval Letter – P. Coakwell

REFERENCES

Abe, T., Hasegawa, S., Taniguchi, K., Yokomizo, A., Kuwano, T., Ono, M., Mori, T., Hori, S., Kohno, K., and Kuwano, M. (1994) Possible involvement of multidrug-resistance-associated protein (MRP) gene expression in spontaneous drug resistance to vincristine, etoposide and adriamycin in human glioma cells. *Int. J. Cancer* **58**: 860-864.

Abe, T., Koike, K., Ohga, T., Kubo, T., Wada, M., Kohno, K., Mori, T., Hidaka, K., and Kuwano, M. (1995) Chemosensitisation of spontaneous multidrug resistance by a 1,4-dihydropyridine analogue and verapamil in human glioma cell lines overexpressing MRP or MDR1. *Br. J. Cancer* 72: 418-423.

Abney, E.R., Williams, B.P., and Raff, M.C. (1983) Tracing the development of oligodendrocytes from precursor cells using monoclonal antibodies, fluorescence-activated cell sorting, and cell culture. *Dev. Biol.* 100: 166-171.

Agresti, C., Aloisi, F., and Levi, G. (1991) Heterotypic and homotypic cellular interactions influencing the growth and differentiation of bipotential oligodendrocyte-type-2 astrocyte progenitors in culture. *Dev. Biol.* 144: 16-29.

Ahmad, S., Okine, L., Le, B., Najarian, P., and Vistica, D.T. (1987) Elevation of glutathione in phenylalanine mustard-resistant murine L1210 leukemia cells. *J. Biol. Chem.* 262: 15048-15053.

Aida, T. and Bodell, W.J. (1987) Effect of caffeine on cytotoxicity and sister chromatid exchange induction in sensitive and resistant rat brain tumor cells treated with 1,3-bis(2-chloroethyl)-1-nitrosourea. *Cancer Res.* 47: 5052-5058.

Aka, K., Bruner, J.M., Bondy, M.L., Ligon, K., Nishi, T., del Giglio, A., Moser, R.P., Levin, V.A., and Saya, H. (1993) Detection of p53 alterations in human astrocytomas using frozen tissue sections for the polymerase chain reaction. *J. Neurooncol.* 16: 125-133.

Ali Osman, F., Stein, D.E., and Renwick, A. (1990) Glutathione content and glutathione-S-transferase expression in 1,3-bis(2-chloroethyl)-1-nitrosourea-resistant human malignant astrocytoma cell lines. *Cancer Res.* 50: 6976-6980.

Ali Osman, F., Berger, M.S., Rajagopal, S., Spence, A., and Livingston, R.B. (1993) Topoisomerase II inhibition and altered kinetics of formation and repair of nitrosourea and cisplatin-induced DNA interstrand cross-links and cytotoxicity in human glioblastoma cells. *Cancer Res.* 53: 5663-5668. Ali-Osman, F. (1989) Quenching of DNA cross-link precursors of chloroethylnitrosoureas and attenuation of DNA interstrand cross-linking by glutathione. *Cancer Res.* **49:** 5258-5261.

Ali-Osman, F., Caughlan, J., and Gray, G.S. (1989) Decreased DNA interstrand cross-linking and cytotoxicity induced in human brain tumor cells by 1,3-bis(2-chloroethyl)-1-nitrosourea after *in vitro* reaction with glutathione. *Cancer Res.* **49:** 5954-5958.

Alwine, J.C., Kemp, D.J., and Stark, G.R. (1977) Method for detection of specific RNAs in agarose gels by transfer to diazobenzyloxymethyl-paper and hybridization with DNA probes. *Proc. Natl. Acad. Sci. U. S. A.* 74: 5350-5354.

Antoniades, H.N., Galanopoulos, T., Neville Golden, J., and Maxwell, M. (1992) Expression of insulin-like growth factors I and II and their receptor mRNAs in primary human astrocytomas and meningiomas; in vivo studies using in situ hybridization and immunocytochemistry. *Int. J. Cancer* **50**: 215-222.

Arai, N., Nomura, D., Yokota, K., Wolf, D., Brill, E., Shohat, O., and Rotter, V. (1986) Immunologically distinct p53 molecules generated by alternative splicing. *Mol. Cell. Biol.* **6**: 3232-3239.

Arap, W., Nishikawa, R., Furnari, F.B., Cavenee, W.K., and Huang, H.J. (1995) Replacement of the p16/CDKN2 gene suppresses human glioma cell growth. *Cancer Res.* 55: 1351-1354.

Asano, T., An, T., Mayes, J., Zwelling, L.A., and Kleinerman, E.S. (1996a) Transfection of human topoisomerase II alpha into etoposide-resistant cells: transient increase in sensitivity followed by down-regulation of the endogenous gene. *Biochem. J.* **319**: 307-313.

Asano, T., Zwelling, L.A., An, T., McWatters, A., Herzog, C.E., Mayes, J., Loughlin, S.M., and Kleinerman, E.S. (1996b) Effect of transfection of a Drosophila topoisomerase II gene into a human brain tumour cell line intrinsically resistant to etoposide. *Br. J. Cancer* 73: 1373-1380.

Babson, J.R., Abell, N.S., and Reed, D.J. (1981) Protective role of the glutathione redox cycle against adriamycin-mediated toxicity in isolated hepatocytes. *Biochem. Pharmacol.* 30: 2299-2304.

Babson, J.R. and Reed, D.J. (1978) Inactivation of glutathione reductase by 2-chloroethyl nitrosourea-derived isocyanates. *Biochem. Biophys. Res. Commun.* 83: 754-762.

Baer, J.C., Freeman, A.A., Newlands, E.S., Watson, A.J., Rafferty, J.A., and

Margison, G.P. (1993) Depletion of O6-alkylguanine-DNA alkyltransferase correlates with potentiation of temozolomide and CCNU toxicity in human tumour cells. *Br. J. Cancer* 67: 1299-1302.

Bakalkin, G., Yakovleva, T., Selivanova, G., Magnusson, K.P., Szekely, L., Kiseleva, E., Klein, G., Terenius, L., and Wiman, K.G. (1994) p53 binds single-stranded DNA ends and catalyzes DNA renaturation and strand transfer. *Proc. Natl. Acad. Sci. U. S. A.* **91:** 413-417.

Baker, S.J., Markowitz, S., Fearon, E.R., Willson, J.K., and Vogelstein, B. (1990) Suppression of human colorectal carcinoma cell growth by wild-type p53. *Science* 249: 912-915.

Barak, Y., Juven, T., Haffner, R., and Oren, M. (1993) mdm2 expression is induced by wild type p53 activity. *EMBO J.* 12: 461-468.

Bargonetti, J., Manfredi, J.J., Chen, X., Marshak, D.R., and Prives, C. (1993) A proteolytic fragment from the central region of p53 has marked sequence-specific DNA-binding activity when generated from wild-type but not from oncogenic mutant p53 protein. *Genes Dev.* 7: 2565-2574.

Barranco, S.C. and Humphrey, R.M. (1971) The effects of 1,3-bis(2-chloroethyl)-1-nitrosourea on survival and cell progression in Chinese hamster cells. *Cancer Res.* 31: 191-195.

Barres, B.A., Hart, I.K., Coles, H.S., Burne, J.F., Voyvodic, J.T., Richardson, W.D., and Raff, M.C. (1992) Cell death and control of cell survival in the oligodendrocyte lineage. *Cell* **70**: 31-46.

Bartlett, P.F., Noble, M.D., Pruss, R.M., Raff, M.C., Rattray, S., and Williams, C.A. (1981) Rat neural antigen-2 (RAN-2): a cell surface antigen on astrocytes, ependymal cells, Muller cells and lepto-meninges defined by a monoclonal antibody. *Brain Res.* 204: 339-351.

Bates, S. and Vousden, K.H. (1996) p53 in signaling checkpoint arrest or apoptosis. Curr. Opin. Genet. Dev. 6: 12-19.

Baudier, J., Delphin, C., Grunwald, D., Khochbin, S., and Lawrence, J.J. (1992) Characterization of the tumor suppressor protein p53 as a protein kinase C substrate and a S100b-binding protein. *Proc. Natl. Acad. Sci. U. S. A.* 89: 11627-11631.

Beck, W.T. and Dalton, W.S. (1997) Pharmacology of cancer chemotherapy. Section 11: Mechanisms of drug resistance. In: *Cancer: Principles & Practice of Oncology*, edited by DeVita, V.T., Jr., Hellman, S., and Rosenberg, S.A. Philadelphia, PA: Lippincott-Raven, pp. 498-512.

Beith, J., Hartley, J., Darling, J., and Souhami, R. (1997) DNA interstrand cross-linking and cytotoxicity induced by chloroethylnitrosoureas and cisplatin in human glioma cell lines which vary in cellular concentration of O6-alkylguanine-DNA alkyltransferase. *Br. J. Cancer* **75**: 500-505.

Belanich, M., Pastor, M., Randall, T., Guerra, D., Kibitel, J., Alas, L., Li, B., Citron, M., Wasserman, P., White, A., Eyre, H., Jaeckle, K., Schulman, S., Rector, D., Prados, M., Coons, S., Shapiro, W., and Yarosh, D. (1996) Retrospective study of the correlation between the DNA repair protein alkyltransferase and survival of brain tumor patients treated with carmustine. *Cancer Res.* 56: 783-788.

Bello, M.J., de Campos, J.M., Kusak, M.E., Vaquero, J., Sarasa, J.L., Pestana, A., and Rey, J.A. (1994) Molecular analysis of genomic abnormalities in human gliomas. *Cancer Genet. Cytogenet.* **73:** 122-129.

Benchimol, S., Pim, D., and Crawford, L. (1982) Radioimmunoassay of the cellular protein p53 in mouse and human cell lines. *EMBO J.* 1: 1055-1062.

Berger, M.S., Leibel, S.A., Bruner, J.M., Finlay, J.L., and Levin, V.A. (1996) Primary central nervous system tumors of the supratentorial compartment. In: *Cancer in the Nervous System*, edited by Levin, V.A. New York, NY: Churchill Livingstone, pp. 57-126.

Biard, D.S., Martin, M., Rhun, Y.L., Duthu, A., Lefaix, J.L., May, E., and May, P. (1994) Concomitant p53 gene mutation and increased radiosensitivity in rat lung embryo epithelial cells during neoplastic development. *Cancer Res.* 54: 3361-3364.

Bienz, B., Zakut Houri, R., Givol, D., and Oren, M. (1984) Analysis of the gene coding for the murine cellular tumour antigen p53. *EMBO J.* **3**: 2179-2183.

Bignami, A. and Dahl, D. (1994) Glial Cells in the Central Nervous System and their Reaction to Injury, Austin, TX: R.G. Landes Company.

Bischoff, J.R., Friedman, P.N., Marshak, D.R., Prives, C., and Beach, D. (1990) Human p53 is phosphorylated by p60-cdc2 and cyclin B-cdc2. *Proc. Natl. Acad. Sci. U.* S. A. 87: 4766-4770.

Bishop, M. and de la Monte, S.M. (1989) Dual lineage of astrocytomas. Am. J. Pathol. 135: 517-527.

Bobola, M.S., Berger, M.S., and Silber, J.R. (1995) Contribution of O6-methylguanine-DNA methyltransferase to resistance to

1,3-(2-chloroethyl)-1-nitrosourea in human brain tumor-derived cell lines. *Mol. Carcinog.* 13: 81-88.

Bodell, W.J., Rupniak, H.T., Rasmussen, J., Morgan, W.F., and Rosenblum, M.L. (1984) Reduced level of DNA cross-links and sister chromatid exchanges in 1,3-bis(2-chloroethyl)-1-nitrosourea-resistant rat brain tumor cells. *Cancer Res.* 44: 3763-3767.

Bodell, W.J., Aida, T., Berger, M.S., and Rosenblum, M.L. (1986) Increased repair of O6-alkylguanine DNA adducts in glioma-derived human cells resistant to the cytotoxic and cytogenetic effects of 1,3-bis(2-chloroethyl)-1-nitrosourea. *Carcinogenesis* 7: 879-883.

Bogler, O., Wren, D., Barnett, S.C., Land, H., and Noble, M. (1990) Cooperation between two growth factors promotes extended self-renewal and inhibits differentiation of oligodendrocyte-type-2 astrocyte (O-2A) progenitor cells. *Proc. Natl. Acad. Sci. U. S. A.* 87: 6368-6372.

Boiardi, A., Silvani, A., Milanesi, I., Botturi, M., and Broggi, G. (1991) Primary glial tumor patients treated by combining cis-platin and etoposide. *J. Neurooncol.* 11: 165-170.

Boise, L.H., Gonzalez Garcia, M., Postema, C.E., Ding, L., Lindsten, T., Turka, L.A., Mao, X., Nunez, G., and Thompson, C.B. (1993) bcl-x, a bcl-2-related gene that functions as a dominant regulator of apoptotic cell death. *Cell* **74:** 597-608.

Bondy, M., Wrensch, M., and Wiencke, J. (1996) Genetic and hereditary syndromes associated with tumors of the central nervous system. In: *Cancer in the Nervous System*, edited by Levin, V.A. New York, NY: Churchill Livingstone, pp. 1-12.

Bortner, C.D., Oldenburg, N.B.E., and Cidlowski, J.A. (1995) The role of DNA fragmentation in apoptosis. *Trends Cell Biol.* 5: 21-26.

Bottenstein, J.E. and Sato, G.H. (1979) Growth of a rat neuroblastoma cell line in serum-free supplemented medium. Proc. Natl. Acad. Sci. U. S. A. 76: 514-517.

Brachman, D.G., Beckett, M., Graves, D., Haraf, D., Vokes, E., and Weichselbaum, R.R. (1993) p53 mutation does not correlate with radiosensitivity in 24 head and neck cancer cell lines [see comments]. *Cancer Res.* 53: 3667-3669.

Branch, P., Aquilina, G., Bignami, M., and Karran, P. (1993) Defective mismatch binding and a mutator phenotype in cells tolerant to DNA damage. *Nature* 362: 652-654.

Brem, S., Tsanaclis, A.M., Gately, S., Gross, J.L., and Herblin, W.F. (1992)

Immunolocalization of basic fibroblast growth factor to the microvasculature of human brain tumors. *Cancer* **70**: 2673-2680.

Bronstein, S.M., Skopek, T.R., and Swenberg, J.A. (1992) Efficient repair of O6-ethylguanine, but not O4-ethylthymine or O2-ethylthymine, is dependent upon O6-alkylguanine-DNA alkyltransferase and nucleotide excision repair activities in human cells [see comments]. *Cancer Res.* **52**: 2008-2011.

Brown, M., Cairncross, J.G., Vick, N.A., Macdonald, D.R., Friedman, H.S., Dropcho, E.J., and Schold, S.C., Jr. (1990) Differential response of recurrent oligodendrogliomas vs astrocytomas to intravenous (IV) melphalan. *Neurology* **40**: 397-398 (Abstract).

Burger, P.C., Shibata, T., Aguzzi, A., and Kleihues, P. (1988) Selective induction by N-nitrosoethylurea of oligodendrogliomas in fetal forebrain transplants. *Cancer Res.* 48: 2871-2875.

Caelles, C., Helmberg, A., and Karin, M. (1994) p53-dependent apoptosis in the absence of transcriptional activation of p53-target genes [see comments]. *Nature* **370**: 220-223.

Cairncross, G., Macdonald, D., Ludwin, S., Lee, D., Cascino, T., Buckner, J., Fulton, D., Dropcho, E., Stewart, D., Schold, C., Jr., Wainman, N., and Eisenhauer, E. (1994) Chemotherapy for anaplastic oligodendroglioma. J. Clin. Oncol. 12: 2013-2021.

Cairneross, J.G., Macdonald, D.R., and Ramsay, D.A. (1992) Aggressive oligodendroglioma: a chemosensitive tumor. *Neurosurgery* **31:** 78-82.

Cairncross, J.G. (1994) Aggressive oligodendroglioma: a chemosensitive tumor. Recent Results Cancer Res. 135: 127-133.

Cairneross, J.G. and Macdonald, D.R. (1988) Successful chemotherapy for recurrent malignant oligodendroglioma. Ann. Neurol. 23: 360-364.

Cairncross, J.G. and Macdonald, D.R. (1991) Chemotherapy for oligodendroglioma. Progress report. Arch. Neurol. 48: 225-227.

Cameron, R.S. and Rakic, P. (1991) Glial cell lineage in the cerebral cortex: a review and synthesis. *Glia* 4: 124-137.

Canman, C.E., Gilmer, T.M., Coutts, S.B., and Kastan, M.B. (1995) Growth factor modulation of p53-mediated growth arrest versus apoptosis. *Genes Dev.* 9: 600-611.

Cappelli, E., Redaelli, A., Rivano, M.E., Abbondandolo, A., and Frosina, G. (1995) Repair of 1-(2-chloroethyl)-3-cyclohexyl-1-nitrosourea-induced damage by mammalian cell extracts. *Carcinogenesis* 16: 2267-2270. Carder, P.J., Hume, R., Fryer, A.A., Strange, R.C., Lauder, J., and Bell, J.E. (1990) Glutathione S-transferase in human brain. *Neuropathol. Appl. Neurobiol.* 16: 293-303.

Chan, C.L., Wu, Z., Eastman, A., and Bresnick, E. (1992) Irradiation-induced expression of O6-methylguanine-DNA methyltransferase in mammalian cells. *Cancer Res.* 52: 1804-1809.

Chaney, S.G. and Sancar, A. (1996) DNA repair: enzymatic mechanisms and relevance to drug response. J. Natl. Cancer Inst. 88: 1346-1360.

Chang, C.H., Horton, J., Schoenfeld, D., Salazer, O., Perez Tamayo, R., Kramer, S., Weinstein, A., Nelson, J.S., and Tsukada, Y. (1983) Comparison of postoperative radiotherapy and combined postoperative radiotherapy and chemotherapy in the multidisciplinary management of malignant gliomas. A joint Radiation Therapy Oncology Group and Eastern Cooperative Oncology Group study. *Cancer* **52**: 997-1007.

Chang, F., Syrjanen, S., and Syrjanen, K. (1995) Implications of the p53 tumor-suppressor gene in clinical oncology [see comments]. J. Clin. Oncol. 13: 1009-1022.

Chen, I.-T., Smith, M.L., O'Connor, P.M., and Fornace, A.J., Jr (1995) Direct interaction of Gadd45 with PCNA and evidence for competitive interaction of Gadd45 and $p21^{Wafl/Cipl}$ with PCNA. Oncogene 11: 1931-1937.

Chen, P.L., Chen, Y.M., Bookstein, R., and Lee, W.H. (1990) Genetic mechanisms of tumor suppression by the human p53 gene. *Science* 250: 1576-1580.

Chen, X., Bargonetti, J., and Prives, C. (1995) p53, through p21 (WAF1/CIP1), induces cyclin D1 synthesis. *Cancer Res.* 55: 4257-4263.

Chin, K.V., Ueda, K., Pastan, I., and Gottesman, M.M. (1992) Modulation of activity of the promoter of the human MDR1 gene by Ras and p53. *Science* 255: 459-462.

Chung, R., Whaley, J., Kley, N., Anderson, K., Louis, D., Menon, A., Hettlich, C., Freiman, R., Hedley Whyte, E.T., Martuza, R., Jenkins, R., Yandell, D., and Seizinger, B.R. (1991) TP53 gene mutations and 17p deletions in human astrocytomas. *Genes Chromosomes Cancer* 3: 323-331.

Citron, M., Decker, R., Chen, S., Schneider, S., Graver, M., Kleynerman, L., Kahn, L.B., White, A., Schoenhaus, M., and Yarosh, D. (1991) O6-methylguanine-DNA methyltransferase in human normal and tumor tissue from brain, lung, and ovary. *Cancer Res.* 51: 4131-4134.

Citron, M., White, A., Decker, R., Wasserman, P., Li, B., Randall, T., Guerra, D.,

Belanich, M., and Yarosh, D. (1995) O6-methylguanine-DNA methyltransferase in human brain tumors detected by activity assay and monoclonal antibodies. *Oncol. Res.* 7: 49-55.

Clarke, A.R., Purdie, C.A., Harrison, D.J., Morris, R.G., Bird, C.C., Hooper, M.L., and Wyllie, A.H. (1993) Thymocyte apoptosis induced by p53-dependent and independent pathways [see comments]. *Nature* 362: 849-852.

Cole, S.P.C. (1986) Rapid chemosensitivity testing of human lung tumor cells using the MTT assay. *Cancer Chemother. Pharmacol.* 17: 259-263.

Cole, S.P.C., Bhardwaj, G., Gerlach, J.H., Mackie, J.E., Grant, C.E., Almquist, K.C., Stewart, A.J., Kurz, E.U., Duncan, A.M.V., and Deeley, R.G. (1992) Overexpression of a transporter gene in a multidrug-resistant human lung cancer cell line. *Science* 258: 1650-1654.

Collins, M.K.L., Marvel, J., Malde, P., and Lopez-Rivas, A. (1992) Interleukin 3 protects murine bone marrow cells from apoptosis induced by DNA damaging agents. *J. Exp. Med.* **176**: 1043-1051.

Collins, V.P. (1995) Gene amplification in human gliomas. Glia 15: 289-296.

Colvin, M., Brundrett, R.B., Cowens, W., Jardine, I., and Ludlum, D.B. (1976) A chemical basis for the antitumor activity of chloroethylnitrosoureas. *Biochem. Pharmacol.* **25:** 695-699.

Craig, R.W. (1995) The bcl-2 gene family. Semin. Cancer Biol. 6: 35-43.

Crawford, L.V., Pim, D.C., and Bulbrook, R.D. (1982) Detection of antibodies against the cellular protein p53 in sera from patients with breast cancer. *Int. J. Cancer* 30: 403-408.

Cross, S.M., Sanchez, C.A., Morgan, C.A., Schimke, M.K., Ramel, S., Idzerda, R.L., Raskind, W.H., and Reid, B.J. (1995) A p53-dependent mouse spindle checkpoint. *Science* 267: 1353-1356.

Cruz Sanchez, F.F., Rossi, M.L., Buller, J.R., Carboni, P., Jr., Fineron, P.W., and Coakham, H.B. (1991) Oligodendrogliomas: a clinical, histological, immunocytochemical and lectin-binding study. *Histopathology* **19:** 361-367.

Curran, W.J., Jr., Scott, C.B., Nelson, J.S., Martin, L.A., Weinstein, A.S., Phillips, T.L., Murray, K., Fischbach, A.J., Yakar, D., Schwade, J.G., Corn, B.W., and Nelson, D.F. (1992) Survival comparison of radiosurgery eligible and ineligible malignant glioma patients treated with hyperfractionated radiation therapy (RT) and

BCNU: a report of radiation therapy oncology group (RTOG) 83-02. Int. J. Radiat. Oncol. Biol. Phys. 24(Suppl.1): 129-130 (Abstract).

Cutarelli, P.E., Roessmann, U.R., Miller, R.H., Specht, C.S., and Grossniklaus, H.E. (1991) Immunohistochemical properties of human optic nerve glioma. Evidence of type 1 astrocyte origin. *Invest. Ophthalmol. Vis. Sci.* 32: 2521-2524.

Dabholkar, M.D., Berger, M.S., Vionnet, J.A., Egwuagu, C., Silber, J.R., Yu, J.J., and Reed, E. (1995) Malignant and nonmalignant brain tissues differ in their messenger RNA expression patterns for ERCC1 and ERCC2. *Cancer Res.* 55: 1261-1266.

Dabholkar, M.D., Berger, M.S., Vionnet, J.A., Overton, L., Thompson, C., Bostick Bruton, F., Yu, J.J., Silber, J.R., and Reed, E. (1996) Comparative analyses of relative ERCC3 and ERCC6 mRNA levels in gliomas and adjacent non-neoplastic brain. *Mol. Carcinog.* 17: 1-7.

de Kant, E., Heide, I., Thiede, C., Herrmann, R., and Rochlitz, C.F. (1996) *MDR1* expression correlates with mutant *p53* expression in colorectal cancer metastases. *J. Cancer Res. Clin. Oncol.* **122:** 671-675.

de la Monte, S.M. (1989) Uniform lineage of oligodendrogliomas. Am. J. Pathol. 135: 529-540.

DeAngelis, L.M., Burger, P.C., Green, S.B., and Cairncross, J.G. (1996) Adjuvant chemotherapy for malignant gliomas: Who benefits? *Ann. Neurol.* 40: 491-492 (Abstract).

del Arco, A., Garcia, J., Arribas, C., Barrio, R., Blazquez, M.G., Izquierdo, J.M., and Izquierdo, M. (1993) Timing of p53 mutations during astrocytoma tumorigenesis. *Hum. Mol. Genet.* 2: 1687-1690.

Del Bigio, M.R. (1995) The ependyma: a protective barrier between brain and cerebrospinal fluid. *Glia* 14: 1-13.

Deneke, S.M., Lawrence, R.A., and Jenkinson, S.G. (1992) Endothelial cell cystine uptake and glutathione increase with N, N-bis(2-chloroethyl)-N-nitrosourea exposure. *Am. J. Physiol.* **262**: L301-4.

Desmeules, M., Mikkelsen, T., and Mao, Y. (1992) Increasing incidence of primary malignant brain tumors: influence of diagnostic methods. J. Natl. Cancer Inst. 84: 442-445.

Deutsch, M., Green, S.B., Strike, T.A., Burger, P.C., Robertson, J.T., Selker, R.G., Shapiro, W.R., Mealey, J., Jr., Ransohoff, J., II, Paoletti, P., Smith, K.R., Jr., Odom,

G.L., Hunt, W.E., Young, B., Alexander, E., Jr., Walker, M.D., and Pistenmaa, D.A. (1989) Results of a randomized trial comparing BCNU plus radiotherapy, streptozotocin plus radiotherapy, BCNU plus hyperfractionated radiotherapy, and BCNU following misonidazole plus radiotherapy in the postoperative treatment of malignant glioma. *Int. J. Radiat. Oncol. Biol. Phys.* 16: 1389-1396.

Dexter, E.U., Yamashita, T.S., Donovan, C., and Gerson, S.L. (1989) Modulation of O6-alkylguanine-DNA alkyltransferase in rats following intravenous administration of O6-methylguanine. *Cancer Res.* **49:** 3520-3524.

Di Leonardo, A., Linke, S.P., Clarkin, K., and Wahl, G.M. (1994) DNA damage triggers a prolonged p53-dependent G1 arrest and long-term induction of Cip1 in normal human fibroblasts. *Genes Dev.* 8: 2540-2551.

Diller, L., Kassel, J., Nelson, C.E., Gryka, M.A., Litwak, G., Gebhardt, M., Bressac, B., Ozturk, M., Baker, S.J., Vogelstein, B., and Friend, S.H. (1990) p53 functions as a cell cycle control protein in osteosarcomas. *Mol. Cell. Biol.* **10**: 5772-5781.

Ding, R., Ghosh, K., Eastman, A., and Bresnick, E. (1985) DNA-mediated transfer and expression of a human DNA repair gene that demethylates O6-methylguanine. *Mol. Cell. Biol.* 5: 3293-3296.

Dirks, P.B., Hubbard, S.L., Murakami, M., and Rutka, J.T. (1997) Cyclin and cyclin-dependent kinase expression in human astrocytoma cell lines. *J. Neuropathol. Exp. Neurol.* 56: 291-300.

Dittmer, D., Pati, S., Zambetti, G., Chu, S., Teresky, A.K., Moore, M., Finlay, C., and Levine, A.J. (1993) Gain of function mutations in p53. *Nat. Genet.* 4: 42-46.

Dolan, M.E., Corsico, C.D., and Pegg, A.E. (1985) Exposure of Hela cells to O^6 -alkylguanines increases sensitivity to the cytotoxic effects of alkylating agents. Biochem. Biophys. Res. Commun. 132: 178-185.

Dolan, M.E., Young, G.S., and Pegg, A.E. (1986) Effect of O^6 -alkylguanine pretreatment on the sensitivity of human colon tumor cells to the cytotoxic effects of chloroethylating agents. *Cancer Res.* 46: 4500-4504.

Dolan, M.E., Moschel, R.C., and Pegg, A.E. (1990a) Depletion of mammalian O6-alkylguanine-DNA alkyltransferase activity by O6-benzylguanine provides a means to evaluate the role of this protein in protection against carcinogenic and therapeutic alkylating agents. *Proc. Natl. Acad. Sci. U. S. A.* 87: 5368-5372.

Dolan, M.E., Stine, L., Mitchell, R.B., Moschel, R.C., and Pegg, A.E. (1990b) Modulation of mammalian O6-alkylguanine-DNA alkyltransferase in vivo by O6-benzylguanine and its effect on the sensitivity of a human glioma tumor to 1-(2-chloroethyl)-3-(4-methylcyclohexyl)-1-nitrosourea. Cancer Commun. 2: 371-377.

Dolan, M.E., Mitchell, R.B., Mummert, C., Moschel, R.C., and Pegg, A.E. (1991) Effect of O6-benzylguanine analogues on sensitivity of human tumor cells to the cytotoxic effects of alkylating agents. *Cancer Res.* **51**: 3367-3372.

Doz, F., Roosen, N., and Rosenblum, M.L. (1993) Metallothionein and anticancer agents: the role of metallothionein in cancer chemotherapy. J. Neurooncol. 17: 123-129.

Dropcho, E.J., Rosenfeld, S.S., Morawetz, R.B., Vitek, J., Brothers, M., Gorum, T., Bell, S., Gillespie, G.Y., Glantz, M., Mahaley, M.S., Jr., and Schold, S.C., Jr. (1992) Preradiation intracarotid cisplatin treatment of newly diagnosed anaplastic gliomas. The CNS Cancer Consortium. J. Clin. Oncol. 10: 452-458.

Druckrey, H., Ivankovic, S., and Preussmann, R. (1966) Teratogenic and carcinogenic effects in the offspring after a single injection of ethylnitrosourea to pregnant rats. *Nature* **210**: 1378-1379.

Druckrey, H. (1973) Specific carcinogenic and teratogenic effects of 'indirect' alkylating methyl and ethyl compounds, and their dependency on stages of ontogenic developments. *Xenobiotica* **3:** 271-303.

Dumenco, L.L., Allay, E., Norton, K., and Gerson, S.L. (1993) The prevention of thymic lymphomas in transgenic mice by human O6-alkylguanine-DNA alkyltransferase. *Science* **259**: 219-222.

Dutly, F. and Schwab, M.E. (1991) Neurons and astrocytes influence the development of purified O-2A progenitor cells. *Glia* 4: 559-571.

Eccleston, P.A. and Silberberg, D.H. (1984) The differentiation of oligodendrocytes in a serum-free hormone-supplemented medium. *Brain Res.* **318**: 1-9.

Eddleston, M. and Mucke, L. (1993) Molecular profile of reactive astrocytes: implications for their role in neurologic disease. *Neuroscience* 54: 15-36.

Edwards, D.R., Parfett, C.L.J., and Denhardt, D.T. (1985) Transcriptional regulation of two serum-induced RNAs in mouse fibroblasts: equivalence of one species to B2 repetitive elements. *Mol. Cell. Biol.* 5: 3280-3288.

Edwards, D.R. and Denhardt, D.T. (1985) A study of mitochondrial and nuclear transcription with cloned cDNA probes. *Exp. Cell Res.* 157: 127-143.

Eisenbarth, G.S., Walsh, F.S., and Nirenberg, M. (1979) Monoclonal antibody to a

plasma membrane antigen of neurons. Proc. Natl. Acad. Sci. U. S. A. 76: 4913-4917.

Ekstrand, A.J., James, C.D., Cavenee, W.K., Seliger, B., Pettersson, R.F., and Collins, V.P. (1991) Genes for epidermal growth factor receptor, transforming growth factor alpha, and epidermal growth factor and their expression in human gliomas in vivo. *Cancer Res.* **51**: 2164-2172.

el Azouzi, M., Chung, R.Y., Farmer, G.E., Martuza, R.L., Black, P.M., Rouleau, G.A., Hettlich, C., Hedley Whyte, E.T., Zervas, N.T., Panagopoulos, K., Nakamura, Y., Gusella, J.F., and Seizinger, B.R. (1989) Loss of distinct regions on the short arm of chromosome 17 associated with tumorigenesis of human astrocytomas. *Proc. Natl. Acad. Sci. U. S. A.* 86: 7186-7190.

el Deiry, W.S., Tokino, T., Velculescu, V.E., Levy, D.B., Parsons, R., Trent, J.M., Lin, D., Mercer, W.E., Kinzler, K.W., and Vogelstein, B. (1993) WAF1, a potential mediator of p53 tumor suppression. *Cell* **76:** 817-825.

Eliyahu, D., Raz, A., Gruss, P., Givol, D., and Oren, M. (1984) Participation of p53 cellular tumour antigen in transformation of normal embryonic cells. *Nature* **312**: 646-649.

Eliyahu, D., Michalovitz, D., Eliyahu, S., Pinhasi-Kimhi, O., and Oren, M. (1989) Wild-type p53 can inhibit oncogene-mediated focus formation. *Proc. Natl. Acad. Sci. U. S. A.* 86: 8763-8767.

Endicott, J.A. and Ling, V. (1989) The biochemistry of P-glycoprotein-mediated multidrug resistance. Annu. Rev. Biochem. 58: 137-171.

Erickson, L.C., Bradley, M.O., and Kohn, K.W. (1978) Measurements of DNA damage in Chinese hamster cells treated with equitoxic and equimutagenic doses of nitrosoureas. *Cancer Res.* 38: 3379-3384.

Erickson, L.C., Laurent, G., Sharkey, N.A., and Kohn, K.W. (1980) DNA cross-linking and monoadduct repair in nitrosourea-treated human tumour cells. *Nature* 288: 727-729.

Espinosa de los Monteros, A., Zhang, M., and De Vellis, J. (1993) O2A progenitor cells transplanted into the neonatal rat brain develop into oligodendrocytes but not astrocytes. *Proc. Natl. Acad. Sci. U. S. A.* **90:** 50-54.

Evans, C.G., Bodell, W.J., Tokuda, K., Doane-Setzer, P., and Smith, M.T. (1987) Glutathione and related enzymes in rat brain tumor cell resistance to 1,3-bis(2-chloroethyl)-1-nitrosourea and nitrogen mustard. *Cancer Res.* 47: 2525-2530.

Fakharzadeh, S.S., Trusko, S.P., and George, D.L. (1991) Tumorigenic potential

associated with enhanced expression of a gene that is amplified in a mouse tumor cell line. *EMBO J.* **10:** 1565-1569.

Fan, S., Smith, M.L., Rivet, D.J., 2nd, Duba, D., Zhan, Q., Kohn, K.W., Fornace, A.J., Jr., and O'Connor, P.M. (1995) Disruption of p53 function sensitizes breast cancer MCF-7 cells to cisplatin and pentoxifylline. *Cancer Res.* 55: 1649-1654.

Farrow, S.N. and Brown, R. (1996) New members of the Bcl-2 family and their protein partners. Curr. Opin. Genet. Dev. 6: 45-49.

Felker, G.M., Friedman, H.S., Dolan, M.E., Moschel, R.C., and Schold, C. (1993) Treatment of subcutaneous and intracranial brain tumor xenografts with O6-benzylguanine and 1,3-bis(2-chloroethyl)-1-nitrosourea. *Cancer Chemother. Pharmacol.* **32:** 471-476.

Feun, L.G., Savaraj, N., and Landy, H.J. (1994) Drug resistance in brain tumors. J. Neurooncol. 20: 165-176.

ffrench Constant, C. and Raff, M.C. (1986) Proliferating bipotential glial progenitor cells in adult rat optic nerve. *Nature* **319**: 499-502.

Fine, H.A., Dear, K.B.G., Loeffler, J.S., Black, P.M., and Canellos, G.P. (1993) Meta-analysis of radiation therapy with and without adjuvant chemotherapy for malignant glioma in adults. *Cancer* 71: 2585-2597.

Finlay, C.A., Hinds, P.W., and Levine, A.J. (1989) The p53 proto-oncogene can act as a suppressor of transformation. *Cell* 57: 1083-1093.

Fiscella, M., Ullrich, S.J., Zambrano, N., Shields, M.T., Lin, D., Lees Miller, S.P., Anderson, C.W., Mercer, W.E., and Appella, E. (1993) Mutation of the serine 15 phosphorylation site of human p53 reduces the ability of p53 to inhibit cell cycle progression. *Oncogene* 8: 1519-1528.

Florell, R.C., Macdonald, D.R., Irish, W.D., Bernstein, M., Leibel, S.A., Gutin, P.H., and Cairneross, J.G. (1992) Selection bias, survival, and brachytherapy for glioma. J. *Neurosurg.* **76:** 179-183.

Fornace, A.J., Jr., Alamo, I., Jr., and Hollander, M.C. (1988) DNA damage inducible transcripts in mammalian cells. *Proc. Natl. Acad. Sci. U. S. A.* 85: 8800-8804.

Fornace, A.J., Jr., Nebert, D.W., Hollander, M.C., Luethy, J.D., Papathanasiou, M., Fargnoli, J., and Holbrook, N.J. (1989) Mammalian genes coordinately regulated by growth arrest signals and DNA-damaging agents. *Mol. Cell. Biol.* **9**: 4196-4203.

Frankel, R.H., Bayona, W., Koslow, M., and Newcomb, E.W. (1992) p53 mutations in human malignant gliomas: comparison of loss of heterozygosity with mutation frequency. *Cancer Res.* 52: 1427-1433.

Frankowski, H., Missotten, M., Fernandez, P.-A., Martinou, I., Michel, P., Sadoul, R., and Martinou, J.-C. (1995) Function and expression of the *Bcl-x* gene in the developing and adult nervous system. *Dev. Neurosci.* 6: 1917-1921.

Friedman, H.S., Colvin, O.M., Griffith, O.W., Lippitz, B., Elion, G.B., Schold, S.C., Jr., Hilton, J., and Bigner, D.D. (1989) Increased melphalan activity in intracranial human medulloblastoma and glioma xenografts following buthionine sulfoximine-mediated glutathione depletion. J. Natl. Cancer Inst. 81: 524-527.

Friedman, H.S., Colvin, O.M., Kaufmann, S.H., Ludeman, S.M., Bullock, N., Bigner, D.D., and Griffith, O.W. (1992a) Cyclophosphamide resistance in medulloblastoma. *Cancer Res.* 52: 5373-5378.

Friedman, H.S., Dolan, M.E., Moschel, R.C., Pegg, A.E., Felker, G.M., Rich, J., Bigner, D.D., and Schold, S.C. (1992b) Enhancement of nitrosourea activity in medulloblastoma and glioblastoma multiforme [published erratum appears in J Natl Cancer Inst 1994 Jul 6;86(13):1027]. J. Natl. Cancer Inst. 84: 1926-1931.

Fritz, G., Tano, K., Mitra, S., and Kaina, B. (1991) Inducibility of the DNA repair gene encoding O6-methylguanine-DNA methyltransferase in mammalian cells by DNA-damaging treatments. *Mol. Cell. Biol.* 11: 4660-4668.

Fritz, G., Dosch, J., Thielmann, H.W., and Kaina, B. (1993) Molecular and cellular characterization of Mex-/methylation-resistant phenotype. Gene and cDNA cloning, serum dependence, and tumor suppression of transfectant strains. J. Biol. Chem. 268: 21102-21112.

Frosina, G., Rossi, O., Arena, G., Gentile, S.L., Bruzzone, E., and Abbondandolo, A. (1990) O6-alkylguanine-DNA alkyltransferase activity in human brain tumors. *Cancer Lett.* 55: 153-158.

Fults, D., Tippets, R.H., Thomas, G.A., Nakamura, Y., and White, R. (1989) Loss of heterozygosity for loci on chromosome 17p in human malignant astrocytoma. *Cancer Res.* 49: 6572-6577.

Fults, D., Brockmeyer, D., Tullous, M.W., Pedone, C.A., and Cawthon, R.M. (1992) p53 mutation and loss of heterozygosity on chromosomes 17 and 10 during human astrocytoma progression. *Cancer Res.* 52: 674-679.

Gant, T.W., Silverman, J.A., and Thorgeirsson, S.S. (1992) Regulation of

P-glycoprotein gene expression in hepatocyte cultures and liver cell lines by a *trans*-acting transcriptional repressor. *Nucleic Acids Res.* **20**: 2841-2846.

Garfield, J., McDonald, H., Punt, J., Sokal, M., de Tribolet, N., Anderes, J.P., Morard, M., Bernasconi, S., Chatel, M., Darcel-Menault, F., Ben Hassel, M., Guegan, Y., Faivre, J., Afra, D., Kocsis, B., Hery, M., Hildebrand, J., Brotchi, J., Dewitte, O., Regnier, R., Aapro, M., Haefliger, J.M., Maat, B., Calliauw, L., Haferkamp, G., Dietz, H., Bernier, J., Goedhart, Z.D., Baris, G., Brucher, J.M., Cheuvart, B., Everard, D., and Mignolet, F. (1991) Cisplatin does not enhance the effect of radiation therapy in malignant gliomas. EORTC Brain Tumor Group. *Eur. J. Cancer* 27: 568-571.

Gerson, S.L., Trey, J.E., Miller, K., and Berger, N.A. (1986) Comparison of O6-alkylguanine-DNA alkyltransferase activity based on cellular DNA content in human, rat and mouse tissues. *Carcinogenesis* 7: 745-749.

Gillaspy, G.E., Mapstone, T.B., Samols, D., and Goldthwait, D.A. (1992) Transcriptional patterns of growth factors and proto-oncogenes in human glioblastomas and normal glial cells. *Cancer Lett.* 65: 55-60.

Girinsky, T., Koumenis, C., Graeber, T.G., Peehl, D.M., and Giaccia, A.J. (1995) Attenuated response of p53 and p21 in primary cultures of human prostatic epithelial cells exposed to DNA-damaging agents. *Cancer Res.* 55: 3726-3731.

Glass, J., Hochberg, F.H., Gruber, M.L., Louis, D.N., Smith, D., and Rattner, B. (1992) The treatment of oligodendrogliomas and mixed oligodendroglioma-astrocytomas with PCV chemotherapy. J. Neurosurg. 76: 741-745.

Goldman, J.E. and Vaysse, P.J. (1991) Tracing glial cell lineages in the mammalian forebrain. Glia 4: 149-156.

Goldsmith, M.E., Gudas, J.M., Schneider, E., and Cowan, K.H. (1995) Wild-type p53 stimulates expression from the human multidrug resistance promoter in a p53-negative cell line. J. Biol. Chem. 270: 1894-1898.

Gottesman, M.M. and Pastan, I. (1993) Biochemistry of multidrug resistance mediated by the multidrug transproter. Annu. Rev. Biochem. 62: 385-427.

Grafstrom, R.C., Pegg, A.E., Trump, B.F., and Harris, C.C. (1984) O6-alkylguanine-DNA alkyltransferase activity in normal human tissues and cells. *Cancer Res.* 44: 2855-2857.

Grant, C.E., Valdimarsson, G., Hipfner, D.R., Almquist, K.C., Cole, S.P., and Deeley, R.G. (1994) Overexpression of multidrug resistance-associated protein (MRP) increases resistance to natural product drugs. *Cancer Res.* 54: 357-361.

Grant, R., Liang, B.C., Page, M.A., Crane, D.L., Greenberg, H.S., and Junck, L. (1995) Age influences chemotherapy response in astrocytomas. *Neurology* **45**: 929-933.

Grant, R. and Ironside, J.W. (1995) Glutathione S-transferases and cytochrome P450 detoxifying enzyme distribution in human cerebral glioma. J. Neurooncol. 25: 1-7.

Green, J.A., Vistica, D.T., Young, R.C., Hamilton, T.C., Rogan, A.M., and Ozols, R.F. (1984) Potentiation of melphalan cytotoxicity in human ovarian cancer cell lines by glutathione depletion. *Cancer Res.* 44: 5427-5431.

Green, S.B., Byar, D.P., Walker, M.D., Pistenmaa, D.A., Alexander, E., Jr., Batzdorf, U., Brooks, W.H., Hunt, W.E., Mealey, J., Jr., Odom, G.L., Paoletti, P., Ransohoff, J., Robertson, J.T., Selker, R.G., Shapiro, W.R., Smith, K.R., Wilson, C.B., and Strike, T.A. (1983) Comparisons of carmustine, procarbazine, and high-dose methylprednisolone as additions to surgery and radiotherapy for the treatment of malignant glioma. *Cancer Treat. Rep.* 67: 121-132.

Greenblatt, M.S., Bennett, W.P., Hollstein, M., and Harris, C.C. (1994) Mutations in the p53 tumor suppressor gene: clues to cancer etiology and molecular pathogenesis. *Cancer Res.* 54: 4855-4878.

Greig, N.H., Ries, L.G., Yancik, R., and Rapoport, S.I. (1990) Increasing annual incidence of primary brain tumors in the elderly. J. Natl. Cancer Inst. 82: 1622-1624.

Grombacher, T., Mitra, S., and Kaina, B. (1996) Induction of the alkyltransferase (MGMT) gene by DNA damaging agents and the glucocorticoid dexamethasone and comparison with the response of base excision repair genes. *Carcinogenesis* 17: 2329-2336.

Grossi-Paoletti, E., Paoletti, P., Schiffer, D., and Fabiani, A. (1970) Experimental brain tumors induced in rats by nitrosourea derivatives. Part II. Morphological aspects of nitrosoethylurea tumors obtained by transplacental induction. J. Neurol. Sci. 11: 573-581.

Halazonetis, T.D., Davis, L.J., and Kandil, A.N. (1993) Wild-type p53 adopts a 'mutant'-like conformation when bound to DNA. *EMBO J.* **12**: 1021-1028.

Han, K.A. and Kulesz Martin, M.F. (1992) Alternatively spliced p53 RNA in transformed and normal cells of different tissue types. *Nucleic Acids Res.* 20: 1979-1981.

Hara, A., Sakai, N., Yamada, H., Niikawa, S., Yoshimi, N., and Mori, H. (1993) Localization of the placental form of glutathione S-transferase messenger ribonucleic acid in human glioma cell lines. *Neurosurgery* 33: 100-104.

Harper, J.W., Adami, G.R., Wei, N., Keyomarsi, K., and Elledge, S.J. (1993) The p21

Cdk-interacting protein Cip1 is a potent inhibitor of G1 cyclin-dependent kinases. *Cell* **75:** 805-816.

Harris, L.C., Remack, J.S., Houghton, P.J., and Brent, T.P. (1996) Wild-type p53 suppresses transcription of the human O6-methylguanine-DNA methyltransferase gene. *Cancer Res.* 56: 2029-2032.

Hartwell, L. (1992) Defects in a cell cycle checkpoint may be responsible for the genomic instability of cancer cells. *Cell* 71: 543-546.

Haupt, Y., Maya, R., Kazaz, A., and Oren, M. (1997) Mdm2 promotes the rapid degradation of p53. *Nature* 387: 296-299.

Hawn, M.T., Umar, A., Carethers, J.M., Marra, G., Kunkel, T.A., Boland, C.R., and Koi, M. (1995) Evidence for a connection between the mismatch repair system and the G2 cell cycle checkpoint. *Cancer Res.* 55: 3721-3725.

Hayashi, Y., Yamashita, J., and Yamaguchi, K. (1991) Timing and role of p53 gene mutation in the recurrence of glioma. *Biochem. Biophys. Res. Commun.* 180: 1145-1150.

Helseth, A. (1995) The incidence of primary central nervous system neoplasms before and after computerized tomography availability. J. Neurosurg. 83: 999-1003.

Henson, J.W., Schnitker, B.L., Correa, K.M., von Deimling, A., Fassbender, F., Xu, H.J., Benedict, W.F., Yandell, D.W., and Louis, D.N. (1994) The retinoblastoma gene is involved in malignant progression of astrocytomas. *Ann. Neurol.* **36**: 714-721.

Hermanson, M., Funa, K., Hartman, M., Claesson Welsh, L., Heldin, C.H., Westermark, B., and Nister, M. (1992) Platelet-derived growth factor and its receptors in human glioma tissue: expression of messenger RNA and protein suggests the presence of autocrine and paracrine loops. *Cancer Res.* **52**: 3213-3219.

Hermanson, M., Funa, K., Koopmann, J., Maintz, D., Waha, A., Westermark, B., Heldin, C.H., Wiestler, O.D., Louis, D.N., von Deimling, A., and Nister, M. (1996) Association of loss of heterozygosity on chromosome 17p with high platelet-derived growth factor alpha receptor expression in human malignant gliomas. *Cancer Res.* 56: 164-171.

Herzinger, T., Funk, J.O., Hillmer, K., Eick, D., Wolf, D.A., and Kind, P. (1995) Ultraviolet B irradiation-induced G2 cell cycle arrest in human keratinocytes by inhibitory phosphorylation of the cdc2 cell cycle kinase. *Oncogene* 11: 2151-2156.

Hess, R., Plaumann, B., Lutum, A.S., Haessler, C., Heinz, B., Fritsche, M., and Brandner, G. (1994) Nuclear accumulation of p53 in response to treatment with
DNA-damaging agents. Toxicol. Lett. 72: 43-52.

Hiesiger, E.M., Green, S.B., Shapiro, W.R., Burger, P.C., Selker, R.G., Mahaley, M.S., Jr., Ransohoff, J., II, VanGilder, J.C., Mealey, J., Jr., Robertson, J.T., Hochberg, F.H., and Young, R.F. (1995) Results of a randomized trial comparing intra-arterial cisplatin and intravenous PCNU for the treatment of primary brain tumors in adults: Brain Tumor Cooperative Group trial 8420A. J. Neurooncol. 25: 143-154.

Higgins, C.F. (1992) ABC transporters: from microorganisms to man. Annu. Rev. Cell Biol. 8: 67-113.

Hill, D.L., Kirk, M.C., and Struck, R.F. (1975) Microsomal metabolism of nitrosoureas. *Cancer Res.* 35: 296-301.

Hinds, P., Finlay, C., and Levine, A.J. (1989) Mutation is required to activate the p53 gene for cooperation with the ras oncogene and transformation. J. Virol. 63: 739-746.

Hollo, Z., Homolya, L., Hegedus, T., and Sarkadi, B. (1996) Transport properties of the multidrug resistance-associated protein (MRP) in human tumour cells. *FEBS Lett.* **383:** 99-104.

Hollstein, M., Sidransky, D., Vogelstein, B., and Harris, C.C. (1991) p53 mutations in human cancers. *Science* 253: 49-53.

Hollstein, M., Shomer, B., Greenblatt, M., Soussi, T., Hovig, E., Montesano, R., and Harris, C.C. (1996) Somatic point mutations in the p53 gene of human tumors and cell lines: updated compilation. *Nucleic Acids Res.* 24: 141-146.

Holowaty, E.J., Marrett, L.D., and Fehringer, G. (1996) Cancer Incidence in Ontario: Trends and Regional Variations in the 1980s, Toronto, Canada: Publications Ontario.

Hoshino, T., Deen, D.F., Williams, M.E., and Sano, Y. (1981) Differential response of elutriated 9L cells to treatment with 1,3-bis(2-chloroethyl)-1-nitrosourea. *Cancer Res.* **41:** 4404-4407.

Hotta, T., Saito, Y., Mikami, T., Kurisu, K., Kiya, K., Uozumi, T., Isowa, G., Ishizaki, K., and Ikenaga, M. (1993) Interrelationship between O6-alkylguanine-DNA alkyltransferase activity and susceptibility to chloroethylnitrosoureas in several glioma cell lines. J. Neurooncol. 17: 1-8.

Hotta, T., Saito, Y., Fujita, H., Mikami, T., Kurisu, K., Kiya, K., Uozumi, T., Isowa, G., Ishizaki, K., and Ikenaga, M. (1994) O6-alkylguanine-DNA alkyltransferase activity of human malignant glioma and its clinical implications. J. Neurooncol. 21: 135-140.

Hsiao, M., Low, J., Dorn, E., Ku, D., Pattengale, P., Yeargin, J., and Haas, M. (1994) Gain-of-function mutations of the p53 gene induce lymphohematopoietic metastatic potential and tissue invasiveness. *Am. J. Pathol.* 145: 702-714.

Hunter, S.B., Bandea, C., Swan, D., Abbott, K., and Varma, V.A. (1993) Mutations in the p53 gene in human astrocytomas: detection by single-strand conformation polymorphism analysis and direct DNA sequencing. *Mod. Pathol.* 6: 442-445.

Hupp, T.R., Meek, D.W., Midgley, C.A., and Lane, D.P. (1992) Regulation of the specific DNA binding function of p53. *Cell* 71: 875-886.

Huppi, K., Siwarski, D., Dosik, J., Michieli, P., Chedid, M., Reed, S., Mock, B., Givol, D., and Mushinski, J.F. (1994) Molecular cloning, sequencing, chromosomal localization and expression of mouse p21 (Waf1). *Oncogene* **9**: 3017-3020.

Ikizler, Y., van Meyel, D.J., Ramsay, D.A., Abdallah, G.L., Allaster, R.M., Macdonald, D.R., Cavenee, W.K., and Cairncross, J.G. (1992) Gliomas in families. *Can. J. Neurol. Sci.* 19: 492-497.

James, C.D., Carlbom, E., Dumanski, J.P., Hansen, M., Nordenskjold, M., Collins, V.P., and Cavenee, W.K. (1988) Clonal genomic alterations in glioma malignancy stages. *Cancer Res.* 48: 5546-5551.

James, C.D., Carlbom, E., Nordenskjold, M., Collins, V.P., and Cavenee, W.K. (1989) Mitotic recombination of chromosome 17 in astrocytomas. *Proc. Natl. Acad. Sci. U. S. A.* 86: 2858-2862.

James, C.D., He, J., Carlbom, E., Mikkelsen, T., Ridderheim, P.-A., Cavenee, W.K., and Collins, V.P. (1990) Loss of genetic information in central nervous system tumors common to children and young adults. *Genes Chromosomes Cancer* 2: 94-102.

James, C.D. and Collins, V.P. (1993) Glial tumors. In: *Molecular Genetics of Nervous System Tumors*, edited by Levine, A.J. and Schmidek, H.H. New York, NY: Wiley-Liss, pp. 241-248.

James, C.D. and Olson, J.J. (1996) Molecular genetics and molecular biology advances in brain tumors. *Curr. Opin. Oncol.* 8: 188-195.

Janzer, R.C. and Raff, M.C. (1987) Astrocytes induce blood-brain barrier properties in endothelial cells. *Nature* 325: 253-257.

Jayaraman, J. and Prives, C. (1995) Activation of p53 sequence-specific DNA binding by short single strands of DNA requires the p53 C-terminus. *Cell* 81: 1021-1029.

Jen, J., Harper, J.W., Bigner, S.H., Bigner, D.D., Papadopoulos, N., Markowitz, S., Willson, J.K.V., Kinzler, K.W., and Vogelstein, B. (1994) Deletion of p16 and p15 in brain tumors. *Cancer Res.* 54: 6353-6358.

Jenkins, J.R., Rudge, K., and Currie, G.A. (1984) Cellular immortalization by a cDNA clone encoding the transformation-associated phosphoprotein p53. *Nature* **312:** 651-654.

Johnson, P., Gray, D., Mowat, M., and Benchimol, S. (1991) Expression of wild-type p53 is not compatible with continued growth of p53-negative tumor cells. *Mol. Cell. Biol.* **11:** 1-11.

Kagi, J.H.R. and Schaffer, A. (1988) Biochemistry of metallothionein. *Biochemistry* 27: 8509-8515.

Kaina, B., Lohrer, H., Karin, M., and Herrlich, P. (1990) Overexpressed human metallothionein IIA gene protects Chinese hamster ovary cells from killing by alkylating agents. *Proc. Natl. Acad. Sci. U. S. A.* 87: 2710-2714.

Kaina, B., Fritz, G., Mitra, S., and Coquerelle, T. (1991) Transfection and expression of human O6-methylguanine-DNA methyltransferase (MGMT) cDNA in Chinese hamster cells: the role of MGMT in protection against the genotoxic effects of alkylating agents. *Carcinogenesis* 12: 1857-1867.

Karlbom, A.E., James, C.D., Boethius, J., Cavenee, W.K., Collins, V.P., Nordenskjold, M., and Larsson, C. (1993) Loss of heterozygosity in malignant gliomas involves at least three distinct regions on chromosome 10. *Hum. Genet.* **92**: 169-174.

Karran, P. and Bignami, M. (1992) Self-destruction and tolerance in resistance of mammalian cells to alkylation damage. *Nucleic Acids Res.* 20: 2933-2940.

Kastan, M.B., Onyekwere, O., Sidransky, D., Vogelstein, B., and Craig, R.W. (1991) Participation of p53 protein in the cellular response to DNA damage. *Cancer Res.* 51: 6304-6311.

Kastan, M.B., Canman, C.E., and Leonard, C.J. (1995) P53, cell cycle control and apoptosis: implications for cancer. *Cancer Metastasis Rev.* 14: 3-15.

Kat, A., Thilly, W.G., Fang, W.H., Longley, M.J., Li, G.M., and Modrich, P. (1993) An alkylation-tolerant, mutator human cell line is deficient in strand-specific mismatch repair. *Proc. Natl. Acad. Sci. U. S. A.* **90:** 6424-6428.

Kelley, S.L., Basu, A., Teicher, B.A., Hacker, M.P., Hamer, D.H., and Lazo, J.S. (1988) Overexpression of metallothionein confers resistance to anticancer drugs. *Science* **241**: 1813-1815.

Kennedy, P.G. and Fok Seang, J. (1986) Studies on the development, antigenic phenotype and function of human glial cells in tissue culture. *Brain* 109: 1261-1277.

Khanna, K.K. and Lavin, M.F. (1993) Ionizing radiation and UV induction of p53 protein by different pathways in ataxia-telangiectasia cells. *Oncogene* 8: 3307-3312.

Kim, L., Hochberg, F.H., Thornton, A.F., Harsh, G.R., Patel, H., Finkelstein, D., and Louis, D.N. (1996) Procarbazine, lomustine, and vincristine (PCV) chemotherapy for grade III and grade IV oligoastrocytomas. *J. Neurosurg.* **85:** 602-607.

Kiwit, J.C.W., Hertel, A., and Matuschek, A.E. (1994) Reversal of chemoresistance in malignant gliomas by calcium antagonists: correlation with the expression of multidrug-resistant p-glycoprotein. J. Neurosurg. 81: 587-594.

Kleihues, P., Burger, P.C., and Scheithauer, B.W. (1993) The new WHO classification of brain tumours. *Brain Pathol.* 3: 255-268.

Kleihues, P., Soylemezoglu, F., Schauble, B., Scheithauer, B.W., and Burger, P.C. (1995) Histopathology, classification, and grading of gliomas. *Glia* 15: 211-221.

Knapp, P.E. (1991) Studies of glial lineage and proliferation in vitro using an early marker for committed oligodendrocytes. J. Neurosci. Res. 30: 336-345.

Koestner, A., Swenberg, J.A., and Wechsler, W. (1971) Transplacental production with ethylnitrosourea of neoplasms of the nervous system in Sprague-Dawley rats. *Am. J. Pathol.* 63: 37-56.

Koga, H., Zhang, S., Kumanishi, T., Washiyama, K., Ichikawa, T., Tanaka, R., and Mukawa, J. (1994) Analysis of p53 gene mutations in low- and high-grade astrocytomas by polymerase chain reaction-assisted single-strand conformation polymorphism and immunohistochemistry. *Acta Neuropathol. Berl.* 87: 225-232.

Koike, K., Abe, T., Hisano, T., Kubo, T., Wada, M., Kohno, K., and Kuwano, M. (1996) Overexpression of multidrug resistance protein gene in human cancer cell lines selected for drug resistance to epipodophyllotoxins. *Jpn. J. Cancer Res.* 87: 765-772.

Kordek, R., Biernat, W., Alwasiak, J., Maculewicz, R., Yanagihara, R., and Liberski, P.P. (1995) p53 protein and epidermal growth factor receptor expression in human astrocytomas. J. Neurooncol. 26: 11-16.

Korsmeyer, S.J., Shutter, J.R., Veis, D.J., Merry, D.E., and Oltvai, Z.N. (1993) Bcl-2/Bax: a rheostat that regulates an anti-oxidant pathway and cell death. *Semin. Cancer Biol.* 4: 327-332.

Kraus, J.A., Bolln, C., Wolf, H.K., Neumann, J., Kindermann, D., Fimmers, R., Forster, F., Baumann, A., and Schlegel, U. (1994) TP53 alterations and clinical outcome in low grade astrocytomas. *Genes Chromosomes Cancer* 10: 143-149.

Kraus, J.A., Koopmann, J., Kaskel, P., Maintz, D., Brandner, S., Schramm, J., Louis, D.N., Wiestler, O.D., and von Deimling, A. (1995) Shared allelic losses on chromosomes 1p and 19q suggest a common origin of oligodendroglioma and oligoastrocytoma. J. Neuropathol. Exp. Neurol. 54: 91-95.

Kros, J.M., Van Eden, C.G., Stefanko, S.Z., Waayer Van Batenburg, M., and van der Kwast, T.H. (1990) Prognostic implications of glial fibrillary acidic protein containing cell types in oligodendrogliomas. *Cancer* 66: 1204-1212.

Kubbutat, M.H.G., Jones, S.N., and Vousden, K.H. (1997) Regulation of p53 stability by Mdm2. *Nature* 387: 299-303.

Kuerbitz, S.J., Plunkett, B.S., Walsh, W.V., and Kastan, M.B. (1992) Wild-type p53 is a cell cycle checkpoint determinant following irradiation. *Proc. Natl. Acad. Sci. U.* S. A. 89: 7491-7495.

Kulesz Martin, M.F., Lisafeld, B., Huang, H., Kisiel, N.D., and Lee, L. (1994) Endogenous p53 protein generated from wild-type alternatively spliced p53 RNA in mouse epidermal cells. *Mol. Cell. Biol.* 14: 1698-1708.

Kussie, P.H., Gorina, S., Marechal, V., Elenbaas, B., Moreau, J., Levine, A.J., and Pavletich, N.P. (1996) Structure of the MDM2 oncoprotein bound to the p53 tumor suppressor transactivation domain [comment]. *Science* 274: 948-953.

Lane, D.P. (1992) Cancer. p53, guardian of the genome [news; comment] [see comments]. *Nature* 358: 15-16.

Lane, D.P. and Crawford, L.V. (1979) T antigen is bound to a host protein in SV40-transformed cells. *Nature* 278: 261-263.

Lantos, P.L. (1972) The fine structure of periventricular pleomorphic gliomas induced transplacentally by N-ethyl-N-nitrosourea in BD-IX rats: with a note on their origin. J. Neurol. Sci. 17: 443-460.

Lantos, P.L. (1986) Development of nitrosourea-induced brain tumours--with a special note on changes occurring during latency. *Food Chem. Toxicol.* 24: 121-127.

Lantos, P.L. and Cox, D.J. (1976) The origin of experimental brain tumors: a sequential study. *Experientia* 32: 1467-1468.

Lee, J.M. and Bernstein, A. (1995) Apoptosis, cancer and the p53 tumour suppressor gene. *Cancer Metastasis Rev.* 14: 149-161.

Lee, S., Elenbaas, B., Levine, A., and Griffith, J. (1995) p53 and its 14 kDa C-terminal domain recognize primary DNA damage in the form of insertion/deletion mismatches. *Cell* 81: 1013-1020.

Lees-Miller, S.P., Chen, Y.-R., and Anderson, C.W. (1990) Human cells contain a DNA-activated protein kinase that phosphorylates simian virus 40 T antigen, mouse p53, and the human Ku autoantigen. *Mol. Cell. Biol.* 10: 6472-6481.

Leibel, S.A., Scott, C.B., and Pajak, T.F. (1991) The management of malignant gliomas with radiation therapy: therapeutic results and research stategies. *Semin. Radiat. Oncol.* 1: 32-49.

Leier, I., Jedlitschky, G., Buchholz, U., Cole, S.P., Deeley, R.G., and Keppler, D. (1994) The MRP gene encodes an ATP-dependent export pump for leukotriene C4 and structurally related conjugates. J. Biol. Chem. 269: 27807-27810.

Leighton, C., Fisher, B., Bauman, G., Depiero, S., Stitt, L., Macdonald, D., and Cairneross, G. (1997) Supratentorial low-grade glioma in adults: an analysis of prognostic factors and timing of radiation. J. Clin. Oncol. 15: 1294-1301.

Leonard, C.J., Canman, C.E., and Kastan, M.B. (1995) The role of p53 in cell-cycle control and apoptosis: implications for cancer. In: *Important Advances in Oncology 1995*, edited by DeVita, V.T., Jr., Hellman, S., and Rosenberg, S.A. Philadelphia, PA: J. B. Lippincott Company, pp. 33-42.

Leveillard, T., Andera, L., Bissonnette, N., Schaeffer, L., Bracco, L., Egly, J.M., and Wasylyk, B. (1996) Functional interactions between p53 and the TFIIH complex are affected by tumour-associated mutations. *EMBO J.* 15: 1615-1624.

Levin, V.A., Silver, P., Hannigan, J., Wara, W.M., Gutin, P.H., Davis, R.L., and Wilson, C.B. (1990) Superiority of post-radiotherapy adjuvant chemotherapy with CCNU, procarbazine, and vincristine (PCV) over BCNU for anaplastic gliomas: NCOG 6G61 final report. *Int. J. Radiat. Oncol. Biol. Phys.* 18: 321-324.

Levin, V.A., Leibel, S.A., and Gutin, P.H. (1997) Neoplasms of the central nervous system. Section 2: Neoplasms of the central nervous system. In: *Cancer: Principles & Practice of Oncology*, edited by DeVita, V.T., Jr., Hellman, S., and Rosenberg, S.A. Philadelphia, PA: Lippincott-Raven Publishers, pp. 2022-2082.

Levine, A.J. (1990) The p53 protein and its interactions with the oncogene products of the small DNA tumor viruses. *Virology* 177: 419-426.

Levine, A.J., Momand, J., and Finlay, C.A. (1991) The p53 tumour suppressor gene. *Nature* 351: 453-456.

Li, R., Waga, S., Hannon, G.J., Beach, D., and Stillman, B. (1994) Differential effects by the p21 CDK inhibitor on PCNA-dependent DNA replication and repair. *Nature* **371**: 534-537.

Liang, B.C., Ross, D.A., and Reed, E. (1995) Genomic copy number changes of DNA repair genes *ERCC1* and *ERCC2* in human gliomas. J. Neurooncol. 26: 17-23.

Lillien, L.E., Sendtner, M., Rohrer, H., Hughes, S.M., and Raff, M.C. (1988) Type-2 astrocyte development in rat brain cultures is initiated by a CNTF-like protein produced by type-1 astrocytes. *Neuron* 1: 485-494.

Lillien, L.E. and Raff, M.C. (1990) Analysis of the cell-cell interactions that control type-2 astrocyte development in vitro. *Neuron* 4: 525-534.

Lin, L.F., Doherty, D.H., Lile, J.D., Bektesh, S., and Collins, F. (1993) GDNF: a glial cell line-derived neurotrophic factor for midbrain dopaminergic neurons. *Science* 260: 1130-1132.

Lindahl, T., Sedgwick, B., Sekiguchi, M., and Nakabeppu, Y. (1988) Regulation and expression of the adaptive response to alkylating agents. *Annu. Rev. Biochem.* 57: 133-157.

Linskey, M.E. and Gilbert, M.R. (1995) Glial differentiation: a review with implications for new directions in neuro-oncology. *Neurosurgery* **36**: 1-21.

Linzer, D.I. and Levine, A.J. (1979) Characterization of a 54K dalton cellular SV40 tumor antigen present in SV40-transformed cells and uninfected embryonal carcinoma cells. *Cell* 17: 43-52.

Liu, L.F. (1989) DNA topoisomerase poisons as antitumor drugs. Annu. Rev. Biochem. 58: 351-375.

Livingstone, L.R., White, A., Sprouse, J., Livanos, E., Jacks, T., and Tlsty, T.D. (1992) Altered cell cycle arrest and gene amplification potential accompany loss of wild-type p53. *Cell* **70**: 923-935.

Loe, D.W., Almquist, K.C., Deeley, R.G., and Cole, S.P. (1996) Multidrug resistance protein (MRP)-mediated transport of leukotriene C4 and chemotherapeutic agents in membrane vesicles. Demonstration of glutathione-dependent vincristine transport. J. Biol. Chem. 271: 9675-9682.

Loeffler, J.S., Alexander, E., III, Wen, P.Y., Shea, W.M., Coleman, C.N., Kooy, H.M., Fine, H.A., Nedzi, L.A., Silver, B., Riese, N.E., and Black, P.M. (1990) Results of stereotactic brachytherapy used in the initial management of patients with glioblastoma. J. Natl. Cancer Inst. 82: 1918-1921.

Loeffler, J.S., Alexander, E., Shea, W.M., Wen, P.Y., Fine, H.A., Kooy, H.M., and Black, P.M. (1992) Radiosurgery as part of the initial management of patients with malignant gliomas. J. Clin. Oncol. 10: 1379-1385.

Lohrer, H., Robson, T., Grindley, H., Foster, S., and Hall, A. (1990) Differential effects on cell killing in metallothionein overexpressing CHO mutant cell lines. *Carcinogenesis* 11: 1937-1941.

Lopes, M.B.S., VandenBerg, S.R., and Scheithauer, B.W. (1993) The World Health Oganization classification of nervous system tumors in experimental neuro-oncology. In: *Molecular Genetics of Nervous System Tumors*, edited by Levine, A.J. and Schmidek, H.H. New York, NY: Wiley-Liss, pp. 1-36.

Lotem, J. and Sachs, L. (1993) Hematopoietic cells from mice deficient in wild-type p53 are more resistant to induction of apoptosis by some agents. *Blood* 82: 1092-1096.

Louis, D.N., von Deimling, A., Chung, R.Y., Rubio, M.P., Whaley, J.M., Eibl, R.H., Ohgaki, H., Wiestler, O.D., Thor, A.D., and Seizinger, B.R. (1993) Comparative study of p53 gene and protein alterations in human astrocytic tumors. *J. Neuropathol. Exp. Neurol.* 52: 31-38.

Louis, D.N. (1994) The p53 gene and protein in human brain tumors. J. Neuropathol. Exp. Neurol. 53: 11-21.

Louis, D.N. and Cavenee, W.K. (1997) Neoplasms of the central nervous system. Section 1: Molecular biology of central nervous system neoplasms. In: *Cancer: Principles & Practice of Oncology*, edited by DeVita, V.T., Jr., Hellman, S., and Rosenberg, S.A. Philadelphia, PA: Lippincott-Raven, pp. 2013-2022.

Louis, J.C., Magal, E., Muir, D., Manthorpe, M., and Varon, S. (1992) CG-4, a new bipotential glial cell line from rat brain, is capable of differentiating in vitro into either mature oligodendrocytes or type-2 astrocytes. J. Neurosci. Res. 31: 193-204.

Lowe, S.W., Ruley, H.E., Jacks, T., and Housman, D.E. (1993a) p53-dependent apoptosis modulates the cytotoxicity of anticancer agents. *Cell* 74: 957-967.

Lowe, S.W., Schmitt, E.M., Smith, S.W., Osborne, B.A., and Jacks, T. (1993b) p53 is required for radiation-induced apoptosis in mouse thymocytes [see comments]. *Nature* **362:** 847-849.

Lu, X. and Lane, D.P. (1993) Differential induction of transcriptionally active p53 following UV or ionizing radiation: defects in chromosome instability syndromes? *Cell* **75:** 765-778.

Ludlum, D.B. (1990) DNA alkylation by the haloethylnitrosoureas: nature of modifications produced and their enzymatic repair or removal. *Mutat. Res.* 233: 117-126.

Macdonald, D.R., Gaspar, L.E., and Cairncross, J.G. (1990) Successful chemotherapy for newly diagnosed aggressive oligodendroglioma. *Ann. Neurol.* 27: 573-574.

Madarasz, E., Theodosis, D.T., and Poulain, D.A. (1991) In vitro formation of type 2 astrocytes derived from postnatal rat hypothalamus or cerebral cortex. *Neuroscience* 43: 211-221.

Maki, C.G., Huibregtse, J.M., and Howley, P.M. (1996) In vivo ubiquitination and proteasome mediated degradation of p53. *Cancer Res.* 56: 2649-2654.

Maki, C.G. and Howley, P.M. (1997) Ubiquitination of p53 and p21 is differentially affected by ionizing and UV radiation. *Mol. Cell. Biol.* 17: 355-363.

Malcomson, R.D., Oren, M., Wyllie, A.H., and Harrison, D.J. (1995) p53-independent death and p53-induced protection against apoptosis in fibroblasts treated with chemotherapeutic drugs. *Br. J. Cancer* 72: 952-957.

Maltzman, W. and Czyzyk, L. (1984) UV irradiation stimulates levels of p53 cellular tumor antigen in nontransformed mouse cells. *Mol. Cell. Biol.* 4: 1689-1694.

Mannervik, B., Awasthi, Y.C., Board, P.G., Hayes, J.D., Di Ilio, C., Ketterer, B., Listowsky, I., Morgenstern, R., Muramatsu, M., Pearson, W.R., Pickett, C.B., Sato, K., Tu, C.-P.D., Widersten, M., and Wolf, C.R. (1992) Nomenclature for human glutathione transferases. *Biochem. J.* **282**: 305-310.

Mao, Y., Desmeules, M., Semenciw, R.M., Hill, G., Gaudette, L., and Wigle, D.T. (1991) Increasing brain cancer rates in Canada. *Can. Med. Assoc. J.* 145: 1583-1591.

Marathi, U.K., Dolan, M.E., and Erickson, L.C. (1994) Anti-neoplastic activity of sequenced administration of O6-benzylguanine, streptozotocin, and 1,3-bis(2-chloroethyl)-1-nitrosourea in vitro and in vivo. *Biochem. Pharmacol.* 48: 2127-2134.

Margison, G.P., Butler, J., and Hoey, B. (1985) O6-Methylguanine methyltransferase activity is increased in rat tissues by ionising radiation. *Carcinogenesis* 6: 1699-1702.

Marieb, E.N. and Mallatt, J. (1997) Fundamentals of the nervous system and nervous

tissue. In: Human Anatomy, edited by Schmid, J., Moller, L., Ekstrom, T., Burch, B., Wagner, A., and Lougee, L. New York, NY: Addison Wesley Longman, pp. 295-315.

Mashiyama, S., Murakami, Y., Yoshimoto, T., Sekiya, T., and Hayashi, K. (1991) Detection of p53 gene mutations in human brain tumors by single-strand conformation polymorphism analysis of polymerase chain reaction products. *Oncogene* **6:** 1313-1318.

Mason, W.P., Krol, G.S., and DeAngelis, L.M. (1996) Low-grade oligodendroglioma responds to chemotherapy. *Neurology* 46: 203-207.

Matlashewski, G., Lamb, P., Pim, D., Peacock, J., Crawford, L., and Benchimol, S. (1984) Isolation and characterization of a human p53 cDNA clone: expression of the human p53 gene. *EMBO J.* **3:** 3257-3262.

Mayr, G.A., Reed, M., Wang, P., Wang, Y., Schweds, J.F., and Tegtmeyer, P. (1995) Serine phosphorylation in the NH2 terminus of p53 facilitates transactivation. *Cancer Res.* 55: 2410-2417.

McBride, O.W., Merry, D., and Givol, D. (1986) The gene for human p53 cellular tumor antigen is located on chromosome 17 short arm (17p13). *Proc. Natl. Acad. Sci. U. S. A.* 83: 130-134.

McCarthy, K.D. and De Vellis, J. (1980) Preparation of separate astroglial and oligodendroglial cell cultures from rat cerebral tissue. J. Cell Biol. 85: 890-902.

McGown, A.T. and Fox, B.W. (1986) A proposed mechanism of resistance to cyclophosphamide and phosphoramide mustard in a Yoshida cell line in vitro. *Cancer Chemother. Pharmacol.* 17: 223-226.

McIlwrath, A.J., Vasey, P.A., Ross, G.M., and Brown, R. (1994) Cell cycle arrests and radiosensitivity of human tumor cell lines: dependence on wild-type p53 for radiosensitivity. *Cancer Res.* 54: 3718-3722.

McKinnon, R.D., Smith, C., Behar, T., Smith, T., and Dubois Dalcq, M. (1993) Distinct effects of bFGF and PDGF on oligodendrocyte progenitor cells. *Glia* 7: 245-254.

McLaughlin, J.K., Malker, H.S., Blot, W.J., Malker, B.K., Stone, B.J., Weiner, J.A., Ericsson, J.L., and Fraumeni, J.F., Jr. (1987) Occupational risks for intracranial gliomas in Sweden. J. Natl. Cancer Inst. 78: 253-257.

Meek, D.W., Simon, S., Kikkawa, U., and Eckhart, W. (1990) The p53 tumour suppressor protein is phosphorylated at serine 389 by casein kinase II. *EMBO J.* 9: 3253-3260.

Meek, D.W. (1994) Post-translational modification of p53. Semin. Cancer Biol. 5: 203-210.

Meltzer, P.S. (1994) MDM2 and p53: a question of balance [editorial; comment]. J. Natl. Cancer Inst. 86: 1265-1266.

Mercer, W.E., Shields, M.T., Amin, M., Sauve, G.J., Appella, E., Romano, J.W., and Ullrich, S.J. (1990) Negative growth control in a glioblastoma tumor cell line that conditionally expresses human wild-type p53. *Proc. Natl. Acad. Sci. U. S. A.* 87: 6166-6170.

Michieli, P., Chedid, M., Lin, D., Pierce, J.H., Mercer, W.E., and Givol, D. (1994) Induction of WAF1/CIP1 by a p53-independent pathway. *Cancer Res.* 54: 3391-3395.

Midgley, C.A., Owens, B., Briscoe, C.V., Thomas, D.B., Lane, D.P., and Hall, P.A. (1995) Coupling between gamma irradiation, p53 induction and the apoptotic response depends upon cell type in vivo. *J. Cell Sci.* 108: 1843-1848.

Miller, C., Mohandas, T., Wolf, D., Prokocimer, M., Rotter, V., and Koeffler, H.P. (1986) Human p53 gene localized to short arm of chromosome 17. *Nature* **319**: 783-784.

Miller, R.H., David, S., Patel, R., Abney, E.R., and Raff, M.C. (1985) A quantitative immunohistochemical study of macroglial cell development in the rat optic nerve: in vivo evidence for two distinct astrocyte lineages. *Dev. Biol.* **111:** 35-41.

Miller, R.H., ffrench Constant, C., and Raff, M.C. (1989) The macroglial cells of the rat optic nerve. Annu. Rev. Neurosci. 12: 517-534.

Miller, R.H. and Raff, M.C. (1984) Fibrous and protoplasmic astrocytes are biochemically and developmentally distinct. J. Neurosci. 4: 585-592.

Milne, D.M., Campbell, D.G., Caudwell, F.B., and Meek, D.W. (1994) Phosphorylation of the tumor suppressor protein p53 by mitogen-activated protein kinases. J. Biol. Chem. 269: 9253-9260.

Milne, D.M., Campbell, L.E., Campbell, D.G., and Meek, D.W. (1995) p53 is phosphorylated in vitro and in vivo by an ultraviolet radiation-induced protein kinase characteristic of the c-Jun kinase, JNK1. J. Biol. Chem. 270: 5511-5518.

Mineura, K., Izumi, I., Watanabe, K., and Kowada, M. (1993) Influence of O6-methylguanine-DNA methyltransferase activity on chloroethylnitrosourea chemotherapy in brain tumors. *Int. J. Cancer* 55: 76-81.

Mineura, K., Watanabe, K., Yanagisawa, T., and Kowada, M. (1996a) Quantification

of O6-methylguanine-DNA methyltransferase mRNA in human brain tumors. *Biochim. Biophys. Acta* **1289**: 105-109.

Mineura, K., Yanagisawa, T., Watanabe, K., Kowada, M., and Yasui, N. (1996b) Human brain tumor O(6)-methylguanine-DNA methyltransferase mRNA and its significance as an indicator of selective chloroethylnitrosourea chemotherapy. *Int. J. Cancer* 69: 420-425.

Mitchell, R.B., Moschel, R.C., and Dolan, M.E. (1992) Effect of O6-benzylguanine on the sensitivity of human tumor xenografts to 1,3-bis(2-chloroethyl)-1-nitrosourea and on DNA interstrand cross-link formation. *Cancer Res.* 52: 1171-1175.

Miyashita, T., Harigai, M., Hanada, M., and Reed, J.C. (1994) Identification of a p53-dependent negative response element in the bcl-2 gene. *Cancer Res.* 54: 3131-3135.

Miyashita, T. and Reed, J.C. (1995) Tumor suppressor p53 is a direct transcriptional activator of the human bax gene. *Cell* 80: 293-299.

Moll, U.M., Riou, G., and Levine, A.J. (1992) Two distinct mechanisms alter p53 in breast cancer: mutation and nuclear exclusion. *Proc. Natl. Acad. Sci. U. S. A.* 89: 7262-7266.

Momand, J., Zambetti, G.P., Olson, D.C., George, D., and Levine, A.J. (1992) The mdm-2 oncogene product forms a complex with the p53 protein and inhibits p53-mediated transactivation. *Cell* 69: 1237-1245.

Montgomery, J.A., James, R., McCaleb, G.S., and Johnston, T.P. (1967) The modes of decomposition of 1,3-bis(2-chloroethyl)-1-nitrosourea and related compounds. *J. Med. Chem.* 10: 668-674.

Moore, W.R., Anderson, M.E., Meister, A., Murata, K., and Kimura, A. (1989) Increased capacity for glutathione synthesis enhances resistance to radiation in *Escherichia coli*: a possible model for mammalian cell protection. *Proc. Natl. Acad. Sci.* U. S. A. 86: 1461-1464.

Morell, P. and Norton, W.T. (1980) Myelin. Sci. Am. 242: 88-118.

Morrow, C.S. and Cowan, K.H. (1990) Multidrug resistance associated with altered topoisomerase II activity-topoisomerases II as targets for rational drug design [comment]. J. Natl. Cancer Inst. 82: 638-639.

Mortimer, J.E., Crowley, J., Eyre, H., Weiden, P., Eltringham, J., and Stuckey, W.J. (1992) A phase II randomized study comparing sequential and combined intraarterial cisplatin and radiation therapy in primary brain tumors. A Southwest Oncology Group

study. Cancer 69: 1220-1223.

Mosner, J., Mummenbrauer, T., Bauer, C., Sczakiel, G., Grosse, F., and Deppert, W. (1995) Negative feedback regulation of wild-type p53 biosynthesis. *EMBO J.* 14: 4442-4449.

Moss, A.R. (1985) Occupational exposure and brain tumors. J. Toxicol. Environ. Health 16: 703-711.

Mousseau, M., Chauvin, C., Nissou, M.F., Chaffanet, M., Plantaz, D., Pasquier, B., Schaerer, R., and Benabid, A. (1993) A study of the expression of four chemoresistance-related genes in human primary and metastatic brain tumours. *Eur. J. Cancer* 29A: 753-759.

Mozell, R.L. and McMorris, F.A. (1991) Insulin-like growth factor I stimulates oligodendrocyte development and myelination in rat brain aggregate cultures. J. Neurosci. Res. 30: 382-390.

Muller, M., Meijer, C., Zaman, G.J., Borst, P., Scheper, R.J., Mulder, N.H., de Vries, E.G., and Jansen, P.L. (1994) Overexpression of the gene encoding the multidrug resistance-associated protein results in increased ATP-dependent glutathione S-conjugate transport. *Proc. Natl. Acad. Sci. U. S. A.* **91**: 13033-13037.

Musicco, M., Filippini, G., Bordo, B.M., Melotto, A., Morello, G., and Berrino, F. (1982) Gliomas and occupational exposure to carcinogens: case-control study. *Am. J. Epidemiol.* **116**: 782-790.

Myrnes, B., Norstrand, K., Giercksky, K.-E., Sjunneskog, C., and Krokan, H. (1984) A simplified assay for O⁶-methylguanine-DNA methyltransferase activity and its application to human neoplastic and non-neoplastic tissues. *Carcinogenesis* **5**: 1061-1064.

Nabeya, Y., Loganzo, F., Jr., Maslak, P., Lai, L., de Oliveira, A.R., Schwartz, G.K., Blundell, M.L., Altorki, N.K., Kelsen, D.P., and Albino, A.P. (1995) The mutational status of p53 protein in gastric and esophageal adenocarcinoma cell lines predicts sensitivity to chemotherapeutic agents. *Int. J. Cancer* 64: 37-46.

Nabors, M.W., Griffin, C.A., Zehnbauer, B.A., Hruban, R.H., Phillips, P.C., Grossman, S.A., Brem, H., and Colvin, O.M. (1991) Multidrug resistance gene (MDR1) expression in human brain tumors. J. Neurosurg. 75: 941-946.

Nakagawa, H., Fujita, T., Kubo, S., Tsuruzono, K., Yamada, M., Tokiyoshi, K., Miyawaki, Y., Kanayama, T., Kadota, T., and Hayakawa, T. (1994) Selective intra-arterial chemotherapy with a combination of etoposide and cisplatin for malignant gliomas: preliminary report. *Surg. Neurol.* 41: 19-27.

National Cancer Institute of Canada (1996) Canadian Cancer Statistics 1996, Toronto, Canada.

National Cancer Institute of Canada (1997) Canadian Cancer Statistics 1997, Toronto, Canada.

Negrini, M., Silini, E., Kozak, C., Tsujimoto, Y., and Croce, C.M. (1987) Molecular analysis of mbcl-2: structure and expression of the murine gene homologous to the human gene involved in follicular lymphoma. *Cell* **49**: 455-463.

Nelson, D.F., Diener West, M., Horton, J., Chang, C.H., Schoenfeld, D., and Nelson, J.S. (1988) Combined modality approach to treatment of malignant gliomas--re-evaluation of RTOG 7401/ECOG 1374 with long-term follow-up: a joint study of the Radiation Therapy Oncology Group and the Eastern Cooperative Oncology Group. *NCI. Monogr.* 279-284.

Nelson, W.G. and Kastan, M.B. (1994) DNA strand breaks: the DNA template alterations that trigger p53-dependent DNA damage response pathways. *Mol. Cell. Biol.* 14: 1815-1823.

New, P. (1995) A phase I/II dose escalation protocol of high dose DITC in the treatment of malignant primary brain tumor. *Proc. Annu. Meet. Am. Soc. Clin. Oncol.* 14: A304 (Abstract).

Newcomb, E.W., Madonia, W.J., Pisharody, S., Lang, F.F., Koslow, M., and Miller, D.C. (1993) A correlative study of p53 protein alteration and p53 gene mutation in glioblastoma multiforme. *Brain Pathol.* **3**: 229-235.

Newlands, E.S., Blackledge, G.R., Slack, J.A., Rustin, G.J., Smith, D.B., Stuart, N.S., Quarterman, C.P., Hoffman, R., Stevens, M.F., Brampton, M.H., and Gibson, A.C. (1992) Phase I trial of temozolomide (CCRG 81045: M&B 39831: NSC 362856). *Br. J. Cancer* 65: 287-291.

Newlands, E.S., O'Reilly, S.M., Glaser, M.G., Bower, M., Evans, H., Brock, C., Brampton, M.H., Colquhoun, I., Lewis, P., Rice Edwards, J.M., Illingworth, R.D., and Richards, P.G. (1996) The Charing Cross Hospital experience with temozolomide in patients with gliomas. *Eur. J. Cancer* 32A: 2236-2241.

Newman, E.A. (1985) Regulation of potassium levels by glial cells in the retina. *Trends Neurosci.* 8: 156-159.

Ng, W.F., Sarangi, F., Zastawny, R.L., Veinot Drebot, L., and Ling, V. (1989) Identification of members of the P-glycoprotein multigene family. *Mol. Cell. Biol.* 9: 1224-1232.

Nguyen, K.T., Liu, B., Ueda, K., Gottesman, M.M., Pastan, I., and Chin, K.-V. (1994) Transactivation of the human multidrug resistance (MDR1) gene promoter by p53 mutants. *Oncol. Res.* 6: 71-77.

Nigro, J.M., Baker, S.J., Preisinger, A.C., Jessup, J.M., Hostetter, R., Cleary, K., Bigner, S.H., Davidson, N., Baylin, S., Devilee, P., Glover, T., Collins, F.S., Weston, A., Modali, R., Harris, C.C., and Vogelstein, B. (1989) Mutations in the p53 gene occur in diverse human tumour types. *Nature* **342**: 705-708.

Nishikawa, R., Ji, X.D., Harmon, R.C., Lazar, C.S., Gill, G.N., Cavenee, W.K., and Huang, H.J. (1994) A mutant epidermal growth factor receptor common in human glioma confers enhanced tumorigenicity. *Proc. Natl. Acad. Sci. U. S. A.* 91: 7727-7731.

Nishikawa, R., Furnari, F.B., Lin, H., Arap, W., Berger, M.S., Cavenee, W.K., and Su Huang, H.J. (1995) Loss of P16INK4 expression is frequent in high grade gliomas. *Cancer Res.* 55: 1941-1945.

Noble, M. (1991) Points of controversy in the O-2A lineage: clocks and type-2 astrocytes. *Glia* 4: 157-164.

Noble, M., Ataliotis, P., Barnett, S.C., Bevan, K., Bogler, O., Groves, A., Jat, P., Wolswijk, G., and Wren, D. (1991) Development, regeneration, and neoplasia of glial cells in the central nervous system. *Ann. N. Y. Acad. Sci.* **633**: 35-47.

Noda, A., Ning, Y., Venable, S.F., Perreira-Smith, O.M., and Smith, J.R. (1994) Cloning of senescent cell-derived inhibitors of DNA synthesis using an expression screen. *Exp. Cell Res.* 211: 90-98.

Nomura, K., Hoshino, T., Knebel, K., Deen, D.F., and Barker, M. (1978) BCNU-induced perturbations in the cell cycle of 9L rat brain tumor cells. *Cancer Treat. Rep.* 62: 747-754.

Norimura, T., Nomoto, S., Katsuki, M., Gondo, Y., and Kondo, S. (1996) p53-dependent apoptosis suppresses radiation-induced teratogenesis [see comments]. *Nat. Med.* 2: 577-580.

Nowell, P.C. (1976) The clonal evolution of tumor cell populations. Science 194: 23-28.

Nutt, C.L., Costello, J.F., Bambrick, L.L., Yarosh, D.B., Swinnen, L.J., Chambers, A.F., and Cairneross, J.G. (1995) O6-methylguanine-DNA methyltransferase in tumors and cells of the oligodendrocyte lineage. *Can. J. Neurol. Sci.* 22: 111-115.

Nutt, C.L., Chambers, A.F., and Cairneross, J.G. (1996) Wild-type p53 renders mouse astrocytes resistant to 1,3-Bis(2-chloroethyl)-1-nitrosourea despite the absent of a

p53-dependent cell cycle arrest. Cancer Res. 56: 2748-2751.

O'Connor, P.M., Jackman, J., Jondle, D., Bhatia, K., Magrath, I., and Kohn, K.W. (1993) Role of the p53 tumor suppressor gene in cell cycle arrest and radiosensitivity of Burkitt's lymphoma cell lines. *Cancer Res.* 53: 4776-4780.

O'Reilly, S.M., Newlands, E.S., Glaser, M.G., Brampton, M., Rice-Edwards, J.M., Illingworth, R.D., Richards, P.G., Kennard, C., Colquhoun, I.R., Lewis, P., and Stevens, M.F.G. (1993) Temozolomide: a new oral cytotoxic chemotherapeutic agent with promising activity against primary brain tumours [published erratum appears in Eur J Cancer 1993;29A(10):1500]. *Eur. J. Cancer* 29A: 940-942.

Ohgaki, H., Eibl, R.H., Wiestler, O.D., Yasargil, M.G., Newcomb, E.W., and Kleihues, P. (1991) p53 mutations in nonastrocytic human brain tumors. *Cancer Res.* 51: 6202-6205.

Oltvai, Z.N., Milliman, C.L., and Korsmeyer, S.J. (1993) Bcl-2 heterodimerizes in vivo with a conserved homolog, Bax, that accelerates programmed cell death. *Cell* 74: 609-619.

Orian, J.M., Vasilopoulos, K., Yoshida, S., Kaye, A.H., Chow, C.W., and Gonzales, M.F. (1992) Overexpression of multiple oncogenes related to histological grade of astrocytic glioma. *Br. J. Cancer* 66: 106-112.

Ostrowski, L.E., von Wronski, M.A., Bigner, S.H., Rasheed, A., Schold, S.C., Jr., Brent, T.P., Mitra, S., and Bigner, D.D. (1991) Expression of O6-methylguanine-DNA methyltransferase in malignant human glioma cell lines. *Carcinogenesis* 12: 1739-1744.

Paleologos, N.A., Zemrus, T., Allen, C.V., Ragin, A., Walsh, A., Lapin, G.D., Groothuis, D.R., and Vick, N.A. (1992) Oligodendrogliomas: CT measurement of transcapillary permeability and response to vincristine chemotherapy. *Ann. Neurol.* 32: 289 (Abstract).

Papathanasiou, M.A., Kerr, N.C.K., Robbins, J.H., McBride, O.W., Alamo, I., Jr., Barrett, S.F., Hickson, I.D., and Fornace, A.J., Jr. (1991) Induction by ionizing radiation of the *gadd45* gene in cultured human cells: lack of mediation by protein kinase C. *Mol. Cell. Biol.* 11: 1009-1016.

Parada, L.F., Land, H., Weinberg, R.A., Wolf, D., and Rotter, V. (1984) Cooperation between gene encoding p53 tumour antigen and ras in cellular transformation. *Nature* **312:** 649-651.

Pavelic, J., Hlavka, V., Poljak, M., Gale, N., and Pavelic, K. (1994) p53 immunoreactivity in oligodendrogliomas. J. Neurooncol. 22: 1-6.

Pavletich, N.P., Chambers, K.A., and Pabo, C.O. (1993) The DNA-binding domain of p53 contains the four conserved regions and the major mutation hot spots. *Genes Dev.* 7: 2556-2564.

Pegg, A.E. (1990a) Properties of mammalian O6-alkylguanine-DNA transferases. *Mutat. Res.* 233: 165-175.

Pegg, A.E. (1990b) Mammalian O6-alkylguanine-DNA alkyltransferase: regulation and importance in response to alkylating carcinogenic and therapeutic agents. *Cancer Res.* **50:** 6119-6129.

Pellegata, N.S., Antoniono, R.J., Redpath, J.L., and Stanbridge, E.J. (1996) DNA damage and p53-mediated cell cycle arrest: a reevaluation. *Proc. Natl. Acad. Sci. U. S.* A. 93: 15209-15214.

Peterson, K., Paleologos, N., Forsyth, P., Macdonald, D.R., and Cairncross, J.G. (1996) Salvage chemotherapy for oligodendroglioma. J. Neurosurg. 85: 597-601.

Petty, R.D., Cree, I.A., Sutherland, L.A., Hunter, E.M., Lane, D.P., Preece, P.E., and Andreotti, P.E. (1994) Expression of the p53 tumour suppressor gene product is a determinant of chemosensitivity. *Biochem. Biophys. Res. Commun.* 199: 264-270.

Phillips, P.C. (1991) Antineoplastic drug resistance in brain tumors. *Neurol. Clin.* 9: 383-404.

Piepmeier, J.M., Fried, I., and Makuch, R. (1993) Low-grade astrocytomas may arise from different astrocyte lineages. *Neurosurgery* 33: 627-632.

Pilkington, G.J. (1994) Tumour cell migration in the central nervous system. Brain Pathol. 4: 157-166.

Plate, K.H., Breier, G., Weich, H.A., and Risau, W. (1992) Vascular endothelial growth factor is a potential tumour angiogenesis factor in human gliomas in vivo. *Nature* **359:** 845-848.

Plate, K.H., Breier, G., and Risau, W. (1994) Molecular mechanisms of developmental and tumor angiogenesis. *Brain Pathol.* 4: 207-218.

Plate, K.H. and Risau, W. (1995) Angiogenesis in malignant gliomas. Glia 15: 339-347.

Preuss, I., Thust, R., and Kaina, B. (1996) Protective effect of O6-methylguanine-DNA methyltransferase (MGMT) on the cytotoxic and recombinogenic activity of different antineoplastic drugs. *Int. J. Cancer* **65**: 506-512.

Raff, M.C., Abney, E.R., Cohen, J., Lindsay, R., and Noble, M. (1983a) Two types of astrocytes in cultures of developing rat white matter: differences in morphology, surface gangliosides, and growth characteristics. *J. Neurosci.* **3**: 1289-1300.

Raff, M.C., Miller, R.H., and Noble, M. (1983b) A glial progenitor cell that develops in vitro into an astrocyte or an oligodendrocyte depending on culture medium. *Nature* **303:** 390-396.

Raff, M.C., Abney, E.R., and Miller, R.H. (1984a) Two glial cell lineages diverge prenatally in rat optic nerve. *Dev. Biol.* 106: 53-60.

Raff, M.C., Williams, B.P., and Miller, R.H. (1984b) The in vitro differentiation of a bipotential glial progenitor cell. *EMBO J.* 3: 1857-1864.

Raff, M.C., Abney, E.R., and Fok Seang, J. (1985) Reconstitution of a developmental clock in vitro: a critical role for astrocytes in the timing of oligodendrocyte differentiation. *Cell* **42**: 61-69.

Raff, M.C. (1989) Glial cell diversification in the rat optic nerve. Science 243: 1450-1455.

Rafferty, J.A., Clarke, A.R., Sellappan, D., Koref, M.S., Frayling, I.M., and Margison, G.P. (1996) Induction of murine O6-alkylguanine-DNA-alkyltransferase in response to ionising radiation is p53 gene dose dependent. *Oncogene* 12: 693-697.

Ranscht, B., Clapshaw, P.A., Price, J., Noble, M., and Seifert, W. (1982) Development of oligodendrocytes and Schwann cells studied with a monoclonal antibody against galactocerebroside. *Proc. Natl. Acad. Sci. U. S. A.* **79**: 2709-2713.

Ransom, B.R. and Sontheimer, H. (1992) The neurophysiology of glial cells. J. Clin. Neurophysiol. 9: 224-251.

Ransom, D.T., Ritland, S.R., Kimmel, D.W., Moertel, C.A., Dahl, R.J., Scheithauer, B.W., Kelly, P.J., and Jenkins, R.B. (1992) Cytogenetic and loss of heterozygosity studies in ependymomas, pilocytic astrocytomas, and oligodendrogliomas. *Genes Chromosomes Cancer* 5: 348-356.

Rasheed, B.K., Fuller, G.N., Friedman, A.H., Bigner, D.D., and Bigner, S.H. (1992) Loss of heterozygosity for 10q loci in human gliomas. *Genes Chromosomes Cancer* 5: 75-82.

Rasheed, B.K., McLendon, R.E., Herndon, J.E., Friedman, H.S., Friedman, A.H., Bigner, D.D., and Bigner, S.H. (1994) Alterations of the TP53 gene in human gliomas. *Cancer Res.* 54: 1324-1330.

Reed, M., Woelker, B., Wang, P., Wang, Y., Anderson, M.E., and Tegtmeyer, P. (1995) The C-terminal domain of p53 recognizes DNA damaged by ionizing radiation. *Proc. Natl. Acad. Sci. U. S. A.* **92:** 9455-9459.

Reifenberger, J., Reifenberger, G., Liu, L., James, C.D., Wechsler, W., and Collins, V.P. (1994) Molecular genetic analysis of oligodendroglial tumors shows preferential allelic deletions on 19q and 1p. Am. J. Pathol. 145: 1175-1190.

Reisman, D., Greenberg, M., and Rotter, V. (1988) Human p53 oncogene contains one promoter upstream of exon 1 and a second, stronger promoter within intron 1. *Proc. Natl. Acad. Sci. U. S. A.* **85:** 5146-5150.

Rogel, A., Popliker, M., Webb, C.G., and Oren, M. (1985) p53 cellular tumor antigen: analysis of mRNA levels in normal adult tissues, embryos, and tumors. *Mol. Cell. Biol.* **5**: 2851-2855.

Russell, S.J., Ye, Y.W., Waber, P.G., Shuford, M., Schold, S.C., Jr., and Nisen, P.D. (1995) p53 mutations, O6-alkylguanine DNA alkyltransferase activity, and sensitivity to procarbazine in human brain tumors. *Cancer* **75**: 1339-1342.

Russo, A., Carmichael, J., Friedman, N., DeGraff, W., Tochner, Z., Glatstein, E., and Mitchell, J.B. (1986a) The roles of intracellular glutathione in antineoplastic chemotherapy. *Int. J. Radiat. Oncol. Biol. Phys.* **12**: 1347-1354.

Russo, A., DeGraff, W., Friedman, N., and Mitchell, J.B. (1986b) Selective modulation of glutathione levels in human normal versus tumor cells and subsequent differential response to chemotherapy drugs. *Cancer Res.* **46**: 2845-2848.

Saarinen, U.M., Pihko, H., and Makipernaa, A. (1990) High-dose thiotepa with autologous bone marrow rescue in recurrent malignant oligodendroglioma: a case report. J. Neurooncol. 9: 57-61.

Sakhi, S., Bruce, A., Sun, N., Tocco, G., Baudry, M., and Schreiber, S.S. (1994) p53 induction is associated with neuronal damage in the central nervous system. *Proc. Natl.* Acad. Sci. U. S. A. 91: 7525-7529.

Sambrook, J., Fritsch, E.F., and Maniatis, T. (1989) *Molecular Cloning. A Laboratory Manual*, Cold Spring Harbor, NY: Cold Spring Harbor Laboratory Press.

Sancar, A. (1994) Mechanisms of DNA excision repair. Science 266: 1954-1956.

Sancar, A. (1995) DNA repair in humans. Annu. Rev. Genetics 29: 69-105.

Sarkar, A., Dolan, M.E., Gonzalez, G.G., Marton, L.J., Pegg, A.E., and Deen, D.F.

(1993) The effects of O6-benzylguanine and hypoxia on the cytotoxicity of 1,3-bis(2-chloroethyl)-1-nitrosourea in nitrosourea-resistant SF-763 cells. *Cancer Chemother. Pharmacol.* 32: 477-481.

Sarkar, C., Roy, S., and Tandon, P.N. (1988) Oligodendroglial tumors. An immunohistochemical and electron microscopic study. *Cancer* **61**: 1862-1866.

Savitz, D.A. and Loomis, D.P. (1995) Magnetic field exposure in relation to leukemia and brain cancer mortality among electric utility workers. *Am. J. Epidemiol.* 141: 123-134.

Saxena, A., Clark, W.C., Robertson, J.T., Ikejiri, B., Oldfield, E.H., and Ali, I.U. (1992) Evidence for the involvement of a potential second tumor suppressor gene on chromosome 17 distinct from p53 in malignant astrocytomas. *Cancer Res.* 52: 6716-6721.

Scheffner, M., Werness, B.A., Huibregtse, J.M., Levine, A.J., and Howley, P.M. (1990) The E6 oncoprotein encoded by human papillomavirus types 16 and 18 promotes the degradation of p53. *Cell* 63: 1129-1136.

Scheffner, M., Huibregtse, J.M., Vierstra, R., and Howley, P.M. (1993) The HPV-16 E6 and E6-AP complex functions as a ubiquitin-protein ligase in the ubiquitination of p53. *Cell* **75**: 495-505.

Scheffner, M., Nuber, U., and Huibregtse, J.M. (1995) Protein ubiquitination involving an E1-E2-E3 ubiquitin thioester cascade. *Nature* 373: 81-83.

Scherer, S.J., Welter, C., Zang, K.D., and Dooley, S. (1996) Specific in vitro binding of p53 to the promoter region of the human mismatch repair gene hMSH2. *Biochem. Biophys. Res. Commun.* 221: 722-728.

Schiffer, D., Giordana, M.T., Pezzotta, S., Lechner, C., and Paoletti, P. (1978) Cerebral tumors induced by transplacental ENU: study of the different tumoral stages, particularly of early proliferation. *Acta Neuropathol. Berl.* **41**: 27-31.

Schmidt, E.E., Ichimura, K., Reifenberger, G., and Collins, V.P. (1994) CDKN2 (p16/MTS1) gene deletion or CDK4 amplification occurs in the majority of glioblastomas. *Cancer Res.* 54: 6321-6324.

Schold, S.C., Jr., Brent, T.P., von Hofe, E., Friedman, H.S., Mitra, S., Bigner, D.D., Swenberg, J.A., and Kleihues, P. (1989) O6-alkylguanine-DNA alkyltransferase and sensitivity to procarbazine in human brain-tumor xenografts. *J. Neurosurg.* **70:** 573-577.

Schwartzman, R.A. and Cidlowski, J.A. (1993) Apoptosis: the biochemistry and

molecular biology of programmed cell death. Endocr. Rev. 14: 133-151.

Schwechheimer, K., Huang, S., and Cavenee, W.K. (1995) EGFR gene amplification--rearrangement in human glioblastomas. *Int. J. Cancer* 62: 145-148.

Seizinger, B.R., Klinger, H.P., Junien, C., Nakamura, Y., Le Beau, M., Cavenee, W., Emanuel, B., Ponder, B., Naylor, S., Mitelman, F., Louis, D., Menon, A., Newsham, I., Decker, J., Kaelbing, I., Henry, I., and von Deimling, A. (1991) Report of the committee on chromosome and gene loss in human neoplasia. *Cytogenet. Cell Genet.* 58: 1080-1096.

Selvakumaran, M., Lin, H.K., Miyashita, T., Wang, H.G., Krajewski, S., Reed, J.C., Hoffman, B., and Liebermann, D. (1994) Immediate early up-regulation of bax expression by p53 but not TGF beta 1: a paradigm for distinct apoptotic pathways. *Oncogene* 9: 1791-1798.

Sen, S. and D'Incalci, M. (1992) Apoptosis. Biochemical events and relevance to cancer chemotherapy. *FEBS Lett.* 307: 122-127.

Shapiro, W.R., Green, S.B., Burger, P.C., Mahaley, M.S., Jr., Selker, R.G., Van Gilder, J.C., Robertson, J.T., Ransohoff, J., Mealey, J., Jr., Strike, T.A., and Pistenmaa, D.A. (1989) Randomized trial of three chemotherapy regimens and two radiotherapy regimens in postoperative treatment of malignant glioma. Brain Tumor Cooperative Group Trial 8001. J. Neurosurg. 71: 1-9.

Shaulsky, G., Goldfinger, N., Tosky, M.S., Levine, A.J., and Rotter, V. (1991) Nuclear localization is essential for the activity of p53 protein. *Oncogene* 6: 2055-2065.

Shivji, M.K., Grey, S.J., Strausfeld, U.P., Wood, R.D., and Blow, J.J. (1994) Cip1 inhibits DNA replication but not PCNA-dependent nucleotide excision-repair. *Curr. Biol.* **4**: 1062-1068.

Sidransky, D., Mikkelsen, T., Schwechheimer, K., Rosenblum, M.L., Cavenee, W., and Vogelstein, B. (1992) Clonal expansion of p53 mutant cells is associated with brain tumour progression. *Nature* 355: 846-847.

Silber, J.R., Mueller, B.A., Ewers, T.G., and Berger, M.S. (1993) Comparison of O6-methylguanine-DNA methyltransferase activity in brain tumors and adjacent normal brain. *Cancer Res.* 53: 3416-3420.

Silber, J.R., Blank, A., Bobola, M.S., Mueller, B.A., Kolstoe, D.D., Ojemann, G.A., and Berger, M.S. (1996) Lack of the DNA repair protein O6-methylguanine-DNA methyltransferase in histologically normal brain adjacent to primary human brain tumors. *Proc. Natl. Acad. Sci. U. S. A.* 93: 6941-6946.

Skoff, R.P. and Knapp, P.E. (1991) Division of astroblasts and oligodendroblasts in postnatal rodent brain: evidence for separate astrocyte and oligodendrocyte lineages. *Glia* **4:** 165-174.

Slichenmyer, W.J., Nelson, W.G., Slebos, R.J., and Kastan, M.B. (1993) Loss of a p53-associated G1 checkpoint does not decrease cell survival following DNA damage. *Cancer Res.* 53: 4164-4168.

Smith, M.L., Chen, I.T., Zhan, Q., Bae, I., Chen, C.Y., Gilmer, T.M., Kastan, M.B., O'Connor, P.M., and Fornace, A.J., Jr (1994) Interaction of the p53-regulated protein Gadd45 with proliferating cell nuclear antigen [see comments]. *Science* 266: 1376-1380.

Smith, M.T., Evans, C.G., Doane-Setzer, P., Castro, V.M., Tahir, M.K., and Mannervik, B. (1989) Denitrosation of 1,3-bis(2-chloroethyl)-1-nitrosourea by class mu glutathione transferases and its role in cellular resistance in rat brain tumor cells. *Cancer Res.* 49: 2621-2625.

Somfai-Relle, S., Suzukake, K., Vistica, B.P., and Vistica, D.T. (1984a) Reduction in cellular glutathione by buthionine sulfoximine and sensitization of murine tumor cells resistant to L-phenylalanine mustard. *Biochem. Pharmacol.* 33: 485-490.

Somfai-Relle, S., Suzukake, K., Vistica, B.P., and Vistica, D.T. (1984b) Glutathione-conferred resistance to antineoplastics: approaches toward its reduction. *Cancer Treat. Rev.* 11: 43-54.

Sonoda, Y., Yoshimoto, T., and Sekiya, T. (1995) Homozygous deletion of the MTS1/p16 and MTS2/p15 genes and amplification of the CDK4 gene in glioma. *Oncogene* 11: 2145-2149.

Soussi, T., Caron de Fromentel, C., Mechali, M., May, P., and Kress, M. (1987) Cloning and characterization of a cDNA from Xenopus laevis coding for a protein homologous to human and murine p53. *Oncogene* 1: 71-78.

Spence, A.M., Berger, M.S., Livingston, R.B., Ali Osman, F., and Griffin, B. (1992) Phase II evaluation of high-dose intravenous cisplatin for treatment of adult malignant gliomas recurrent after chloroethylnitrosourea failure. J. Neurooncol. 12: 187-191.

Stammberger, I., Schmahl, W., and Nice, L. (1990) The effects of X-irradiation, N-ethyl-N-nitrosourea or combined treatment on O6-alkylguanine-DNA alkyltransferase activity in fetal rat brain and liver and the induction of CNS tumours. *Carcinogenesis* 11: 219-222.

Stammberger, I. and Tempel, K. (1991) Absence of O6-alkylguanine-DNA alkyltransferase induction in chick embryo liver and brain following X-irradiation or

treatment with bleomycin. Comp. Biochem. Physiol. C. 100: 573-576.

Steck, P.A., Pershouse, M.A., Jasser, S.A., Yung, W.K., Lin, H., Ligon, A.H., Langford, L.A., Baumgard, M.L., Hattier, T., Davis, T., Frye, C., Hu, R., Swedlund, B., Teng, D.H., and Tavtigian, S.V. (1997) Identification of a candidate tumour suppressor gene, MMAC1, at chromosome 10q23.3 that is mutated in multiple advanced cancers. *Nat. Genet.* 15: 356-362.

Stenning, S.P., Freedman, L.S., and Bleehen, N.M. (1987) An overview of published results from radomized studies of nitrosoureas in primary high grade malignant glioma. *Br. J. Cancer* 56: 89-90.

Stewart, N., Hicks, G.G., Paraskevas, F., and Mowat, M. (1995) Evidence for a second cell cycle block at G2/M by p53. *Oncogene* 10: 109-115.

Stockli, K.A., Lillien, L.E., Naher Noe, M., Breitfeld, G., Hughes, R.A., Raff, M.C., Thoenen, H., and Sendtner, M. (1991) Regional distribution, developmental changes, and cellular localization of CNTF-mRNA and protein in the rat brain. J. Cell Biol. 115: 447-459.

Strange, R.C., Fryer, A.A., Matharoo, B., Zhao, L., Broome, J., Campbell, D.A., Jones, P., Pastor, I.C., and Singh, R.V. (1992) The human glutathione S-transferases: comparison of isoenzyme expression in normal and astrocytoma brain. *Biochim. Biophys.* Acta 1139: 222-228.

Strauss, B.E. and Haas, M. (1995) The region 3' to the major transcriptional start site of the MDR1 downstream promoter mediates activation by a subset of mutant p53 proteins. *Biochem. Biophys. Res. Commun.* 217: 333-340.

Sturzbecher, H.W., Brain, R., Addison, C., Rudge, K., Remm, M., Grimaldi, M., Keenan, E., and Jenkins, J.R. (1992) A C-terminal alpha-helix plus basic region motif is the major structural determinant of p53 tetramerization. *Oncogene* 7: 1513-1523.

Sugano, T., Nitta, M., Ohmori, H., and Yamaizumi, M. (1995) Nuclear accumulation of p53 in normal human fibroblasts is induced by various cellular stresses which evoke the heat shock response, independently of the cell cycle. *Jpn. J. Cancer Res.* 86: 415-418.

Swenberg, J.A., Bedell, M.A., Billings, K.C., Umbenhauer, D.R., and Pegg, A.E. (1982) Cell-specific differences in O6-alkylguanine DNA repair activity during continuous exposure to carcinogen. *Proc. Natl. Acad. Sci. U. S. A.* **79:** 5499-5502.

Tanaka, Y., Abe, Y., Tsugu, A., Takamiya, Y., Akatsuka, A., Tsuruo, T., Yamazaki, H., Ueyama, Y., Sato, O., Tamaoki, N., and Nakamura, M. (1994) Ultrastructural

localization of P-glycoprotein on capillary endothelial cells in human gliomas. Virchows Arch. 425: 133-138.

Tannock, I.F. (1992) Experimental chemotherapy. In: The Basic Science of Oncology, edited by Tannock, I.F. and Hill, R.P. Toronto, Canada: McGraw-Hill, pp. 338-359.

Teicher, B.A., Holden, S.A., Kelley, M.J., Shea, T.C., Cucchi, C.A., Rosowsky, A., Henner, W.D., and Frei, E., III (1987) Characterization of a human squamous carcinoma cell line resistant to *cis*-diamminedichloroplatinum(II). *Cancer Res.* 47: 388-393.

Tew, K.D., Bomber, A.M., and Hoffman, S.J. (1988) Ethacrynic acid and piriprost as enhancers of cytotoxicity in drug resistant and sensitive cell lines. *Cancer Res.* 48: 3622-3625.

Timme, T.L. and Thompson, T.C. (1994) Rapid allelotype analysis of p53 knockout mice. *Biotechniques* 17: 460, 462-3.

Tobey, R.A. (1975) Different drugs arrest cells at a number of distinct stages in G_2 . *Nature* 254: 245-247.

Tobey, R.A. and Crissman, H.A. (1975) Comparative effects of three nitrosourea derivatives on mammalian cell cycle progression. *Cancer Res.* 35: 460-470.

Tong, W.P., Kirk, M.C., and Ludlum, D.B. (1982) Formation of the cross-link $1-[N^3-deoxycytidy], 2-[N^1-deoxyguanosinyl]$ -ethane in DNA treated with N,N^2 -bis(2-chloroethyl)-N-nitrosourea. Cancer Res. 42: 3102-3105.

Trojan, J., Johnson, T.R., Rudin, S.D., Ilan, J., and Tykocinski, M.L. (1993) Treatment and prevention of rat glioblastoma by immunogenic C6 cells expressing antisense insulin-like growth factor I RNA. *Science* 259: 94-97.

Ueki, K., Ono, Y., Henson, J.W., Efird, J.T., von Deimling, A., and Louis, D.N. (1996) CDKN2/p16 or RB alterations occur in the majority of glioblastomas and are inversely correlated. *Cancer Res.* 56: 150-153.

Unger, E.R. (1992) In situ and northern hybridizations. Technical considerations guiding clinical application. *Cancer* 69: 1532-1535.

Usuki, S., Maekawa, A., Kang, H., Shumiya, S., and Nagase, S. (1992) High susceptibility of analbuminemic rats to neurogenic tumor induction by transplacental administration of N-ethyl-N-nitrosourea. *Jpn. J. Cancer Res.* 83: 146-152.

Utz, I., Gekeler, V., Ise, W., Beck, J., Spitaler, M., Grunicke, H., and Hofmann, J.

(1996) Protein kinase C isoenzymes, p53, accumulation of rhodamine 123, glutathione-S-transferase, topoisomerase II and MRP in multidrug resistant cell lines. *Anticancer Res.* 16: 289-296.

Van Meir, E.G., Kikuchi, T., Tada, M., Li, H., Diserens, A.C., Wojcik, B.E., Huang, H.J., Friedmann, T., de Tribolet, N., and Cavenee, W.K. (1994) Analysis of the p53 gene and its expression in human glioblastoma cells. *Cancer Res.* 54: 649-652.

van Meyel, D.J., Ramsay, D.A., Casson, A.G., Keeney, M., Chambers, A.F., and Cairneross, J.G. (1994) p53 mutation, expression, and DNA ploidy in evolving gliomas: evidence for two pathways of progression. *J. Natl. Cancer Inst.* **86:** 1011-1017.

Venter, D.J., Bevan, K.L., Ludwig, R.L., Riley, T.E., Jat, P.S., Thomas, D.G., and Noble, M.D. (1991) Retinoblastoma gene deletions in human glioblastomas. *Oncogene* **6**: 445-448.

Versantvoort, C.H., Broxterman, H.J., Bagrij, T., Scheper, R.J., and Twentyman, P.R. (1995) Regulation by glutathione of drug transport in multidrug-resistant human lung tumour cell lines overexpressing multidrug resistance-associated protein. *Br. J. Cancer* **72**: 82-89.

Vikhanskaya, F., D'Incalci, M., and Broggini, M. (1995) Decreased cytotoxic effects of doxorubicin in a human ovarian cancer-cell line expressing wild-type p53 and WAF1/CIP1 genes. *Int. J. Cancer* **61**: 397-401.

Vogelstein, B. and Kinzler, K.W. (1992) p53 function and dysfunction. Cell 70: 523-526.

von Deimling, A., Eibl, R.H., Ohgaki, H., Louis, D.N., von Ammon, K., Petersen, I., Kleihues, P., Chung, R.Y., Wiestler, O.D., and Seizinger, B.R. (1992a) p53 mutations are associated with 17p allelic loss in grade II and grade III astrocytoma. *Cancer Res.* **52**: 2987-2990.

von Deimling, A., Louis, D.N., von Ammon, K., Petersen, I., Hoell, T., Chung, R.Y., Martuza, R.L., Schoenfeld, D.A., Yasargil, M.G., Wiestler, O.D., and Seizinger, B.R. (1992b) Association of epidermal growth factor receptor gene amplification with loss of chromosome 10 in human glioblastoma multiforme. J. Neurosurg. 77: 295-301.

von Deimling, A., Louis, D.N., von Ammon, K., Petersen, I., Wiestler, O.D., and Seizinger, B.R. (1992c) Evidence for a tumor suppressor gene on chromosome 19q associated with human astrocytomas, oligodendrogliomas, and mixed gliomas. *Cancer Res.* 52: 4277-4279.

von Deimling, A., von Ammon, K., Schoenfeld, D., Wiestler, O.D., Seizinger, B.R.,

and Louis, D.N. (1993) Subsets of glioblastoma multiforme defined by molecular genetic analysis. *Brain Pathol.* 3: 19-26.

von Deimling, A., Louis, D.N., and Wiestler, O.D. (1995) Molecular pathways in the formation of gliomas. *Glia* 15: 328-338.

Waga, S., Hannon, G.J., Beach, D., and Stillman, B. (1994) The p21 inhibitor of cyclin-dependent kinases controls DNA replication by interaction with PCNA. *Nature* **369:** 574-578.

Walker, M.D., Alexander, E., Jr., Hunt, W.E., MacCarty, C.S., Mahaley, M.S., Mealey, J., Jr., Norrell, H.A., Owens, G., Ransohoff, J., Wilson, C.B., Gehan, E.A., and Strike, T.A. (1978) Evaluation of BCNU and/or radiotherapy in the treatment of anaplastic gliomas. A cooperative clinical trial. *J. Neurosurg.* **49**: 333-343.

Walker, M.D., Green, S.B., Byar, D.P., Alexander, E., Jr., Batzdorf, U., Brooks, W.H., Hunt, W.E., MacCarty, C.S., Mahaley, M.S., Mealey, J., Jr., Owens, G., Ransohoff, J., Robertson, J.T., Shapiro, W.R., Smith, K.R., Wilson, C.B., and Strike, T.A. (1980) Radomized comparisons of radiotherapy and nitrosoureas for the treatment of malignant glioma after surgery. *N. Engl. J. Med.* 303: 1323-1329.

Walther, W., Stein, U., and Pfeil, D. (1995) Gene transfer of human TNF alpha into glioblastoma cells permits modulation of mdr1 expression and potentiation of chemosensitivity. *Int. J. Cancer* **61**: 832-839.

Wang, P., Reed, M., Wang, Y., Mayr, G., Stenger, J.E., Anderson, M.E., Schwedes, J.F., and Tegtmeyer, P. (1994) p53 domains: structure, oligomerization, and transformation. *Mol. Cell. Biol.* 14: 5182-5191.

Wang, X.W., Forrester, K., Yeh, H., Feitelson, M.A., Gu, J.R., and Harris, C.C. (1994) Hepatitis B virus X protein inhibits p53 sequence-specific DNA binding, transcriptional activity, and association with transcription factor ERCC3. *Proc. Natl.* Acad. Sci. U. S. A. 91: 2230-2234.

Wang, X.W., Yeh, H., Schaeffer, L., Roy, R., Moncollin, V., Egly, J.-M., Wang, Z., Freidberg, E.C., Evans, M.K., Taffe, B.G., Bohr, V.A., Weeda, G., Hoeijmakers, J.H.J., Forrester, K., and Harris, C.C. (1995) p53 modulation of TFIIH-associated nucleotide excision repair activity. *Nat. Genet.* **10**: 188-195.

Wang, X.W. and Harris, C.C. (1996) TP53 tumour suppressor gene: clues to molecular carcinogenesis and cancer therapy. *Cancer Surv.* 28: 169-196.

Wang, Y., Reed, M., Wang, P., Stenger, J.E., Mayr, G., Anderson, M.E., Schwedes, J.F., and Tegtmeyer, P. (1993) p53 domains: identification and characterization of two

autonomous DNA-binding regions. Genes Dev. 7: 2575-2586.

Wang, Y. and Prives, C. (1995) Increased and altered DNA binding of human p53 by S and G2/M but not G1 cyclin-dependent kinases. *Nature* 376: 88-91.

Wani, G., Wani, A.A., and D'Ambrosio, S.M. (1992) In situ hybridization of human kidney tissue reveals cell-type-specific expression of the O6-methylguanine-DNA methyltransferase gene. *Carcinogenesis* 13: 463-468.

Wechsler, W., Kleihues, P., Matsumoto, S., Zulch, K.J., Ivankovic, S., Preussmann, R., and Druckrey, H. (1969) Pathology of experimental neurogenic tumors chemically induced during prenatal and postnatal life. *Ann. N. Y. Acad. Sci.* **159:** 360-408.

Wedge, S.R. and Newlands, E.S. (1996) O6-benzylguanine enhances the sensitivity of a glioma xenograft with low O6-alkylguanine-DNA alkyltransferase activity to temozolomide and BCNU. *Br. J. Cancer* 73: 1049-1052.

Westermark, B., Heldin, C.-H., and Nister, M. (1995) Platelet-derived growth factor in human glioma. *Glia* 15: 257-263.

Wiencke, J.K. and Wiemels, J. (1995) Genotoxicity of 1,3-bis(2-chloroethyl)-1-nitrosourea (BCNU). *Mutat. Res.* 339: 91-119.

Wiestler, O., Kleihues, P., and Pegg, A.E. (1984) O6-alkylguanine-DNA alkyltransferase activity in human brain and brain tumors. *Carcinogenesis* 5: 121-124.

Will, K., Warnecke, G., Bergmann, S., and Deppert, W. (1995) Species- and tissue-specific expression of the c-terminal alternatively spliced form of the tumor suppressor p53. *Nucleic Acids Res.* 23: 4023-4028.

Williams, B.P., Abney, E.R., and Raff, M.C. (1985) Macroglial cell development in embryonic rat brain: studies using monoclonal antibodies, fluorescence activated cell sorting, and cell culture. *Dev. Biol.* 112: 126-134.

Wilson, R.E., Hoey, B., and Margison, G.P. (1993) Ionizing radiation induces O6-alkylguanine-DNA-alkyltransferase mRNA and activity in mouse tissues. *Carcinogenesis* 14: 679-683.

Wolf, D., Harris, N., and Rotter, V. (1984) Reconstitution of p53 expression in a nonproducer Ab-MuLV-transformed cell line by transfection of a functional p53 gene. *Cell* 38: 119-126.

Wolswijk, G., Riddle, P.N., and Noble, M. (1991) Platelet-derived growth factor is mitogenic for O-2Aadult progenitor cells. *Glia* 4: 495-503.

Wolswijk, G. and Noble, M. (1992) Cooperation between PDGF and FGF converts slowly dividing O-2Aadult progenitor cells to rapidly dividing cells with characteristics of O-2Aperinatal progenitor cells. J. Cell Biol. 118: 889-900.

Wong, A.J., Ruppert, J.M., Bigner, S.H., Grzeschik, C.H., Humphrey, P.A., Bigner, D.S., and Vogelstein, B. (1992) Structural alterations of the epidermal growth factor receptor gene in human gliomas. *Proc. Natl. Acad. Sci. U. S. A.* 89: 2965-2969.

Wrensch, M., Bondy, M.L., Wiencke, J., and Yost, M. (1993) Environmental risk factors for primary malignant brain tumors: a review. J. Neurooncol. 17: 47-64.

Wu, J.K., Ye, Z., and Darras, B.T. (1993) Frequency of p53 tumor suppressor gene mutations in human primary brain tumors. *Neurosurgery* 33: 824-830.

Wu, Y., Liu, Y., Lee, L., Miner, Z., and Kulesz Martin, M. (1994) Wild-type alternatively spliced p53: binding to DNA and interaction with the major p53 protein in vitro and in cells. *EMBO J.* 13: 4823-4830.

Wu, Z.N., Chan, C.L., Eastman, A., and Bresnick, E. (1992) Expression of human O6-methylguanine-DNA methyltransferase in a DNA excision repair-deficient Chinese hamster ovary cell line and its response to certain alkylating agents. *Cancer Res.* 52: 32-35.

Xiao, H., Pearson, A., Coulombe, B., Truant, R., Zhang, S., Regier, J.L., Triezenberg, S.J., Reinberg, D., Flores, O., Ingles, C.J., and Greenblatt, J. (1994) Binding of basal transcription factor TFIIH to the acidic activation domains of VP16 and p53. *Mol. Cell. Biol.* 14: 7013-7024.

Xiong, Y., Hannon, G.H., Zhang, H., Casso, D., Kobayashi, R., and Beach, D. (1993) p21 is a universal inhibitor of cyclin kinases. *Nature* 366: 701-704.

Yamaizumi, M. and Sugano, T. (1994) U.v.-induced nuclear accumulation of p53 is evoked through DNA damage of actively transcribed genes independent of the cell cycle. *Oncogene* 9: 2775-2784.

Yarosh, D.B. (1985) The role of O6-methylguanine-DNA methyltransferase in cell survival, mutagenesis and carcinogenesis. *Mutat. Res.* 145: 1-16.

Yarosh, D.B., Hurst Calderone, S., Babich, M.A., and Day, R.S., III (1986) Inactivation of O6-methylguanine-DNA methyltransferase and sensitization of human tumor cells to killing by chloroethylnitrosourea by O6-methylguanine as a free base. *Cancer Res.* 46: 1663-1668.

Yin, X.M., Oltvai, Z.N., and Korsmeyer, S.J. (1995) Heterodimerization with Bax is

required for Bcl-2 to repress cell death. Curr. Top. Microbiol. Immunol. 194: 331-338.

Yin, Y., Tainsky, M.A., Bischoff, F.Z., Strong, L.C., and Wahl, G.M. (1992) Wild-type p53 restores cell cycle control and inhibits gene amplification in cells with mutant p53 alleles. *Cell* **70**: 937-948.

Yonish Rouach, E., Resnitzky, D., Lotem, J., Sachs, L., Kimchi, A., and Oren, M. (1991) Wild-type p53 induces apoptosis of myeloid leukaemic cells that is inhibited by interleukin-6. *Nature* **352**: 345-347.

Yoshida, K. and Gage, F.H. (1992) Cooperative regulation of nerve growth factor synthesis and secretion in fibroblasts and astrocytes by fibroblast growth factor and other cytokines. *Brain Res.* 569: 14-25.

Yung, W.K., Janus, T.J., Maor, M., and Feun, L.G. (1992) Adjuvant chemotherapy with carmustine and cisplatin for patients with malignant gliomas. *J. Neurooncol.* 12: 131-135.

Zakut-Houri, R., Oren, M., Bienz, B., Lavie, V., and Givol, D. (1983) A single gene and a pseudogene for the cellular tumour antigen p53. *Nature* **306**: 594-597.

Zaman, G.J., Flens, M.J., van Leusden, M.R., de Haas, M., Mulder, H.S., Lankelma, J., Pinedo, H.M., Scheper, R.J., Baas, F., Broxterman, H.J., and Borst, P. (1994) The human multidrug resistance-associated protein MRP is a plasma membrane drug-efflux pump. *Proc. Natl. Acad. Sci. U. S. A.* **91**: 8822-8826.

Zaman, G.J., Lankelma, J., van Tellingen, O., Beijnen, J., Dekker, H., Paulusma, C., Oude Elferink, R.P., Baas, F., and Borst, P. (1995) Role of glutathione in the export of compounds from cells by the multidrug-resistance-associated protein. *Proc. Natl. Acad. Sci. U. S. A.* 92: 7690-7694.

Zastawny, R.L., Salvino, R., Chen, J., Benchimol, S., and Ling, V. (1993) The core promoter region of the *P*-glycoprotein gene is sufficient to confer differential responsiveness to wild-type and mutant *p53*. Oncogene 8: 1529-1535.

Zelazowski, A.J., Garvey, J.S., and Hoeschele, J.D. (1984) In vivo and in vitro binding of platinum to metallothionein. Arch. Biochem. Biophys. 229: 246-252.

Zhang, H., Hannon, G.J., and Beach, D. (1994) p21-containing cyclin kinases exist in both active and inactive states. *Genes Dev.* 8: 1750-1758.

Zhang, S., Endo, S., Koga, H., Ichikawa, T., Feng, X., Onda, K., Washiyama, K., and Kumanishi, T. (1996) A comparative study of glioma cell lines for p16, p15, p53 and p21 gene alterations. *Jpn. J. Cancer Res.* 87: 900-907.













O 1993, Applied Image, Inc., All Rights Reserved